LOCAL, SEMI-LOCAL AND GLOBAL MODELS FOR TEXTURE, OBJECT AND SCENE RECOGNITION

Svetlana Lazebnik, Ph.D. Department of Computer Science University of Illinois at Urbana-Champaign, 2006 Jean Ponce, Advisor

This dissertation addresses the problems of recognizing textures, objects, and scenes in photographs. We present approaches to these recognition tasks that combine salient *local image features* with spatial relations and effective discriminative learning techniques. First, we introduce a *bag of features* image model for recognizing textured surfaces under a wide range of transformations, including viewpoint changes and non-rigid deformations. We present results of a large-scale comparative evaluation indicating that bags of features can be effective not only for texture, but also for object categization, even in the presence of substantial clutter and intra-class variation. We also show how to augment the purely local image representation with statistical co-occurrence relations between pairs of nearby features, and develop a learning and classification framework for the task of classifying individual features in a multi-texture image. Next, we present a more structured alternative to bags of features for object recognition, namely, an image representation based on *semi-local parts*, or groups of features characterized by stable appearance and geometric layout. Semi-local parts are automatically learned from small sets of unsegmented, cluttered images. Finally, we present a global method for recognizing scene categories that works by partitioning the image into increasingly fine sub-regions and computing histograms of local features found inside each sub-region. The resulting *spatial pyramid* representation demonstrates significantly improved performance on challenging scene categorization tasks.

LOCAL, SEMI-LOCAL AND GLOBAL MODELS FOR TEXTURE, OBJECT AND SCENE RECOGNITION

BY

SVETLANA LAZEBNIK

B.S., DePaul University, 2000 M.S., University of Illinois at Urbana-Champaign, 2002

DISSERTATION

Submitted in partial fulfillment of the requirements for the degree of Doctor of Philosophy in Computer Science in the Graduate College of the University of Illinois at Urbana-Champaign, 2006

Urbana, Illinois

 \bigodot Copyright by Svetlana Lazebnik, 2006

ABSTRACT

This dissertation addresses the problems of recognizing textures, objects, and scenes in photographs. We present approaches to these recognition tasks that combine salient *local image features* with spatial relations and effective discriminative learning techniques. First, we introduce a *bag of features* image model for recognizing textured surfaces under a wide range of transformations, including viewpoint changes and non-rigid deformations. We present results of a large-scale comparative evaluation indicating that bags of features can be effective not only for texture, but also for object categization, even in the presence of substantial clutter and intra-class variation. We also show how to augment the purely local image representation with statistical co-occurrence relations between pairs of nearby features, and develop a learning and classification framework for the task of classifying individual features in a multi-texture image. Next, we present a more structured alternative to bags of features for object recognition, namely, an image representation based on *semi-local parts*, or groups of features characterized by stable appearance and geometric layout. Semi-local parts are automatically learned from small sets of unsegmented, cluttered images. Finally, we present a global method for recognizing scene categories that works by partitioning the image into increasingly fine sub-regions and computing histograms of local features found inside each sub-region. The resulting *spatial pyramid* representation demonstrates significantly improved performance on challenging scene categorization tasks.

To Max

ACKNOWLEDGMENTS

The research presented in this dissertation was partially supported by the National Science Foundation under grants IIS-0308087, ITR-0312438 and IIS-0535152, Toyota, the UIUC Campus Research Board, the Beckman Institute for Advanced Science and Technology, the UIUC-CNRS collaboration agreement, and the European project LAVA. I would also like to acknowledge the UIUC Computer Science Department and the College of Engineering for supporting me with the SURGE fellowship and various awards.

I am grateful to all the members of my Ph.D. thesis committee, and especially to Jean Ponce and Cordelia Schmid, for guiding and encouraging my research. I would also like to thank Marcin Marszalek and Jianguo Zhang for implementing extensions to my texture recognition methods and providing the experimental results for Section 3.2. Everlasting gratitude is due to my wonderful husband Maxim Raginsky and to the rest of my family, including my sister Maria (who will herself be defending her Ph.D. a year or two from now), my parents, and grandmother.

TABLE OF CONTENTS

CHAPTER

PAGE

1	Introduction					
	1.1	Recog	tion Tasks	;		
		1.1.1	Cexture Recognition 6	;		
		1.1.2	Dbject Recognition 7	7		
		1.1.3	Scene Recognition	;		
	1.2	Them	Contributions and Outline)		
2	Pre	vious '	ork	,		
	2.1	Featur	Detectors and Descriptors 13	;		
	2.2	Textu	Recognition 16	j		
	2.3	Objec	Recognition $\ldots \ldots \ldots$	7		
	2.4	Scene	nd Context Recognition)		
3	ΑB	ag-of-	eatures Model for Texture and Object Recognition	3		
-	3.1	Textu	Recognition Using Local Affine Regions	Ļ		
	0.1	3.1.1	Components of the Approach	7		
			Affine Region Detection	7		
			3.1.1.2 Rotation-Invariant Descriptors	;		
			3.1.1.3 Signatures and the Earth Mover's Distance			
		3.1.2	Experimental Evaluation	;		
			5.1.2.1 Evaluation Strategy	;		
			1.2.2 UIUC Texture Database	j		
			.1.2.3 Brodatz Database 40)		
	3.2	An Ex	nded Evaluation: From Texture to Object Recognition	ļ		
		3.2.1	Kernel-based classification 46	j		
		3.2.2	Cexture Recognition)		
			2.2.1 Comparing Invariance Levels and Descriptor Types 49)		
			2.2.2.2 Comparative Evaluation: UIUC Database, Brodatz, CUReT . 51	_		
		3.2.3	Dbject Recognition: Caltech6, Caltech101, Graz	j		
	3.3	Discus	on			

4	Neighborhood Co-occurrence Relations for Texture Recognition 63				
	4.1	A Maximum Entropy Framework for Combining Local Features and Their			
		Relations			
		4.1.1 The Maximum Entropy Classifier			
		4.1.2 Texton Vocabulary and Feature Functions			
		4.1.3 Experimental Results: UIUC Database and Brodatz 69			
	4.2	A Two-Stage Approach for Recognizing Local Texture Regions			
		4.2.1 Modeling Textures			
		4.2.1.1 Density Estimation $\dots \dots \dots$			
		4.2.1.2 Neighborhood Statistics			
		$4.2.1.3 \text{Relaxation} \dots \dots \dots \dots \dots \dots \dots \dots \dots $			
		4.2.1.4 Classification and Retrieval			
		4.2.2 Experimental Results			
		$4.2.2.1 \text{The Indoor Scene} \dots \dots$			
		$4.2.2.2 \text{Animals} \dots \dots$			
	4.3	Discussion			
	~				
5	Sen	ni-Local Parts for Object Recognition			
	5.1	Motivation $\dots \dots \dots$			
	5.2	Semi-Local Parts			
		5.2.1 Matching of Pairs of Images			
		5.2.2 Validation			
	F 0	5.2.3 Recognition			
	5.3	Recognition Experiments			
		5.3.1 Affine-Invariant Parts104			
	- ,	5.3.2 Scale-Invariant Parts			
	5.4	Discussion			
G	Spa	tial Dynamid Matching for Scone Decognition 100			
U	Spa 6 1	Mativation 122			
	0.1 6 9	Spatial Dyramid Matching			
	0.2	6.2.1 Dyramid Match Kornela 124			
		6.2.2 Spatial Matching Schome			
	63	5.2.2 Spatial Matching Scheme			
	0.3 6.4	Feature Extraction			
	0.4	6.4.1 Score Category Perception 120			
		$6.4.2 \text{Caltach}101 \qquad \qquad 125$			
		6.4.2 Callectiful			
	65	0.4.5 The Graz Dataset			
	0.0				
7	Cor	nclusion			
•	7.1	Summary			
	7.2	Extensions			
	• • -				

APPENDIX A Harris and Laplacian Region Detectors	.145
A.1 Spatial and Scale Selection	.145
A.2 Affine Adaptation	. 148
REFERENCES	. 151
AUTHOR'S BIOGRAPHY	. 163

CHAPTER 1

Introduction

The recognition of materials, objects, and scene categories from photographs are among the most central problems in the field of computer vision. However, despite decades of intensive research, even the most sophisticated recognition systems today (see [139] for a summary of the state of the art) remain incapable of handling more than just a few simple classes, or of functioning under unconstrained real-world conditions. What makes recognition so difficult is the seemingly limitless variability of natural imagery that arises from viewpoint and lighting changes, movement and deformation of non-rigid or articulated objects, intraclass appearance variations, and the presence of occlusion and background clutter (see Figure 1.1 for examples of different kinds of visual classes with some representative variations). This dissertation proposes novel models of image content that are robust to many of these sources of variability and can achieve high recognition performance on challenging datasets.

As basic building blocks for our models, we use *local image features*, or appearance-based descriptors computed over regions of support whose size is typically much smaller than that of the entire image. As shown in Figure 1.2, a patch-based description or representation of an image can be obtained by processing the image with one of the specialized scale- or affine-invariant salient region detectors that have been introduced over the last decade [61, 62, 83, 90, 101, 104, 153]. Alternatively, for applications that do not require geometric invariance, we can simply sample rectangular patches randomly [100] or on a regular grid [30]. In recent literature, local features have been used for image indexing and retrieval [103, 137, 138],



Texture and material recognition: Textured surfaces may be photographed in a wide range of scales and orientations, and significant perspective distortions may be present in images. Some materials, like marble, are highly nonhomogeneous, and others, like cloth, are non-rigid.



Object recognition: Objects may be difficult to detect because of clutter and occlusion. Animals are non-rigid and appear in different poses. Some classes, like giraffes, can be characterized well by their texture, but others, like egrets, are textureless and require a more structured representation.



Scene recognition: Natural scene categories have significant intra-class variation. Some indoor categories, such as kitchen and living room, may be difficult to distinguish from one another.

Figure 1.1 Examples of images for each of the recognition tasks considered in this dissertation.



Figure 1.2 Two alternative patch-based representations of two natural images. Left: original images, center: patches found by the Harris detector, right: patches found by the Laplacian detector.

wide-baseline matching [3, 134, 135, 153], description and matching of video clips [129, 144], recognizing different views of the same object [34, 130] and different instances of the same object class [1, 23, 33, 163]. Indeed, using local features as image primitives has several important advantages:

- Robustness: Because local features are relatively small and compact, they can be preserved even when large portions of an image are affected by clutter or occlusion.
- Repeatability: Many existing local feature detectors can reliably identify corresponding features in different images despite geometric transformations, changes in lighting, or minor appearance variations. These may be features corresponding to the same surface patch in two views of the same object, or features corresponding to analogous structures, such as eyes, on different instances of the same class.

- Expressiveness: Unlike the geometric features historically used in computer vision (points or line segments), today's local features contain information not only about their shape (circular, elliptical, or rectangular), but also about their appearance. Rich high-dimensional descriptors of appearance provide strong consistency constraints for matching tasks.
- Local geometric invariance: Depending on the requirements of a particular application, one may choose to use scale-, rotation- or affine-invariant local features. In particular, affine invariance offers robustness to a wide range of geometric transformations that can be *locally* approximated by a linear model, including perspective distortions and non-rigid deformations.
- Compactness or sparsity: The number of features returned by most detectors is typically orders of magnitude smaller than the total number of pixels in the original image. The resulting patch-based description is extremely compact, thus reducing processing time and storage requirements.

In this dissertation, we present several novel approaches for modeling visual classes. These approaches can be classified into three types, based on the kinds of geometric relations between local features that they encode (Figure 1.3): *local* models disregard all spatial information; *semi-local* models capture relations between groups of nearby regions; and *global* models encode the spatial layout of all features in the image. In our work, local models (Chapter 3) are aimed primarily at texture recognition, though in Section 3.2, we show them to be surprisingly effective for object categorization as well. Semi-local models (Chapter 5) are used to create a part-based object representation. Finally, global models (Chapter 6) are used to describe scenes, though they can also work well for recognizing object classes in the absence of geometric deformations. The next section discusses the tasks of texture, object and scene recognition in more detail, together with their associated models.



Figure 1.3 The three kinds of models used in this thesis. Top left: a *local* model is a distribution or histogram of appearance descriptors of salient regions extracted from the image. This model is most appropriate for texture recognition. Top right: a *semi-local* model is a geometrically stable group of neighboring local features lying on the object of interest. We use *semi-local parts* to represent object structure. Bottom: a *global* model is a set of histograms computed over a multi-level pyramid decomposition of the image. This model is most effective for scene recognition.

1.1 Recognition Tasks

1.1.1 Texture Recognition

We want to recognize images of textured surfaces subject to viewpoint changes and nonrigid deformations. For this task, we use a *local* model in which spatial constraints are absent, i.e., each patch is considered separately, without any information about its neighborhood or its position in the image. In the framework presented in Chapter 3, each image is represented by the distribution of appearance descriptors of the features contained in it. The distribution is learned by quantizing the descriptors in the image and forming a *signature*, or a set of all cluster centers together with weights indicating the relative sizes of the clusters. Signatures of different images are compared using *Earth Mover's Distance* (EMD) [131, 132], which solves a partial matching problem between sets of possibly unequal cardinality, and is robust to noise, clutter, and outliers. We also investigate an alternative *bag-of-features* approach, in which local features are quantized into "textons" or "visual words" drawn from some universal vocabulary, and their distributions in images are represented as histograms of texton labels. This approach is analogous to the *bag-of-words* paradigm for text document analysis [9, 52, 102, 114, 133].

A major shortcoming of bag-of-features methods is their disregard of spatial relations. For many natural textures, the spatial layout of local features captures perceptually important information about the class. Augmenting our texture representation with geometric information can be expected to increase its ability to discriminate between textures that have similar local elements but different geometric patterns. In the first part of Chapter 4, we present a texton-and-relations model that takes inspiration from *bigram models* used to describe text documents. Specifically, our model represents not only the frequencies of individual textons, but also the frequencies of co-occurrences of pairs of texton labels at nearby locations in the image. In the second part of Chapter 4, we consider the task of labeling individual features in multi-texture images. This task is more challenging than classification of single-texture images, because the local appearance of the feature itself is often insufficiently discriminative (for example, if we observe a uniformly blue image patch, we do not know whether it is water or sky). To reduce this ambiguity, we introduce a two-level model that first computes the probabilities of class membership of individual regions using a purely local model, and then refines these probabilities using a relaxation framework that relies on spatial relations to obtain contextual information.

1.1.2 Object Recognition

Our second target problem is recognizing object categories despite 3D viewpoint changes, intra-class appearance variations, non-rigid motions, as well as clutter and occlusion in the test images. At the most basic level, object recognition may be considered as a *whole-image classification problem*, i.e., identifying which object class is present in an image, without attempting to segment or localize that object. For this (admittedly simplified) task, it is possible to represent the "visual texture" of images containing objects using the orderless bagof-features model described in Section 1.1.1. Such models have been used in several recent approaches to visual categorization [20, 164], unsupervised discovery of visual "topics" [30, 122, 143], and video retrieval [144]. In these publications and in the evaluation presented in the second half of Chapter 3, orderless models have achieved surprisingly high levels of accuracy on several challenging datasets.

Despite its practical advantages, a bag of features is an extremely impoverished representation for object classes, since it ignores all geometric information about the object class, fails to distinguish between foreground and background features, and cannot segment an object from its surroundings. In chapter 5, we propose a more structured object representation based on composite *semi-local parts*, defined as geometrically stable configurations of multiple local features that are robust against approximately rigid deformations and intra-class variations. An object model consists of a collection of multiple semi-local parts, and these parts, in turn, can be connected to each other by looser geometric relations. Semi-local parts are detected using an alignment-like geometric search, which can in principle work with any rigid 2D transformation model, from translation to projective (in the experiments presented in Chapter 5, we demonstrate scale- and affine-invariant parts). Note, however, that in searching for semi-local parts, we do not assume that the entire object of interest is *globally* planar and/or rigid, only that it possesses some sub-components that can be approximately characterized in this way. Thus, our approach extends the level of geometric invariance of existing part-based object models [1, 28, 33, 163] which are mostly limited to recognizing fronto-parallel views of upright objects. Our learning framework does not require objects in training images to be hand-segmented or hand-aligned, it can work with minimal amounts of training data, and is robust to substantial noise, background variation, and occlusion.

1.1.3 Scene Recognition

Our third problem is recognizing semantic scene categories such as beach, mountain, school, office, etc. Just as texture and object recognition, this task can be approached using purely local models [30, 91]. However, in this dissertation we show that improved performance can be achieved by a *global* model that takes into account the absolute positions of the features. Intuitively, a global model has improved discriminative power because it captures spatial regularities that are important to our perception of natural scenes (for example, sky is usually above ground or water, building walls are usually vertical, the horizon is usually a horizontal line, etc.). In Chapter 6, we propose a *spatial pyramid* model that is based on forming a multi-level quadtree decomposition of an image and aggregating statistics of local features over each cell of the decomposition at each level (Figure 1.3, bottom). This approach is inspired by the *pyramid matching* framework of Grauman and Darrell [45], who form a multi-scale decomposition of the high-dimensional space of appearance descriptors, instead of the two-dimensional image space. Note that the number of levels in the image

decomposition is an implementation parameter of our approach; the higher it is, the more precisely individual features are localized. When the number of levels is zero, the spatial pyramid representation reduces to a standard bag of features. Our experiments demonstrate that the spatial pyramid achieves a significant improvement in performance over bags of features not only for scene classification tasks, but also for object classification tasks that do not require geometric invariance.

1.2 Themes, Contributions and Outline

The recognition approaches presented in this dissertation are based on combining salient local features with various kinds of geometric relations and effective discriminative learning techniques. Let us briefly discuss the major issues or themes associated with each of these components.

When dealing with local features, we are usually most concerned with the issue of *invariance*. In computer vision literature, there exist feature extraction schemes that are invariant to scaling, scaling with rotation, and to affine transformations (see Section 2.1 for a literature review). As discussed in Chapter 3, by using these features, we can compensate for various geometric transformations in the images, from global scale changes, rotations and affine deformations, to perspective changes and even non-rigid distortions. Given datasets that contain a wide range of these transformations, approaches that do not have intrinsically invariant features are at a disadvantage, since they require training exemplars to sample all the possible changes that may appear in the test set. Thus, with geometrically invariant features, we can learn from much smaller training sets. However, there is a tradeoff: the higher the level of invariance, the greater the amount of discriminative information that is lost during the feature normalization process. In the course of the experimental evaluation reported in Section 3.2, we have found that the best results are usually achieved with the lowest level of invariance absolutely required for a given application. Finally, in some ap-

plications such as scene recognition (Chapter 6), the sources of intra-class variability are so complex that they are best dealt with by statistical methods instead of geometric ones. In these cases, invariant features are not likely to bring much of a benefit at all, and an exemplar-based approach is more appropriate.

Another theme that runs through this dissertation is that of matching or correspondence. The bag-of-features method of Chapter 3 is based on comparing distributions of features in two images. To perform this comparison, we represent the distributions as two discrete sets of points in high-dimensional appearance space and use the Earth Mover's Distance (EMD) to find the lowest-cost matching between these two sets. Note that in this case, the matching does not take into account any spatial information. By contrast, our global model of Chapter 6 is based on the idea of matching two sets of features in image space, not in appearance space. Since our target application of scene recognition requires only a rough notion of spatial consistency, we obtain an efficient approximation to the matching cost using a pyramid match kernel that can also be used to approximate EMD [45]. Finally, our part-based object recognition approach of Chapter 5 uses a much more precise operation of matching two sets of features that have consistent appearance and geometric layout. We implement this operation as a constrained alignment search. While it is more computationally expensive than pyramid matching, it has the advantage of providing us with explicit correspondences, and also allowing us to match groups of features that have been subjected to (approximately rigid) affine transformations.

An important unifying theme associated with the learning component of our methods is our emphasis on *weakly supervised learning*, i.e., the ability to construct texture, object and scene models from unsegmented and cluttered training images. Today, fully supervised learning is still the standard for many recognition approaches — it is routine, for example, to hand-segment training images and hand-label individual features as belonging to various object classes. However, the effort and expense associated with these tasks will make detailed annotation impractical for the large-scale datasets that vision algorithms will have to handle in the future. Therefore, reducing the amount of manual supervision required by vision systems is crucial for enabling them to function in real-world applications.

The main contributions of this dissertation can be summarized as follows:

- Our approach to texture recognition (Chapter 3) is the first one to rely on local features produced by scale- and affine-invariant interest region detectors. The resulting texture representation is intrinsically invariant to a wide range of geometric transformations, and therefore can be learned from relatively few images, unlike other existing approaches, which require exemplars of every possible transformation to be included in the training set.
- In Section 3.2, we extend our bag-of-features method with a Support Vector Machine (SVM) classifier, and the resulting system outperforms other state-of-the-art approaches on several challenging object databases. This part of our work is performed in collaboration with Jianguo Zhang and Marcin Marszalek at INRIA Rhône-Alpes.
- To our knowledge, our approach for labeling individual texture regions (Section 4.2) is the only one to date that can learn texture models in a *weakly supervised* manner, i.e., from multi-texture images that are not segmented, only labeled by the textures that they contain.
- Our object representation in terms of semi-local parts (Chapter 5) has a number of advantages over constellation models [163, 33] and methods based on low-distortion correspondence [5]: it does not assume that the object is globally planar and/or rigid, it can achieve a higher degree of geometric invariance (up to affine, which can approximate viewpoint changes), and can tolerate a much greater degree of clutter during learning.
- The spatial pyramid method introduced in Chapter 6) is a simple, yet effective extension of the basic bag-of-features model, capable of achieving significantly higher

performance on scene classification, and on object classification in the absence of clutter and geometric variation. In particular, our method exceeds the state of the art on the Caltech101 dataset [29].

The rest of this dissertation is organized as follows. Chapter 2 reviews existing computer vision literature on local image features, as well as methods for texture, object, and scene recognition. Chapter 3 presents our bag-of-features model for texture and object classification. Chapter 4 describes two extensions to this model that incorporate pairwise relations between nearby features. Chapter 5 presents our representation of object classes in terms of semi-local parts, and Chapter 6 presents our global spatial pyramid model for scene classification. Finally, Chapter 7 closes the dissertation with a summary of our contributions and discussion of possible extensions and future research directions. The work described in this dissertation has been previously published in [71, 72, 73, 74, 75, 76, 168].

CHAPTER 2

Previous Work

This chapter reviews existing work on feature detectors and descriptors, as well as methods for texture, object, and scene recognition.

2.1 Feature Detectors and Descriptors

Early work on the extraction of local features in natural images includes the interest operators of Moravec [108] and Harris [50], as well as various image decompositions into perceptually salient blob-like primitives [19, 95, 160]. Blostein and Ahuja [10] were the first to introduce a multiscale blob detector based on maxima of the Laplacian. Lindeberg [82] has extended this detector in the framework of *automatic scale selection*, where a "blob" is defined by a scale-space location where a normalized Laplacian measure attains a local maximum. Informally, the spatial coordinates of the maximum become the coordinates of the center of the blob, and the scale at which the maximum is achieved becomes its *characteristic scale*. Gårding and Lindeberg [40] have also shown how to design an affine blob detector using an *affine adaptation* process based on the second moment matrix. This process forms an important part of two affine-invariant¹ region detection schemes [3, 104] that rely on a multiscale version of the Harris operator to localize interest points in space.

¹It is more technically correct to refer to these regions as *affine-covariant*, which means that the regions detected in an affinely transformed version of an image can be obtained by subjecting the regions found in the original image to the same transformation. However, the term *affine-invariant* is common in the literature [104, 105], and so we use it throughout this dissertation for consistency.



Figure 2.1 The decomposition of the region of support used by the shape context (left) and the SIFT descriptor (right).

In the work presented in Chapters 3 - 5 of this dissertation, we use the Laplacian detector of Gårding and Lindeberg [40] and the Harris detector of Mikolajczyk and Schmid [104, 105]. A brief description of these detectors, including details of the scale selection and affine adaptation procedures mentioned above, is given in the Appendix. Alternative region extraction schemes include the "entropy detector" of Kadir and Brady [62], the differenceof-Gaussians (or DoG) detector of Lowe [90] (see also Crowley and Parker [19] for related work), the "maximally stable extremal regions" of Matas et al. [101], and the corner- and intensity-based operators of Tuytelaars and Van Gool [153]. Of the above, [62, 90] are scaleinvariant, while [101, 153] are fully affine-invariant. This proliferation of region detectors, motivated primarily by applications to wide-baseline stereo matching and image retrieval, attests to the increased importance accorded to the spatial and shape selection principles in the computer vision community. A comparative evaluation of several state-of-the-art region detectors is presented in [107].

Detection of local image regions is only the first part of the feature extraction process; the second part is the computation of descriptors to characterize the appearance of these regions. A good descriptor should be distinctive, so as to provide strong consistency constraints for image matching, yet robust to illumination changes and other "nuisance" appearance variations. A classic method of image description is in terms of differential invariants derived from the *local jet*, or the vector of image derivatives computed up to a specified order [65]. Another classic method is to describe the appearance of an image region by the outputs of a set of linear filters, such as Gabor filters [39, 98], wavelets [158], or steerable filters [37]. In current literature, there exists a wide variety of filter banks custom-engineered for specific applications (see [125] for a survey). Despite the historical popularity of filter banks, they do not always provide the best method of image description. For example, it has been shown that raw pixel values can be more effective than filter outputs for texture classification [157]. In recent literature, very promising results have been achieved by so-called *distribution-based* descriptors [106], such as the shape context [4] and the Scale-Invariant Feature Transform (SIFT) [89, 90], that work by subdividing the region of support and counting appearance attributes (edge points or oriented edge points) inside each subregion. The shape context uses a log-polar decomposition of a circular image region, and computes counts of edge points in each spatial bin. The example in Figure 2.1, left uses 12 angular bins and 4 radial bins, for a 48-dimensional descriptor. The SIFT descriptor (Figure 2.1, right) divides a square patch into a 4×4 grid and computes a histogram of gradient orientations in each subregion. Eight gradient orientations are used, resulting in a 128-dimensional feature vector. Histogramming provides stability against deformations of the image pattern, while subdividing the support region offsets the potential loss of spatial information. In this way, a compromise is achieved between the conflicting requirements of greater geometric invariance on the one hand and greater discriminative power on the other. Intuitively, descriptors based on this compromise should be simultaneously richer and more robust than filter banks and differential invariants, which are functions of the entire region of support. Indeed, in a recent comparative evaluation [106], SIFT descriptors and shape contexts decisively outperform these more traditional methods. In Section 3.1.1.2, we introduce two new distribution-based descriptors, *spin image* and *RIFT*, that are tailored to achieve greater geometric invariance and computational efficiency as opposed to greater distinctiveness.

2.2 Texture Recognition

The automated analysis of image textures has been the topic of extensive research in the past forty years, dating back at least to Julesz in 1962 [59]. Traditional techniques for modeling texture include co-occurrence statistics [49, 59], filter banks [97, 125], and random fields [99, 167]. One of the most important challenges in the field of texture analysis and recognition is achieving invariance to a wide range of geometric and photometric transformations. Early research in this domain has concentrated on global 2D image transformations, such as rotation and scaling [18, 99]. However, such models do not accurately capture the effects of 3D transformations (even in-plane rotations) of textured surfaces. More recently, there has been a great deal of interest in recognizing images of textured surfaces subjected to lighting and viewpoint changes [21, 22, 80, 87, 156, 157, 165]. A few methods [80, 87, 165] are based on explicit reasoning about the 3D structure of the surface, which may involve registering samples of a material so that the same pixel across different images corresponds to the same physical point on the surface [80] or applying photometric stereo to reconstruct the depth map of the 3D texture [87, 165]. While such approaches capture the appearance variations of 3D surfaces in a principled manner, they require specially calibrated datasets collected under controlled laboratory conditions. For example, the 3D texton representation of Leung and Malik [80] naturally lends itself to the task of classifying a "stack" of registered images of a test material with known imaging parameters, but its applicability is limited in most practical situations.

In our own work, we are interested in classifying *unregistered* texture images. This problem has been addressed by Cula and Dana [21] and Varma and Zisserman [156, 157], who have developed several dense 2D texton-based representations capable of very high accuracy on the challenging Columbia-Utrecht reflectance and texture (CUReT) database [22]. The descriptors used in these representations are filter bank outputs [21, 156] and raw pixel values [157]. Even though these methods have proven effective in the complex task of classifying images of materials despite significant appearance changes, the *representations* themselves are not invariant to the changes in question. In particular, the support regions for computing descriptors are the same in all images; no adaptation is performed to compensate for changes in surface orientation with respect to the camera.

Because they lack representation-level invariance, the above methods require the use of multiple models or prototypes to represent a single texture. As long as the training set adequately samples all viewpoints and lighting directions that can be encountered at test time, the texture can be successfully recognized. On the other hand, when the test set contains images not represented in the training set (e.g., images at significantly different scales), performance drops dramatically [156]. In Chapter 3, we show how to reduce this dependence on representative training sets by developing a texture representation with builtin geometric invariance.

2.3 Object Recognition

The earliest recognition systems [13, 88, 112, 127] were *model-based*, i.e., designed to perform localization and pose estimation from a single view of an object given a representation of its 3D shape (possibly as a combination of "parts" or geometric primitives such as generalized cylinders). This work has emphasized the relatively abstract issues of viewpointindependent representation of generic 3D shapes and formal rule-based geometric reasoning. While model-based vision systems represent important conceptual milestones in the field of recognition, their practical usefulness is severely limited by their reliance on relatively weak and uninformative image features such as line and curve segments and their lack of flexibility for modeling non-parametric deformations or complex intra-class variations (though some model-based approaches [170] have made a systematic attempt to represent deformable objects such as animals from their 2D silhouettes). In this dissertation, we follow instead an alternative *appearance-based* recognition paradigm, which emphasizes not precise 3D geometric description, but statistical modeling of 2D object apperance using highly expressive and discriminative image features.

Initial work on appearance-based object recognition has mainly utilized global descriptions: The raw images themselves can be subjected to eigenspace analysis [110, 152] or used as feature vectors for support vector machine classification [120]. Another class of early methods [113, 136] is based on characterizing the entire image using color or texture histograms. The main drawback of such methods is their lack of robustness to clutter and occlusion. For this reason, recognition methods that work by characterizing the entire image have been gradually supplanted over the last decade by *part-based* methods that seek to identify statistically or structurally significant "atoms" of object appearance. In the older model-based literature, a part was usually understood to be a 3D geometric primitive in some formal scheme for representing general shapes. By contrast, appearance-based approaches have a much more flexible image-based notion of a part. Schneiderman and Kanade [140] define parts as groups of highly correlated input variables (in their case, wavelet coefficients). Most other approaches obtain parts by sampling image fragments [93, 142, 154], or by detecting corner-like interest points [1, 162, 163] or scale-invariant salient regions [29, 33].

In the current literature, a popular object recognition paradigm is that of probabilistic *constellation* or *parts-and-shape* models that represent not only the statistics of individual parts, but also their spatial layout [14, 29, 33, 163]. The idea of such models goes back at least thirty years to the *pictorial structures* of Fischler and Elschlager [35]. Unfortunately, the level of geometric invariance of existing constellation models is limited to scale and translation, since a principled probabilistic treatment of rotation or affine invariance is prohibitively complex [78]. Learning and inference problems for these models become even

more intractable in a *weakly supervised* setting where the location of the object in a training image has not been marked by hand. This setting requires the use of the *Expectation Maximization* (EM) algorithm [8] to perform the assignment of object parts to image regions in the presence of occlusion and clutter. However, the combinatorics of EM-based learning severely limits both the expressiveness of the corresponding object models and the amount of clutter and/or occlusion that can be tolerated during learning.

In response to the above difficulties with parts-and-shape models, several vision researchers have proposed orderless *bag-of-features* models [20, 164, 143], which are obtained by extracting scale- or affine-invariant local features from images, quantizing them into "visual words," and learning the distributions of these words for each object class. This is essentially the same strategy that we use in designing the local texture model presented in Chapter 3. However, while an orderless representation is appropriate for texture images, which lack clutter and have uniform statistical properties, it is somewhat problematic as a method for representing object classes, since it ignores spatial relations and makes no distinction between features generated by the object and those generated by the background. In fact, bag-offeatures methods are prone to much the same pitfalls as earlier global and histogram-based recognition approaches. One way to increase the robustness of bag-of-features methods to clutter and occlusion is to use statistical techniques such as feature selection [24] or boosting [118] to retain only the most discriminative features for recognition. Another solution is to design novel kernels that can yield high discriminative power despite the noise and irrelevant information that may be present in local feature sets [45, 92, 161]. In Chapter 3, we follow this idea by introducing a kernel based on EMD, which is designed for partial matching, and therefore performs well in clutter and occlusion. With the addition of this kernel, our local texture representation becomes capable of achieving and even exceeding state-of-the-art classification results on multiple object databases.

Even though we can obtain good object recognition results using a completely orderless method, our ultimate goal in this disseration is to overcome the limitations of Bayesian partsand-shape models without sacrificing spatial relations. We can achieve this by adopting a non-probabilistic approach to learning and detecting object parts that is based on direct search for visual correspondence. In recent literature, the low-distortion correspondence method of Berg et al. [5] is an example of this philosophy. In Chapter 5 we present an object representation in which composite parts are found using alignment techniques [54]. Our *semi-local part* approach is capable of a higher degree of geometric invariance than existing constellation models, and, in combination with a discriminative maximum entropy framework (Section 4.1), enables the learning of relatively complex object models consisting of many features in heavily cluttered images.

2.4 Scene and Context Recognition

In the 1990s, several researchers have considered "semantic" scene categorization tasks such as distinguishing city views from landscapes [44, 155] and indoor from outdoor images [147]. Subsequent computational approaches to scene description and recognition have often drawn their inspiration from the literature on human perception. For example, it is known that people can recognize scenes by considering them in a "holistic" manner, without having to recognize individual objects [7]. Recently, it has been shown that human subjects can perform high-level categorization tasks extremely rapidly [148] and in the near absense of attention [31]. Renninger and Malik [126] propose an orderless bag-of-textons model to replicate human performance on rapid scene categorization tasks. Another perceptually inspired approach, due to Oliva and Torralba [117], computes a low-dimensional representation of a scene based on several global properties such as "openness" and "ruggedness." A few recent scene recognition approaches [91, 159] try to find effective intermediate representations in terms of basic natural texture categories, such as water, sky, sand, grass, foliage, etc. A major drawback of these methods is that these categories have to be learned in a fully supervised fashion, which requires human participants either to hand-segment the training images and label their constituent categories or to provide numerous sample patches of each category. Fei-Fei and Perona [30] present an alternative unsupervised approach that represents scenes as mixtures of a small number of "themes," or characteristic textures.

Several of the approaches listed above [44, 117, 147] are based on the observation that coarse spatial localization of image features carries discriminative information for the scene recognition task. They take advantage of this insight by partitioning the image into multiple subblocks and computing statistics of features in each subblock. A similar decomposition strategy is used by the *Viper* content-based image retrieval system developed by the University of Geneva [145], which subdivides the image into a two-level quadtree. Our *spatial pyramid* scene recognition method presented in Chapter 6 also uses a quadtree decomposition, but gives it a novel interpretation in terms of partial matching of two sets of features [45]. Just like its precursors, our method is *global* because the spatial image decomposition is not spatially invariant, and the image representation it induces may change depending on the absolute positions of the individual features. However, because of the considerable statistical regularities in the spatial layout of natural scenes, this sensitivity to position actually becomes a source of additional discriminative power.

It is important to note that global image representations are useful not only for scene classification considered as a goal in itself, but also for context recognition, as a prerequisite for object identification tasks. The "gist" of an image [111, 150], which is related to the global low-dimensional representation of Oliva and Torralba [117] may be used to inform the subsequent search for specific objects (e.g., if the image, based on its global description, is likely to be a highway, we have a high probability of finding a car, but not a toaster).

So far, we have focused on the problem of assigning a single *global* category label to the entire image. However, there exists a considerable body of work on labeling *local* image re-

gions according to high-level concepts. Some of the tasks considered in the literature include labeling of image regions into natural vs. manmade [11, 68], classifying them into one of several natural texture categories [69], or into one of several geometric classes determined by their surface orientations (e.g., horizontal, fronto-parallel, slanted, etc.) [53]. The latter approach extracts 3D information about the scene that can be used as a type of context for predicting likely positions and scales of objects of interest in the image. In this dissertation, we also address the labeling problem, albeit somewhat restrictively formulated as classification of individual local features in multi-texture images (Section 4.2). One advantage of our approach is that, unlike all the labeling methods listed above, it works in a *weakly supervised* setting, i.e., it can learn multiple texture categories without requiring segmented images or samples of individual textures.

CHAPTER 3

A Bag-of-Features Model for Texture and Object Recognition

This chapter presents an orderless bag-of-features representation for texture and object classes that works by characterizing the distribution of appearance descriptors of local features in the image. The first part of this chapter, Section 3.1, describes a texture recognition method that relies on local affine-invariant features to achieve robustness to significant rotations, scale changes, perspective distortions, and non-rigid deformations. This method combines multiple local feature detectors and descriptors in a nearest-neighbor classification framework that relies on Earth Mover's Distance (EMD) to compare distributions. Experimental results presented in Section 4.2.2 show promising performance on two texture databases. This texture recognition approach has been published in [72, 75].

The second part of this chapter, Section 3.2, presents an in-depth evaluation that extends our original work in several ways. First, we augment the classification framework with a Support Vector Machine kernel, which greatly improves performance over the original nearest-neighbor method and increases the effectiveness of combining multiple feature channels. Second, we consider different levels of invariance of local features, and show that the best level of invariance depends on the given task. Third, we perform a comparative evaluation with state-of-the-art methods on three texture and three object datasets. This evaluation was performed in collaboration with Jianguo Zhang and Marcin Marszalek at INRIA Rhône-Alpes and has appeared in [168].

3.1 Texture Recognition Using Local Affine Regions

This section presents a texture recognition method that is invariant to geometric transformations that can be *locally* approximated by an affine model. Since sufficiently small patches on the surfaces of smooth 3D objects are always approximately planar, local affine invariants are appropriate for modeling not only global 2D affine transformations of the image, but also perspective distortions that arise in imaging a planar textured surface, as well as non-rigid deformations that preserve the locally flat structure of the surface, such as the bending of paper or cloth. Like other approaches based on *textons*, or primitive texture elements [21, 80, 156, 157], our method involves representing distributions of 2D image features; unlike these, however, it performs shape selection, ensuring that descriptors are computed over neighborhoods whose shape is adapted to changes in surface orientation and scale caused by camera movements or scene deformations. In addition, our method performs spatial selection by computing descriptors at a sparse set of image locations output by local affine region detectors. This is a significant departure from the traditional feature extraction framework, which involves processing every pixel location in the image. Apart from being memory- and computation-intensive, this *dense* approach produces redundant texton dictionaries that may include, for instance, many slightly shifted versions of the same basic element [96]. As Figure 3.1 illustrates, spatial selection is an effective way to reduce this redundancy.

Our approach consists of the following steps (see also Figure 3.2):

- 1. Extract a sparse set of *affine-invariant regions* (or *affine regions* for short) in the shape of ellipses from a texture image. The two region detectors used for this purpose are described in Section 3.1.1.1.
- 2. Normalize the shape of each elliptical region by transforming it into a circle. This reduces the affine ambiguity to a rotational one. Full affine invariance is achieved



Figure 3.1 The effect of spatial selection on a texton dictionary. (a) Original texture image. (b) Top 20 textons found by clustering all 13×13 patches of the image. (c) A sparse set of regions found by the Laplacian detector described in Section 3.1.1.1. Each region is normalized to yield a 13×13 patch. (d) Textons obtained by clustering the normalized patches. For the sake of this illustration, we disregard the orthogonal ambiguity inherent in the normalization process (see Section 3.1.1.1 for details). Because we are clustering the normalized patches themselves, instead of rotation-invariant descriptors as in Section 3.1.1.2, the resulting description of patch appearance in this case is rotation-dependent. This can be seen from the fact that the second and third clusters of (d) are rotated versions of each other.

by computing rotation-invariant descriptors over the normalized regions. In Section 3.1.1.2, we introduce two rotation-invariant descriptors: one based on *spin images* used for matching range data [58], and one based on Lowe's *SIFT descriptor* [90].

- 3. Perform clustering on the affine-invariant descriptors to obtain a more compact representation of the distribution of features in each image (Section 3.1.1.3). Summarize this distribution in the form of a *signature*, containing a representative descriptor from each cluster and a weight indicating the relative size of the cluster.
- 4. Compare signatures of different images using the Earth Mover's Distance (EMD) [131, 132], which is a convenient and effective dissimilarity measure applicable to many types of image information. The output of this stage is an *EMD matrix* whose entries record the distances between each pair of signatures in the database. The EMD matrix can be used for retrieval and classification tasks, as described in Section 3.1.2.1.



Figure 3.2 The architecture of the feature extraction system proposed in this chapter.

In Section 3.1.2, we will use two datasets to evaluate the capabilities of the proposed texture representation. The first dataset, introduced in Section 3.1.2.2, consists of photographs of textured surfaces taken from different viewpoints and featuring large scale changes, perspective distortions, and non-rigid transformations. It must be noted that, even though our method relies on implicit assumptions of local flatness and Lambertian appearance, and is thus theoretically applicable primarily to "albedo textures" due to spatial albedo variations on smooth surfaces, the results in Section 3.1.2.2 show that in practice it also tends to perform well on "3D textures" arising from local relief variations on a surface. Our second set of experiments, described in Section 3.1.2.3, is carried out on the Brodatz database [12], a collection of images that contains many diverse texture classes, but there are no geometric transformations between members of the same class. Because affine invariance is not required in this case, we modify the basic feature extraction framework to use neighborhood shape as a discriminative feature to improve performance.

Components	Previous work [21, 80, 156, 157]	This chapter
Spatial selection	None: every pixel is considered	Laplacian and Harris affine region detectors [40, 104]
Neighborhood shape selection	None: neighborhood size is fixed	Affine adaptation process [40]
Descriptor computation	Filter banks [21, 80, 156], pixel values [157]	Distribution-based descriptors: spin images, SIFT, RIFT
Finding textons	Clustering, universal texton dic- tionaries	Clustering, separate texton representation for each image
Representing/comparing tex- ton distributions	Histograms/ χ^2 distance	Signatures/Earth Mover's Dis- tance [132, 131]

Table 3.1 The components of our approach, contrasted with other 2D texton-based methods.

Table 3.1 summarizes the main components of our approach and contrasts it with existing 2D texton-based methods [21, 80, 156, 157]. Because they do not use geometrically invariant features, these methods do not perform well when the test set contains geometric deformations not represented in the training set. In our work, we shift away from this dependence on representative training sets by developing a texture representation with built-in geometric invariance. Note, however, that our present method does not explicitly account for changes in lighting direction and associated 3D effects such as self-shadowing; therefore, to achieve robust recognition in the presence of such effects, we must still rely on multiple prototypes in the training set.

3.1.1 Components of the Approach

3.1.1.1 Affine Region Detection

In this work, we use two types of detectors: the Harris-affine detector of Mikolajczyk and Schmid [104] and the Laplacian blob detector of Gårding and Lindeberg [40]. Refer back to Figure 1.2 for examples of the two kinds of regions extracted from two texture images. The Harris detector tends to find corners and points at which significant intensity changes occur, while the Laplacian detector is (in general) attracted to points that can be thought of as centers of roughly elliptical regions of uniform intensity. Intuitively, the two
detectors provide complementary kinds of information about the image: The former responds to regions of "high information content" [104], while the latter produces a perceptually plausible decomposition of the image into a set of blob-like primitives.

The technical details of the implementation of Laplacian and Harris regions can be found in [40, 82, 104] and are summarized in the Appendix for completeness. For the purposes of this chapter, it is sufficient to note that the affine regions localized by these detectors are represented as ellipses. We can *normalize* these regions by mapping the corresponding ellipses onto a unit circle. Because the circle is invariant under rotations and reflections, it can be easily shown that the normalization process has an inherent orthogonal ambiguity. In some existing work on wide-baseline matching, this ambiguity is resolved by estimating a dominant gradient direction of the patch and aligning this direction with the positive xaxis [90, 104]. We will use this strategy in Section 3.2, but for the experiments presented in Section 3.1.2, we omit this step and eliminate the rotational ambiguity by representing each normalized patch by a rotationally invariant descriptor — a strategy similar to [3, 135].

Note. To achieve invariance to local affine transformations, as in the experiments of Section 3.1.2.2, we discard the information contained in the affine shape of the patches. However, as a glance at Figure 1.2 suggests, this shape can be a distinctive feature when affine invariance is not required. This point will be revisited in Section 3.1.2.3.

3.1.1.2 Rotation-Invariant Descriptors

This section introduces two novel rotation-invariant descriptors used in the experiments of this chapter: *intensity-domain spin images*, inspired by the method for matching range data developed by Johnson and Hebert [58]; and *RIFT* descriptors, based on the *Scale-Invariant Feature Transform (SIFT)* developed by Lowe [90].

Intensity-domain spin images. Our first rotation-invariant descriptor is inspired by the *spin images* introduced by Johnson and Hebert [58] for matching range data. The *intensity-domain* spin image proposed in this chapter is a two-dimensional histogram encoding the

distribution of image brightness values in the neighborhood of a particular reference (center) point. The two dimensions of the histogram are d, distance from the center point, and i, the intensity value. The "slice" of the spin image corresponding to a fixed d is simply the histogram of the intensity values of pixels located at a distance d from the center. Since the d and i parameters are invariant under orthogonal transformations of the image neighborhood, spin images offer an appropriate degree of invariance for representing affinenormalized patches. In the experiments reported in Section 3.1.2, we used 10 bins for distance and 10 for intensity value, resulting in 100-dimensional descriptors.

We implement the spin image as a "soft histogram" where each pixel within the support region contributes to more than one bin. Specifically, the contribution of a pixel located in x to the bin indexed by (d, i) is given by

$$\exp\left(-\frac{(|x-x_0|-d)^2}{2\alpha^2} - \frac{|I(x)-i|^2}{2\beta^2}\right)\,,$$

where x_0 is the location of the center pixel, and α and β are the parameters representing the "soft width" of the two-dimensional histogram bin. Note that the soft histogram can be seen as a set of samples from the Parzen estimate (with Gaussian windows) of the joint density of intensity values *i* and distances *d*. The use of soft histograms has also been advocated by Koenderink and Van Doorn [66] because it alleviates aliasing effects. Figure 3.3 illustrates the principle behind the construction of spin images.

To achieve invariance to affine transformations of the image intensity function (that is, transformations of the form $I \mapsto aI + b$), it is sufficient to normalize the range of the intensity function within the support region of the spin image [134]. To alleviate the potential sensitivity of the normalization to noise and resampling artifacts (these are particularly severe for patches that are only a few pixels wide), we slightly blur the normalized patches with a Gaussian kernel before computing the spin image.

RIFT descriptors. To obtain a complementary representation of local appearance of normalized patches, we have developed an additional rotation-invariant descriptor that gen-



Figure 3.3 Construction of spin images. Three sample points in the normalized patch (left) map to three different locations in the descriptor (right).

eralizes Lowe's SIFT [90], which has been noted for its superior performance in retrieval tasks [106]. Recall from Section 2.1 that SIFT is based on subdividing a square image patch into a 4×4 pattern of smaller squares and computing eight-dimensional gradient orientation histograms inside each of these, yielding a 128-dimensional feature vector. To achieve rotation invariance, we must use concentric rings instead of squares, which produces a descriptor of lower dimensionality. Our descriptor, dubbed *Rotation-Invariant Feature Transform*, or *RIFT*, is constructed by dividing the circular normalized patch into four concentric rings of equal width and computing orientation histograms within each ring (Figure 3.4). The dimensionality of the resulting features is $4 \times 8 = 32$. In order to maintain rotation invariance, we must measure the gradient orientation at each point relative to the direction pointing outward from the center. Note that the RIFT descriptor as described above is not invariant to flipping of the normalized patch, which reverses the order of directions in the orientation histogram. However, we are not concerned with this circumstance in our current work, since realistic imaging conditions do not involve reversing the orientation of a textured surface.



Figure 3.4 Construction of RIFT descriptors. Three sample points in the normalized patch (left) map to three different locations in the descriptor (right).

3.1.1.3 Signatures and the Earth Mover's Distance

One commonly thinks of a texture image as being "generated" by a few basic primitives, or textons [60], repeated many times and arranged in some regular or stochastic spatial pattern. In the field of texture analysis, clustering is the standard technique for discovering a small set of primitives based on a large initial collection of texture element instances. Accordingly, we perform clustering on each texture image separately to form its *signature* $\{(m_1, u_1), (m_2, u_2), \ldots, (m_k, u_k)\}$, where k is the number of clusters, m_i is the center of the *i*th cluster, and u_i is the relative weight of the cluster (the size of the cluster divided by the total number of descriptors extracted from the image). To compare two signatures $S_1 = \{(m_1, u_1), (m_2, u_2), \ldots, (m_k, u_k)\}$ and $S_2 = \{(n_1, v_1), (n_2, v_2), \ldots, (n_l, v_l)\}$, we compute their *Earth Mover's Distance* (EMD) [131, 132], which is a cross-bin dissimilarity measure that can handle unequal signature lengths:

$$\text{EMD}(S_1, S_2) = \frac{\sum_i \sum_j f_{ij} d(m_i, n_j)}{\sum_i \sum_j f_{ij}}$$

where the scalars f_{ij} are flow values that are determined by solving a linear programming problem, and the scalars $d(m_i, n_j)$ are the ground distances between different cluster centers. The theoretical justification of this formula and the specifics of the optimization setup are beyond the scope of this dissertation; we refer the interested reader to [81, 132] for more details. In our case, m_i and n_j may be spin images and RIFT descriptors, and the ground distance is simply the Euclidean distance. Since our descriptors are normalized to have unit norm, the ground distances lie in the range [0, 2]. We rescale this range to [0, 1], thus ensuring that all EMD's are between 0 and 1 as well.

An alternative to our signature/EMD framework is given by the histogram/ χ^2 distance framework used in many other texture recognition approaches [21, 80, 156, 157]. In this framework, a global *texton vocabulary* (or *visual vocabulary*) is obtained by clustering descriptors from a special training set, and then each image is represented as a histogram of texton labels. Given a global texton vocabulary of size k, the *i*th entry of a histogram is the proportion of all descriptors in the image having label *i*. To compare two histograms $H_1 = (u_1, \ldots, u_k)$ and $H_2 = (v_1, \ldots, v_k)$, we use the χ^2 distance [121] defined as

$$\chi^2(H_1, H_2) = \frac{1}{2} \sum_{i=1}^k \frac{(u_i - v_i)^2}{u_i + v_i}.$$

Empirically, the signature/EMD and the histogram/ χ^2 setups produce very similar performance in practice, provided that the descriptive power of the global texture vocabulary is equivalent to that of the local signatures [168]. However, in our implementation, we favor signatures with EMD, primarily because of the considerable computational expense involved in computing a global vocabulary (see [168] for a detailed evaluation of running times). Moreover, the use of signatures with EMD has some important theoretical advantages over the use of histograms with χ^2 distance. In particular, a signature is more descriptive than a histogram, and it avoids the quantization and binning problems associated with histograms, especially in high dimensions [132] (recall that our spin images and RIFT descriptors are 100and 32-dimensional, respectively, and the SIFT descriptor used in the experiments of Section 3.2 is 128-dimensional). One advantage of EMD over χ^2 distance is its relative insensitivity to the number of clusters, i.e., when one of the clusters is split during signature computation, replacing a single center with two, the resulting EMD matrix is not much affected [131]. This is a very desirable property, since automatic selection of the number of clusters remains an unsolved problem.

3.1.2 Experimental Evaluation

3.1.2.1 Evaluation Strategy

Channels. Tuytelaars and Van Gool [153] have articulated the goal of building an opportunistic neighborhood extraction system that would combine the output of several region detectors tuned to different kinds of image structure. In this spirit, the texture representation proposed in this chapter is designed to support multiple region detectors and descriptors. Each detector/descriptor pair is treated as an independent *channel* that generates its own signature representation for each image in the database, and its own EMD matrix of pairwise inter-image distances. To combine the outputs of several channels, we simply add the corresponding entries in the EMD matrices. This approach was empirically determined to be superior to forming linear combinations with varying weights, or taking the minimum or maximum of the distances.

Since our experimental setup involves the evaluation of two region detectors and two descriptors, we end up with four channels: Harris regions and spin images (H/SPIN), Harris regions and RIFT descriptors (H/RIFT), Laplacian regions and spin images (L/SPIN), and finally, Laplacian regions and RIFT descriptors (L/RIFT). In addition, we will introduce in Section 3.1.2.3 the Harris and Laplacian *ellipse channels*, denoted H/ELL and L/ELL, respectively. To simplify the notation for combined channels, we will use (in a purely formal,

"syntactic" manner) the distributive law: For example, we will write (H+L)(RIFT) for the combination of the Harris/RIFT and Laplacian/RIFT channels, and (H+L)(SPIN+RIFT) for the combination of all four detector/descriptor channels.

Retrieval. We use the standard procedure followed by several Brodatz database evaluations [84, 119, 166]. Given a query image, we select other images from our database in increasing order of EMD, i.e., from the most similar to the least similar. Each image in the database is used as a query image once, and the performance is summarized as a plot of average recall vs. the number of retrievals. Average recall is defined as the number of images retrieved from the same class as the query image over the total number of images in that class (minus one to account for the query itself), averaged over all the queries. For example, perfect performance for a given class would correspond to average recall of 100% after n-1retrievals, where n is the number of images in that class.

Classification. In effect, the evaluation framework described above measures how well each texture class can be modeled by individual samples. It is not surprising that retrieval can fail in the presence of sources of variability that are not fully accounted for by the invariance properties of the representation (recall that our representation provides invariance to local geometric deformations and affine illumination changes, but not to complex viewpoint- and lighting-dependent appearance changes). To obtain a more balanced assessment of performance, a texture representation should be evaluated using classification as well as retrieval. In the classification framework, a model for a class is created not from a single (possibly atypical) image, but from a set of multiple training images, thereby compensating for the effects of intra-class variability.

For the experiments presented in this section, we use nearest-neighbor classification with EMD. The training set is selected as a fixed-size random subset of the class, and all remaining images comprise the test set. To eliminate the dependence of the results on the particular training images used, we report the average of the classification rates obtained for different randomly selected training sets. More specifically, a single sequence of 200 random subsets is generated and used to evaluate all the channel combinations seen in Tables 3.2 and 3.4. This ensures that all the rates are directly comparable, i.e., small differences in performance cannot be attributed to random "jitter."

3.1.2.2 UIUC Texture Database

To test the invariance properties of our proposed representation, we have collected a texture database consisting of 1000 uncalibrated, unregistered images: 40 samples each of 25 textures. Figure 3.5 shows four sample images from each class (the resolution of the samples is 640 × 480 pixels). The database includes surfaces whose texture is due to albedo variations (e.g., wood and marble), 3D shape (e.g., gravel and fur), as well as a mixture of both (e.g., carpet and brick). Significant viewpoint changes and scale differences are present within each class, and illumination conditions are uncontrolled. During data acquisition, we have taken care to exercise additional sources of variability wherever possible. These include non-planarity of the textured surface (bark), significant non-rigid deformations between different samples of the same class (fur, fabric, and water), inhomogeneities of the texture pattern (bark, wood, and marble), and viewpoint-dependent appearance variations (glass).

Each image in the database is processed with the Harris and Laplacian detectors. The median number of Harris (resp. Laplacian) regions extracted per image is 926 (resp. 4591). The median number of combined regions is 5553, or about 1.8% of the total number of pixel locations in the image. Thus, we can see that the spatial selection performed by the detectors results in a drastic compression of the amount of data that needs to be handled by the subsequent processing stages, especially clustering, which is a notoriously memory-intensive operation. In our implementation, clustering was performed using the k-means algorithm with k = 40 centers.



Figure 3.5 The UIUC texture database: Four samples each of the 25 classes used in the experiments of Section 3.1.2.2. The entire database may be downloaded from http://www-cvr.ai.uiuc.edu/ponce_grp.



Figure 3.6 Retrieval curves for the texture database. (a) The Harris channels. (b) The Laplacian channels. (c) The combined Harris and Laplacian channels.

Figure 3.6 shows retrieval results for the texture database. First, spin images perform better than RIFT descriptors, and the combination of the two descriptors performs slightly better than spin images alone. Next, Laplacian regions (part (b) of the figure) perform better than Harris (a), and the combination of the two (c) is slightly better than Laplacian alone. The solid curve in part (c) of the figure shows the retrieval performance obtained by combining all four detector/descriptor channels. The recall after 39 retrievals is 58.92%. This relatively low number is a reflection of the considerable intra-class variability of the database. As discussed in Section 3.1.2.1, we cannot expect that all samples of the same class will be well represented by a single prototype. Accordingly, the combined (H+L)(S+R) classification rate is only 62.15% for one training sample, but it goes up rapidly to 92.61% for 10 samples and 96.03% for 20 samples. Table 3.2 shows a comparison of 10-sample classification rates for different channel combinations. The same trends that were seen in Figure 3.6 are echoed here: Spin images perform better than RIFT, Laplacian regions perform better than Harris, and the combination (H+L)(S+R) has the highest performance.

We may wonder whether the superior performance of Laplacian points is due to their higher density (recall that the Laplacian detector finds almost five times as many regions as the Harris). To check this conjecture, we have repeated the recognition experiments after thresholding the output of the Laplacian detector so that equal numbers of Laplacian

	Н	L	H+L
SPIN	83.32	88.29	90.15
RIFT	79.27	83.18	85.47
SPIN+RIFT	88.14	91.96	92.61

Table 3.2 Classification results for 10 training samples per class. First column (top to bottom): H/SPIN, H/RIFT, H(SPIN+RIFT). Second column: L/SPIN, L/RIFT, L(SPIN+RIFT). Third column: (H+L)SPIN, (H+L)RIFT, (H+L)(SPIN+RIFT).

and Harris regions are produced for each image. The results of the "truncated" and "full" Laplacian representations may be compared by looking at columns 3 and 4 of Table 3.3. Interestingly, while the rates may vary significantly for individual textures, the averages (bottom row) are almost the same: 91.93% and 91.96% for "truncated" and "full", respectively. Thus, recognition performance is not a simple function of representation density.

Finally, Table 3.3 allows us to analyze the "difficulty" of individual textures for our system. To this end, the textures are arranged in order of increasing (H+L)(SPIN+RIFT) classification rate (last column). Roughly speaking, classification rate is positively correlated with the homogeneity of the texture: Some of the lowest rates belong to inhomogeneous coarse-scale textures like bark (T02, T03) and marble (T09), while some of the highest belong to homogeneous fine-scale textures like glass (T17), water (T07), and fabric (T20, T24). However, this relationship is not universal. For example, granite (T08), which is fine-grained and quite uniform, has a relatively low rate of 86.78%, while the large-scale, inhomogeneous wall (T13) has a relatively high rate of 95.92%. It is also interesting (and somewhat unexpected) that the performance of different classes does not depend on their nature as 3D or albedo textures. Ultimately, the intrinsic characteristics of the various textures do not provide a clear pattern for predicting performance because classification accuracy is not related directly to intra-class variability, but to the extent of separation between the classes in feature space.

	Class	H(SPIN+RIFT)	L(SPIN+RIFT)	L(SPIN+RIFT)	(H+L)
			(trunc.)	(full)	(SPIN+RIFT)
	T03 (bark)	74.55	72.48	75.12	79.53
	T19 (carpet)	72.70	85.92	82.07	81.07
	T02 (bark)	80.77	81.67	80.18	84.67
	T08 (granite)	83.52	88.55	88.08	86.78
· J	T09 (marble)	75.15	83.60	89.50	87.73
	T21 (paper)	74.45	93.35	92.30	88.80
	T16 (glass)	76.10	97.53	93.67	88.82
	T12 (gravel)	79.47	87.03	90.73	90.12
	T14 (brick)	83.07	90.48	89.65	90.28
1 H	T01 (bark)	89.72	92.95	87.03	90.63
	T23 (fabric)	88.98	95.12	90.82	91.98
	T15 (brick)	89.87	90.35	90.67	92.38
10	T11 $(stone)$	89.20	94.48	94.88	93.72
	T05 (wood)	89.83	81.63	92.50	94.42
5.1 5.1	T10 $(stone)$	85.00	94.88	96.87	94.92
	T22 (fur)	94.53	94.23	92.08	95.08
	T25 (fabric)	93.30	95.90	93.43	95.90
	T13 (wall)	92.88	95.08	95.90	95.92
	T06 (wood)	96.90	95.35	92.90	97.43
	T18 (carpet)	96.60	92.82	98.00	97.47
5	T04 (wood)	98.68	98.62	96.18	98.53
* * * *	T20 (fabric)	99.08	97.02	99.73	99.48
	T07 (water)	99.80	99.40	99.05	99.75
	T24 (fabric)	99.37	99.90	97.80	99.80
	T17 (glass)	100	100	99.85	100
	Mean	88.14	91.93	91.96	92.61

Table 3.3 Detailed breakdown of classification results summarized in Table 3.2. The classes are sorted in order of increasing classification rate (last column). See text for discussion.

3.1.2.3 Brodatz Database



Figure 3.7 Examples of three images from the Brodatz database (there are 111 images total). Each image is divided into 9 non-overlapping sub-images for experiments.

The Brodatz database [12] (Figure 3.7) is perhaps the best known benchmark for texture recognition algorithms. In recent years, it has been criticized because of the lack of intra-class variation that it exhibits. However, we feel that it is premature to dismiss the Brodatz database as a challenging platform for performance analysis. For one, relatively few publications actually report results on the entire database (the only studies known to us are [42, 84, 119, 166]). In addition, while near-perfect overall results have been shown for the CUReT database [157], the best (to our knowledge) retrieval performance on the Brodatz database is around 84% [166]. The reason for this is the impressive diversity of Brodatz textures, some of which are quite perceptually similar, while others are so inhomogeneous that a human observer would arguably be unable to group their samples "correctly". The variety of the scales and geometric patterns of the Brodatz textures, combined with an absence of intra-class transformations, makes them a good platform for testing the discriminative power of an additional *local shape* channel in a context where affine invariance is not necessary, as described below.

The shape channel. The shape of an affinely adapted region is encoded in its *local shape* matrix, which can also be thought of as the equation of an ellipse. Let E_1 and E_2 be two

ellipses in the image plane. We eliminate the translation between E_1 and E_2 by aligning their centers, and then compute the dissimilarity between the regions as

$$d(E_1, E_2) = 1 - \frac{\operatorname{Area}\left(E_1 \cap E_2\right)}{\operatorname{Area}\left(E_1 \cup E_2\right)}.$$

In the experiments of this section, we use local shape to obtain two additional channels, HE and LE, corresponding to the ellipses found by the Harris and Laplacian detectors, respectively. Notice that the ellipse ground distance, and consequently all shape-based EMD's, must be between 0 and 1. Because the descriptor-based EMD's lie in the same range, the shape-based EMD's can be combined with them through simple addition.

Finally, it is worth noting that the ellipse "distance" as defined above takes into account the relative orientations of the two ellipses. If it is necessary to achieve rotation invariance, we can simply align the major and minor axes of the two ellipses before comparing their areas.

Results. The Brodatz database consists of 111 images. Following the same procedure as previous evaluations [84, 119, 166], we form classes by partitioning each image into nine nonoverlapping fragments, for a total of 999 images. Fragment resolution is 215×215 pixels. By comparison with the texture database discussed in the previous section, relatively few regions are extracted from each image: The median values are 132 for the Harris detector, 681 for the Laplacian, 838 combined. Some images contain less than 50 regions total. Given such small numbers, it is difficult to select a fixed number of clusters suitable for all images in the database. To cope with this problem, we replace k-means by an agglomerative clustering algorithm that repeatedly merges clusters until the average intra-cluster distance exceeds a specified threshold [63]. This process results in variable-sized signatures, which are successfully handled by the EMD framework. An additional advantage of agglomerative clustering as opposed to k-means is that it can be used for the shape channel, since it can take as input the matrix of pairwise distances between the ellipses.



Figure 3.8 Retrieval curves for the Brodatz database. (a) Harris descriptor channels. (b) Laplacian descriptor channels. (c) Combined Harris and Laplacian descriptor channels. (d) Comparison of combined performance with and without the ellipse channels (H/ELL and L/ELL).

Figure 3.8 shows retrieval results for the Brodatz database. Similarly to the results of the previous section, the Laplacian channels, shown in part (b) of the figure, have better performance than the Harris channels, shown in part (a). Interestingly, though, for the Brodatz database RIFT descriptors perform better than spin images — the opposite of what we have found in Section 3.1.2.2. However, this discrepancy is due at least in part to the variability of signature size (due to the use of agglomerative clustering) in the experimental setup of this section. On average, the RIFT-based signatures of the Brodatz images have more clusters than the spin-based signatures, and we conjecture that this raises the discriminative power of the RIFT channel. Another interesting point is that combining the Harris and Laplacian channels, as shown in (c), results in a slight drop of performance as compared to the Laplacian channels alone. Finally, (d) shows the effect of adding the shape channels into the mix. By themselves, these channels are relatively weak, since after 8 retrievals the (H+L)E recall is only 44.59%. However, adding these channels to (H+L)(SPIN+RIFT) boosts the recall from 70.94% to 76.26%.

The trends noted above are also apparent in the classification results presented in Table 3.4. By looking at the first two rows, we can easily confirm the relatively strong performance of the RIFT descriptor (particularly for the Laplacian detector), as well as the marginal drop

in performance of the L+H channels as compared to L alone. The latter effect is also seen in the last row of the table, where the (H+L)(SPIN+RIFT+ELL) classification rate is slightly inferior to the L(SPIN+RIFT+ELL) rate.

	Н	L	H+L
SPIN	61.36	75.70	75.31
RIFT	60.00	80.23	76.40
ELL	36.78	49.32	54.13
SPIN+RIFT+ELL	77.61	88.15	87.44

Table 3.4 Brodatz database classification results for 3 training samples.

We can get a more detailed look at the performance of our system by examining Figure 3.9, which shows a histogram of classification rates for all 111 classes using three training samples per class. The histogram reveals that the majority of textures are highly distinguishable, and only a few stragglers are located at the low end of the spectrum. In fact, 36 classes have 100% classification rate, 49 classes have classification rate at least 99%, and 73 classes (almost two thirds of the total number of classes) have rate at least 90%. The mean rate is 87.44%. Figure 3.10 shows four textures that were classified successfully and four textures that were classified unsuccessfully. Not surprisingly, the latter examples are highly non-homogeneous.

The best retrieval performance curve of our system, corresponding to the (H+L)(SPIN+ RIFT+ELL) combination, has 76.26% recall after 8 retrievals. This is a slightly higher than the results reported in [84, 119], but below Xu et al. [166], who report 84% recall using the multiresolution simultaneous autoregressive (MRSAR) model. MRSAR models texture as a stationary random field and uses a dense representation with fixed neighborhood shape and size. A known shortcoming of MRSAR is its limited ability to measure perceptual similarity — the method tends to confuse textures that appear very different to human observers [84]. Most significantly, the MRSAR model is difficult to extend with affine invariance. By con-



Figure 3.9 Histogram of classification rates for 3 training samples.



Figure 3.10 Left: successes. Right: failures. The average classification rates are shown below the corresponding class labels. Note that the representation for D48 has an average number of only 27 combined Laplacian and Harris regions per sample.

trast, our representation is specifically formulated with geometric invariance in mind and is non-parametric.

3.2 An Extended Evaluation: From Texture to Object Recognition

The previous section has demonstrated the promise of our method for visual classification tasks in the presence of significant geometric changes and appearance variations; however, we have not yet explored alternative implementation choices (including the level of invariance of the feature detector, the type of descriptor, or the classifier), made direct comparisons with other methods in the literature, or examined the full range of imagery to which our method may be applicable. In particular, because a bag of features has rather limited descriptive power, we have originally intended our approach exclusively for the application of texture recognition. However, the appearance in recent literature of several effective orderless local methods for object recognition tasks [20, 45, 143, 164] has subsequently motivated us to evaluate our approach for this application as well. This section discusses an extended evaluation of our method conducted in collaboration with Jianguo Zhang and Marcin Marszalek at IN-RIA Rhône-Alpes, who have contributed the experimental results presented in the following. Our study, which has been published in [168], addresses three issues:

Classification framework. In Section 3.1, we have used the simplest possible classification framework to demonstrate the power of the image representation alone. However, we show in Section 3.2.1 that EMD combined with an SVM kernel greatly improves performance. In particular, it improves the effectiveness of combining multiple feature channels. In the experiments of the previous section, we have occasionally observed a decrease in performance after combining several channels with differential discriminative power. However, as demonstrated by additional experiments of Section 3.2.2, with the addition of a discriminative classifier, this undesirable effect is greatly reduced.

Level of invariance. In the experiments of Section 3.1.2.2, we have shown that affineinvariant features produce good results in the presence of significant viewpoint and scale changes. However, we have not determined what is the *minimal* level of invariance required for effective performance on similar tasks. To answer this question, in Section 3.2.2 we evaluate the performance of features with different levels of invariance, and compare our approach with other non-invariant dense texture recognition methods. For many texture and object datasets currently available, we show that features with the highest possible level of invariance do not always yield the best performance. Thus, in attempting to design the most effective recognition system for a practical application, one should seek to incorporate multiple types of complementary features, but make sure that their local invariance properties do not exceed the level absolutely required for a given application.

Comparison with existing methods. We conduct a comparative evaluation with several state-of-the-art methods for texture and object classification on three texture and three object databases. For texture classification (Section 3.2.2), our approach outperforms existing methods on Brodatz and UIUC datasets, and obtains comparable results on the CUReT dataset [22]. For object category classification (Section 3.2.3), our approach outperforms previous methods on the Caltech6 [33], Caltech101 [29], and Graz [118] databases. Note that the power of orderless bag-of-keypoints representations is not particularly surprising in the case of texture images, which lack clutter and have uniform statistical properties. However, it is not *a priori* obvious that such representations are sufficient for object category classification, since they ignore spatial relations and fail to separate foreground from background features.

3.2.1 Kernel-based classification

Recently, *Support Vector Machine* (SVM) classifiers [141] have shown their promise for visual classification tasks (see [120] for an early example), and the development of specialized kernels suitable for use with local features has emerged as a fruitful line of research [15, 25, 45, 92, 116, 161]. To date, most evaluations of methods combining kernels and local features have been small-scale and limited to one or two datasets. This motivates us to build an effective image classification method combining a bag-of-keypoints representation with a kernel-based learning method and to test the limits of its performance on the most challenging databases available today.

Let us briefly review the basics of SVM classification. For a two-class problem, the SVM decision function for a test sample x has the following form:

$$g(x) = \sum_{i} \alpha_i y_i K(x_i, x) - b, \qquad (3.1)$$

where $K(x_i, x)$ is the value of a kernel function for the training sample x_i and the test sample x, y_i the class label of x_i (+1 or -1), α_i the learned weight of the training sample x_i , and b is a learned threshold parameter. The training samples with weight $\alpha_i > 0$ are usually called support vectors. In the following, we use the two-class setting for binary detection, i.e., classifying images as containing or not a given object class. To obtain a detector response, we use the raw output of the SVM, given by Eq. (3.1). By placing different thresholds on this output, we vary the decision function to obtain Receiver Operating Characteristic (ROC) curves (plots of true positive rates vs. false positive rates). For multi-class classification, different methods to extend binary SVMs tend to perform similarly in practice [141]. We use the one-against-one technique, which trains a classifier for each possible pair of classes. When classifying a pattern, we evaluate all binary classifiers, and classify according to which of the classes gets the highest number of votes. A vote for a given class is defined as a classifier putting the pattern into that class. Experimental comparisons on our dataset confirm that the one-against-one and one-against-other techniques give almost the same results.

To incorporate EMD into the SVM classification framework, we use an extended Gaussian kernel [16, 57]:

$$K(S_i, S_j) = \exp\left(-\frac{1}{A}D(S_i, S_j)\right), \qquad (3.2)$$

where $D(S_i, S_j)$ is EMD between two signatures S_i and S_j . The parameter A of the EMD kernel is the mean value of pairwise distances between all training images. To combine different channels, we simply sum their distances as in Section 3.1 and then apply the generalized Gaussian kernel to the combined distance.

We do not know whether the EMD kernel satisfies the Mercer condition, i.e., whether it corresponds to an inner product in some higher-dimensional space obtained by transforming the original feature space with some "lifting" function (see [141] for details about this condition and its significance for SVM classifiers). However, in our experiments, this kernel has always yielded positive definite Gram matrices (i.e., matrices of kernel values for all pairs of training feature vectors). In addition, it must be noted that even non-Mercer kernels often work well in real applications [16]. In contrast with some alternative kernels, such as the ones based on the Kullback-Leibler divergence [109], the EMD kernel does not require the estimation of distribution parameters using a time-consuming EM/MCMC algorithm as in [109]. Finally, we must mention that our approach is related to that of Grauman and Darrell [45], who have developed a kernel that approximates the optimal partial matching between two feature sets. By contrast, our kernel is based on EMD, which is the exact partial matching cost [132].

Table 3.5 compares the performance of the EMD kernel with nearest-neighbor classification (the setup of Section 3.1) on UIUC and Brodatz, as well as on one additional texture database, KTH-TIPS [51], and on one object database, Xerox7 [164].¹ All these results are obtained with scale- and rotation-invariant Laplacian regions and SIFT descriptors, and fixed signature size of 40 clusters per image. Note that these experimental settings are not the same as those of Section 3.1, and therefore the nearest-neighbor classification rates for UIUC and Brodatz shown in this table differ slightly from the ones reported in Sections 3.1.2.2 and 3.1.2.3 (the effects of different descriptors and levels of invariance will be examined in more detail in Section 3.2.2). We can see that the EMD kernel yields substantially higher results than nearest neighbor classification, i.e., that a discriminative approach gives a significant improvement. The difference is especially dramatic for the Xerox7 dataset, which contains substantial intra-class variation. Additional evidence of the power of the kernel-based classification approach comes from the comparative evaluation of several methods on the CUReT database [22] presented in Section 3.2.2: Table 3.11 shows that our original method obtains 72.5% accuracy on this database, whereas the improved kernel-based method of this section obtains 95.3% accuracy.

¹The KTH-TIPS and Xerox7 databases are included in Table 3.5 as two additional data points for the comparison, but are not further discussed in this dissertation for the sake of conciseness. The interested reader should refer to our report [168] for additional details.

Database	EMD+NN	EMD+Kernel
UIUC	95.0 ± 0.8	97.7 ± 0.6
Brodatz	86.5 ± 1.2	94.1 ± 0.8
KTH-TIPS [51]	88.2 ± 1.6	92.5 ± 1.5
Xerox7 [164]	59.4 ± 4.1	92.4 ± 1.7

Table 3.5 Classification accuracy of EMD with nearest neighbors vs. the EMD kernel for four different databases. The number of training images per class is 20 for UIUC and 3 for Brodatz. For details of the experimental settings of the KTH-TIPS and Xerox7 databases, see [168]. For the three texture databases, we randomly create 100 different training/test splits and report the average classification rate and its standard deviation over the 100 runs. For Xerox7, we use tenfold cross-validation as in [164].

3.2.2 Texture Recognition

This section evaluates the performance of the kernel-based method with different levels of detector invariance and different descriptors, and then reports results of a large-scale comparison of our method to other state-of-the-art texture recognition methods on three databases: UIUC, Brodatz, and CUReT [22].

3.2.2.1 Comparing Invariance Levels and Descriptor Types

First, we evaluate features with different levels of invariance: scale (S), scale with rotation (SR), and affine (A). Scale-invariant versions of the Harris and Laplacian detectors output circular regions at a certain characteristic scale. To achieve rotation invariance, we rotate the circular regions in the direction of the dominant gradient orientation [90, 105], which is computed as the average of all gradient orientations in the region. Finally, affine-invariant features are the same as the ones used in Section 3.1, except that here we eliminate the rotational ambiguity of the normalized regions by finding their dominant gradient orientation.

Tables 3.6 and 3.7 show the performance of the three types of features on the UIUC and Brodatz datasets, respectively. In this test, all regions are described with the SIFT descriptor and the EMD kernel is used for classification. As in Table 3.5, results are reported as the average and standard deviation of the classification rate obtained over 100 runs with different randomly selected training sets. From Table 3.6, we can see that for UIUC, rotation-invariant (SR) features work about as well as affine-invariant (A) ones. This is somewhat surprising, since from a theoretical standpoint, affine adaptation is expected to provide better invariance to many of the deformations present in this dataset, especially perspective distortions. The strong performance of scale- and rotation-invariant features on the UIUC database could be due to their greater robustness, as the affine adaptation process can often be unstable in the presence of large affine or perspective distortions. As Table 3.7 shows, for the Brodatz database, pure scale invariance (S) performs best. Because this database has no rotation or affine deformations, using features that are invariant to these transformations only destroys discriminative information (recall from Section 3.1.2.3 that we were able to achieve improved performance on the Brodatz database by adding an "ellipse channel" containing the shape information that was discarded by the affine-invariant detectors).

	Н	L	H+L
S	89.7 ± 1.5	91.2 ± 1.5	92.2 ± 1.4
\mathbf{SR}	97.1 ± 0.6	97.7 ± 0.6	98.0 ± 0.5
А	97.5 ± 0.6	97.5 ± 0.7	98.0 ± 0.6

Table 3.6 Evaluation of different levels of invariance on the UIUC database with 20 training images per class. Affine (A) and rotation-invariant (SR) features have similar levels of performance, while scale-invariant (S) features are much weaker.

Next, we evaluate the performance of different descriptors on UIUC and Brodatz databases. Tables 3.8 and 3.9 show the results of combining different descriptor channels using the EMD kernel. Overall, the rotation-invariant SPIN and RIFT descriptors perform slightly worse than SIFT, since they average intensity values and gradient orientations over ring-shaped regions and therefore loses important spatial information. Just as in Section 3.1.2, the

	Н	L	H+L
\mathbf{S}	89.2 ± 1.0	94.9 ± 0.7	94.4 ± 0.7
\mathbf{SR}	89.2 ± 1.0	94.1 ± 0.8	94.0 ± 0.9
А	84.7 ± 1.1	90.8 ± 0.9	91.3 ± 1.1

Table 3.7 Evaluation of different levels of invariance on the Brodatz database with 3 training images per class. Scale (S) and rotation-invariant (SR) features perform similarly, while affine-invariant (A) features are weaker.

SIFT+SPIN and RIFT+SPIN combinations exhibit boosted performance because the two descriptors capture different kinds of information (gradients vs. intensity values). As expected, however, the SIFT+SPIN+RIFT combination shows only an insignificant improvement over SIFT+SPIN, as SIFT and RIFT capture the same type of information. It is interesting to note that in the nearest-neighbor classification framework of Section 3.1.2, direct combination of different channels could result in a significant decrease of overall performance. For example, referring back to Table 3.4, we can see that combining the L/RIFT channel with the weaker H/RIFT channel significantly reduced the performance of the former. However, in the kernel-based classification framework, this problem is avoided: in the new results of Table 3.9, the performance of the (H+L)RIFT combination is higher than L/RIFT in isolation. Note, however, that a "negative interference" effect is still possible in the improved kernel-based framework: For example, in Table 3.9, the performance of the L/SIFT channel is slightly higher than the performance of L(SIFT+SPIN).

3.2.2.2 Comparative Evaluation: UIUC Database, Brodatz, CUReT

Next, we present a comparative evaluation of our two approaches with three state-of-theart texture classification methods in the literature. Our first method, denoted 'Lazebnik' in the following, is the one described in Section 3.1: It uses affine-invariant features with the (H+L)(SPIN+RIFT) channel combination and nearest neighbors with EMD for classifica-

	Н	L	H+L
SIFT	97.5 ± 0.6	97.5 ± 0.7	98.0 ± 0.6
SPIN	95.5 ± 0.8	96.0 ± 0.9	97.0 ± 0.7
RIFT	94.8 ± 0.8	96.4 ± 0.7	97.0 ± 0.7
SIFT+SPIN	97.9 ± 0.6	98.1 ± 0.6	98.5 ± 0.5
RIFT+SPIN	97.1 ± 0.7	97.8 ± 0.6	98.0 ± 0.6
SIFT+SPIN+RIFT	98.1 ± 0.6	98.5 ± 0.5	98.7 ± 0.4

Table 3.8 Detector and descriptor evaluation on UIUC with affine-invariant features using20 training images per class.

	Н	L	H+L
SIFT	89.2 ± 1.0	94.1 ± 0.9	94.0 ± 0.9
SPIN	86.1 ± 1.1	87.9 ± 1.0	90.2 ± 1.0
RIFT	82.7 ± 1.0	88.5 ± 0.9	89.6 ± 1.0
SIFT+SPIN	92.2 ± 0.9	93.2 ± 0.8	94.3 ± 0.8
RIFT+SPIN	89.8 ± 1.1	91.4 ± 0.9	92.8 ± 0.8
SIFT+SPIN+RIFT	92.8 ± 1.0	94.1 ± 0.9	94.9 ± 0.8

Table 3.9 Detector and descriptor evaluation on Brodatz using scale- and rotation-invariant (SR) features and 3 training images per class.

tion. Our second method, denoted 'Zhang,' uses S or SR features² with the (H+L)(SIFT+ SPIN) combination and the EMD kernel for classification. Two other methods are due to Varma and Zisserman (VZ) [157] and Hayman et al. [51]. We have chosen these two methods for comparison because they achieve state-of-the-art classification rates to date on the popular CUReT database. The VZ method uses a dense set of 2D textons; the descriptors are raw pixel values measured in fixed-size neighborhoods. In our experiments we use 7×7 neighborhoods resulting in 49-dimensional feature vectors. Each image is represented by the histogram of its texton labels, and classification is performed using nearest neighbors with χ^2 distance. Hayman's method is an extension of VZ that uses a generalized Gaussian kernel with χ^2 distance for classification. Finally, we have implemented a baseline traditional approach that works by classifying the mean and standard deviation of the responses of Gabor filters computed over the entire image [98]. We use the same Gabor filters as in [98] (six orientations and four scales), and perform nearest neighbor classification based on Mahalanobis distance.

Figure 3.11 shows the classification accuracy of the five methods on the UIUC database. The results are presented as plots of classification rate vs. the number of training images. We can observe that our two methods work much better than Hayman and VZ, demonstrating the advantage of intrinsically invariant features. Moreover, the improvement of Zhang over Lazebnik and of Hayman over VZ demonstrates one more time the advantage of discriminative learning over nearest-neighbor classification. Finally, the baseline Gabor method performs the worst, since averaging the Gabor filter outputs over all pixels loses discriminative information. Interestingly, Figure 3.11 reveals the reliance of the three non-invariant methods (Hayman, VZ, Gabor) on multiple prototypes: their performance improves steadily as the number of training images is increased, whereas the performance of our two methods is already relatively high with only one or two training samples, and more or less levels off after

 $^{^2{\}rm S}$ features are used for all comparisons except the UIUC database, where SR features make a big difference in performance.



Figure 3.11 Comparison of different methods on the UIUC database.

five. The outcome of the comparison validates our intuition that intrinsic representation-level invariance is necessary to achieve robustness to large viewpoint and scale changes, especially for very small training sets, where the lack of invariance cannot be fully compensated by storing multiple prototypes of each texture.

Next, Table 3.10 presents a comparison of the five methods on the Brodatz database for one and three training samples per class. The two kernel-based methods (Zhang and Hayman) perform the best, followed by VZ, followed by Lazebnik and Gabor. The results seem to indicate that our representation, based on sparse local invariant features, and the traditional representation, based on dense non-invariant features, are equally able to capture the structures of the Brodatz textures. Furthermore, the large difference between Zhang and Lazebnik, both of which have very similar features, once again shows the power of kernel-based learning.

Mothods	training images per class		
Methods	1	3	
Zhang	88.8 ± 1.0	95.4 ± 0.3	
Hayman	88.7 ± 1.0	95.0 ± 0.8	
VZ	87.1 ± 0.9	92.9 ± 0.8	
Lazebnik	80.0 ± 1.3	89.8 ± 1.0	
Gabor mean+std	80.4 ± 1.2	87.9 ± 1.0	

Table 3.10 Comparison on the Brodatz database.

Our next evaluation is on the Columbia-Utrecht Reflectance and Texture Database (CUReT) [22]. We use the same subset of this database as [156, 157]. This subset contains 61 texture classes with 92 images for each class. These images are captured under different illuminations with seven different viewing directions. The changes of viewpoint, and, to a greater extent, of the illumination direction, significantly affect the texture appearance, cf. Figure 3.12. However, the CUReT database also has some limitations: its images have no scale variation (all materials are held at the same distance from the camera, only the orientation is changed), limited in-plane rotation, and the same physical surface patch is represented in all samples. As Table 3.11 shows, Hayman and VZ obtain the best results, which is not surprising, since they were optimized specifically for this database. By contrast, our feature extraction framework is not ideally suited for CUReT, since most of its textures are very homogeneous and exhibit high-frequency patterns devoid of salient structures such as blobs and corners. Moreover, because the training set is fairly large, and therefore capable of representing all possible appearance variations, the intrinsic invariance of our representation is not a big advantage. Despite this unfavorable setup, the Zhang method still achieves results that are very close to VZ, while the Lazebnik method comes in below the baseline, demonstrating the failure of sparse features for this type of data in the absence of a discriminative classifier.



Figure 3.12 Image examples of CUReT textures under different illuminations and viewpoints.

Overall, our results show that for most datasets, combining geometric invariance at the representation stage with a discriminative classifier at the learning stage results in a very effective texture recognition system. Note that even though impressive results are obtained using raw pixel descriptors on the CUReT dataset, this method does not perform as well on the other datasets, thus showing its limited applicability. Our results further show that high-frequency, more homogeneous textures such as CUReT are best handled by methods using dense fixed-size patches, while more non-homogeneous low-frequency textures are best handled by a sparse local feature representation. Another important factor affecting the performance of local feature methods is image resolution, since keypoint extraction tends to not work well on low-resolution images. For example, CUReT images of size 200×200 have an average of 236 Harris-Laplace regions and 908 Laplacian regions, while UIUC images of size 640×480 have an average of 2152 and 8551 regions, respectively. It is not surprising that such an order-of-magnitude difference in the number of features extracted makes a difference in classification performance.

3.2.3 Object Recognition: Caltech6, Caltech101, Graz

When we first introduced our bag-of-features image representation in [72], we have intended it primarily for applications of texture recognition, because it operationalizes the

Methods	Ave. classification rate
Hayman	98.6 ± 0.2
VZ	96.0 ± 0.4
Zhang	95.3 ± 0.4
Gabor mean+std	92.4 ± 0.5
Lazebnik	72.5 ± 0.7

Table 3.11 Comparison on the CUReT texture database using 43 training images per class.

long-standing view of texture as a visual phenomenon formed by the repetition of a certain number of basic "tonal primitives" [2, 49]. Stated another way, a bag-of-features model conforms to the accepted practice of modeling textures as stationary Markov random fields. Informally, stationarity means that the probability distribution for the appearance of an individual feature does not depend on its location in the image, while the Markov property means that this distribution depends only on that feature's immediate neighborhood, not on the entire image. For most textures, we lose very little discriminative power by defining the neighbrhood as being empty, thus eliminating spatial dependencies altogether and arriving at an orderless local model. For object classes, on the other hand, the Markov assumptions are clearly violated, since features lying on the object have different statistical properties than features lying on the background, and, moreover, object parts are typically differentiated from each other (i.e., each part potentially has a different appearance distribution), and spatial dependencies between parts may be long-ranging. Nevertheless, over the last few years, several bag-of-features methods [20, 45, 164] have been proposed for object recognition and have achieved surprisingly good performance using local features similar to ours. This motivates us to evaluate our approach on several object category datasets and to compare our results to the best ones reported in the literature. The specific datasets used in the evaluation are Caltech6 [33], Caltech101 [28], Graz [118], and PASCAL [26]. In the following experiments, just as in Section 3.2.2, we use scale-invariant (H+L)(SIFT+SPIN) features, and the EMD kernel for classification.



Figure 3.13 Image examples of the six categories of Caltech6 dataset. The dataset may be downloaded from http://www.robots.ox.ac.uk/~vgg/data.

The Caltech6 database [33] contains airplanes (side) (1074 images), cars (rear) (1155 images), cars (side) ³ (720 images), faces (front) (450 images), motorbikes (side) (826 images), spotted cats (200 images), and a background set (900 images). The original category of spotted cats is from the Corel image library and contains 100 images. Here we flipped the the original images to have the same set as used in [33]. We use the same training and test set for two-class classification (object vs. background) as [33]. Some images are presented in Fig. 3.13. Table 3.12 compares the equal error rates on the Caltech6 dataset of our approach to two of the state-of-the-art methods, the bag-of-features Xerox approach [164] and the constellation model of Fergus et al. [33]. We can see that our method performs best for all of the object classes except cars (rear); however, the results obtained by the other methods are also quite high, indicating the relatively low level of difficulty of the Caltech dataset. Note that our method is fairly similar to Xerox (both use local invariant features and SVMs), and our higher performance in the same experimental setting is most likely due to better

³The car (side) images are originally from the UIUC car dataset [1], which may be downloaded from http://l2r.cs.uiuc.edu/~cogcomp/Data/Car.

implementation choices. In particular, we improve performance by using a combination of several detectors and descriptors and a more discriminative SVM kernel (EMD vs. linear, see [168] for details).

	Zhang	Xerox [164]	Fergus [33]
airplanes	98.8	97.1	90.2
cars (rear)	98.3	98.6	90.3
cars (side)	95.0	87.3	88.5
faces	100	99.3	96.4
motorbikes	98.5	98.0	92.5
spotted cats	97.0	N/A	90.0

Table 3.12 ROC equal error rates on the Caltech6 dataset.

The Caltech101 dataset [29] contains 101 object categories with 31 to 800 images per category. Some of the images are shown in Fig. 3.14. Caltech-101 is the most diverse object database available today, though it is not without shortcomings. Namely, most images feature relatively little clutter, and the objects are centered and occupy most of the image. In addition, a number of categories, such as pagoda, are affected by "corner" artifacts resulting from artificial image rotation. Though these artifacts are semantically irrelevant, they can provide cues resulting in misleadingly high recognition rates. We follow the experimental setup of Grauman and Darrell [45], i.e., we randomly select 30 training images per class and test on the remaining images reporting the average accuracy. We repeat the selection 10 times and report the average classification rate and its standard deviation.

Table 3.13 shows the results for multi-class classification on Caltech101 dataset. Our approach outperforms Grauman and Darrell for the same setup. The best results on this dataset (48%) published to date are due to Berg et al. [5]. However, their results are not directly comparable to ours since they use 15 training images per class. Also note that in Chapter 6 we will introduce a *spatial matching* method that achieves 64.6%. Fig. 3.14 shows



Figure 3.14 Image examples of the Caltech101 dataset. On the left the three classes with the best classification rates and on the right those with the lowest rates. The dataset may be downloaded from http://www.vision.caltech.edu/Image_Datasets/Caltech101.

a few categories with the best and worst classification rates for our method. We can observe that some of the lowest rates are obtained for categories that are characterized by their shape as opposed to texture, such as anchors.

	Zhang	Berg $[5]$	Grauman [45]
overall rate	53.9	48	43

 Table 3.13 Classification accuracy on the Caltech101 dataset.

The Graz dataset [118] contains persons (460 images), bikes (373 images) and a background class (270 images). Some of the images are shown in Figure 3.15. We use the same training and test set for two-class classification as [118], for a total of 350 images. We also tested our method for two-class classification on the Graz dataset [118] (Table 3.14). Our method performs significantly better than Opelt et al. [118]. Figure 3.15 shows some images correctly classified by our method and misclassified ones. Misclassified bikes are either observed from the front, very small, or only partially visible. Misclassified people are either observed from the back, occluded, or very small.



Figure 3.15 Image examples of the two categories and a background class of Graz dataset. The images on the left are correctly classified with our approach, the images on the right were misclassified. The dataset may be obtained from http://www.emt.tugraz.at/~pinz/data.

To summarize, our method achieves the best results to date on Caltech6, Caltech101, and Graz datasets. In addition, as reported in [168], it also does better than existing methods on the Xerox7 database [164] and on test set 2 of the PASCAL challenge [26]. We omit a detailed discussion of these results from the dissertation for conciseness. Our model outperforms not only other bag-of-features methods [45, 164], but also constellation models [33] and methods based on non-rigid geometric correspondence [5]. However, it is important to emphasize that we do not propose basic bag-of-features methods as a solution to the general object recognition problem. Instead, we demonstrate that, given the right implementation choices, simple orderless image representations with suitable kernels can be effective on a wide variety of imagery. Thus, they can serve as good baselines for measuring the difficulty of newly acquired datasets and for evaluating more sophisticated recognition approaches, e.g., ones based on parts-and-relations object models such as those discussed in Chapter 5.

3.3 Discussion

This chapter has described an orderless image representation based on salient local features. Section 3.1 has introduced the basic components of this approach for the application

	Zhang	Opelt [118]
bikes	92.0	86.5
people	88.0	80.8

Table 3.14 ROC equal error rates on the Graz database.

of texture recognition. Traditional texture representations are *dense*, i.e., they use features computed at every pixel. To the best of our knowledge, ours is the first texture recognition method that uses features computed at a small set of image locations. The experiments of Section 3.1.2 show that it is possible to successfully recognize many textures based on information contained in a very small number of image regions. Next, Section 3.2 has augmented our basic approach with an improved classification step using an SVM kernel based on Earth Mover's Distance. By augmenting our method with a kernel-based classifier, we have made it more effective not only for texture, but also for object recognition. The results of the comparative evaluations of Section 3.2 convincingly demonstrate the promise of our method.

The rest of this dissertation will focus on extending the purely local bag-of-features method with a variety of spatial constraints. Chapter 4 will explore the weakest form of constraints, namely, loose statistical co-occurrence relations between neighboring local features. The resulting representation is a two-level scheme with intensity-based textons at the first level and histograms of texton distributions over local neighborhoods at the second level (see, e.g., [96, 137]). Next, Chapter 5 will present stronger *semi-local* constraints, i.e., rigid geometric constraints between multiple neighboring regions that will enable us to discard clutter and to identify groups of object features that form stable spatial configurations. Finally, Chapter 6 will present a *global* method that takes into account the absolute image positions of all features to effectively capture regularities in the spatial structure of pictures of natural scenes.

CHAPTER 4

Neighborhood Co-occurrence Relations for Texture Recognition

This chapter presents two strategies for enriching bag-of-features texture models with pairwise co-occurrence relations between nearby local features. Section 4.1 presents a texture model that uses a discriminative *maximum entropy* classifier to combine individual texton frequencies with co-occurrence frequencies in a principled manner. This model is suitable for the same whole-image classification tasks that were considered in the previous chapter. Section 4.2 addresses a more challenging problem, that of classifying or labeling individual local features in multi-texture images. For this problem, we adopt a two-stage framework that first learns an orderless local model for the appearance of individual features, and then uses relaxation to incorporate context information provided by co-occurrence relations. The research presented in the first and second parts of this chapter has been published in [74] and [71], respectively.

4.1 A Maximum Entropy Framework for Combining Local Features and Their Relations

This section presents a texture classification method that learns a textons-and-relations model analogous to a *bigram model* used in language modeling and classifies images with the help of a discriminative *maximum entropy* framework, which has been used successfully for text document classification [6, 114] and image annotation [56]. This framework has several
characteristics that make it attractive for visual categorization as well: It directly models the posterior distribution of the class label given the image, leading to convex (and tractable) parameter estimation; moreover, classification is performed in a true multi-class fashion, requiring no distinguished background class. Because the maximum entropy framework makes no independence assumptions, it offers a principled way of combining multiple kinds of features (e.g., regions produced by different detectors), as well as co-occurrence relations between pairs of features, into the image representation. While maximum entropy has been widely used in the computer vision for *generative* tasks, e.g., modeling of images as Markov random fields [169], where it runs into issues of intractability for learning and inference, it can be far more efficient for *discriminative* tasks. For example, Mahamud et al. [94] have used maximum entropy to combine multiple nearest-neighbor discriminators, and Keysers et al. [64] have applied it to digit recognition. In this dissertation, we explore the usefulness of this framework for combining local features and their relations.

4.1.1 The Maximum Entropy Classifier

In a discriminative maximum entropy framework [6], we seek to estimate the posterior distribution of class labels given image features that matches the statistics of the features observed in the training set, and yet remains as uniform as possible. Intuitively, such a distribution properly reflects our uncertainty about making a decision given ambiguous or inconclusive image data. (By contrast, some generative methods, e.g., mixtures of Gaussians, tend to yield peaky or "overconfident" posterior distributions.) Suppose that we have defined a set of *feature functions* $f_k(I,c)$ that depend both on the image I and the class label c (the definitions of the specific feature functions used in this work will appear in Section 4.1.2). To estimate the posterior of the class label given the features, we constrain the expected values of the features under the estimated distribution P(c|I) to match those observed in the training set \mathcal{T} . The observed "average" value of feature f_k in the training set \mathcal{T} is

$$\hat{f}_k = \frac{1}{|\mathcal{T}|} \sum_{I \in \mathcal{T}} f_k(I, c(I)) \, .$$

Given a particular posterior distribution P(c|I), the expected value of f_k , taken with respect to the observed empirical distribution P(I) over the training set, is

$$E[f_k] = \frac{1}{|\mathcal{T}|} \sum_{I \in \mathcal{T}} \sum_{c} P(c|I) f_k(I, c) \,.$$

We seek the posterior distribution that has the maximum *conditional entropy*

$$H = -\frac{1}{|\mathcal{T}|} \sum_{I \in \mathcal{T}} \sum_{c} P(c|I) \log P(c|I)$$

subject to the constraints $E[f_k] = \hat{f}_k$. It can be shown that the desired distribution has the exponential form

$$P(c|I) = \frac{1}{Z} \exp\left(\sum_{k} \lambda_k f_k(I,c)\right), \qquad (4.1)$$

where

$$Z = \sum_{c} \exp\left(\sum_{k} \lambda_k f_k(I, c)\right)$$

is the normalizing factor,¹ and the λ_k are parameters whose optimal values are found by maximizing the likelihood of the training data under the exponential model (4.1). This optimization problem is convex and the global maximum can be found using the improved iterative scaling (IIS) algorithm [6, 114]. At each iteration of IIS, we compute an update δ_k to each λ_k , such that the likelihood of the training data is increased. To do this, we bound $L(\lambda+\delta)-L(\lambda)$ from below by a positive function $F(\delta)$, and find the value of δ that maximizes this function. The derivation of updates is omitted here, but it can be shown [6, 114] that

¹Note that Z involves only a sum over the classes, and thus can be computed efficiently. If we were modeling the distribution of features given a class instead, Z would be a sum over the exponentially many possible combinations of feature values — a major source of difficulty for a generative approach. By contrast, the discriminative approach described here is more related to logistic regression. It is easy to show that (4.1) yields binary logistic discrimination in the two-class case.

when the features are *normalized*, i.e., when $\sum_{k} f_k(I, c)$ is a constant S for all I and c, updates can be found efficiently in closed form:

$$\delta_k = \frac{1}{S} \left(\log \hat{f}_k - \log E_\lambda[f_k] \right) \,. \tag{4.2}$$

Because of the computational efficiency gained in this case, we use only normalized features in the present work.

Because of the form of (4.2), zero values of \hat{f}_k cause the optimization to fail, and low values cause excessive growth of the weights. This is a symptom of one of the biggest potential pitfalls of the maximum entropy framework: overfitting. When the training set is small, the observed averages may deviate significantly from the "true" expectations, leading to a poor estimate of the posterior distribution. This problem can be alleviated by adding a zero-mean Gaussian prior on the weights [114]. However, in our experiments, we have achieved better results with a basic IIS setup where simple transformations of the feature functions are used to force expectations away from zero. Specifically, for all our feature functions, we use the standard Laplace smoothing, i.e., adding one to each feature value and renormalizing. To simplify the subsequent presentation, we will omit this operation from all feature function definitions.

We close this section with a note concerning the technique we use to design feature functions. Instead of directly defining class-dependent features $f_k(I, c)$, it is much more convenient to obtain them from a common pool of *class-independent* features $g_k(I)$ using the following simple transformation [114]:

$$f_{d,k}(I,c) = \begin{cases} g_k(I) & \text{if } c = d, \\ 0 & \text{otherwise.} \end{cases}$$

Then we have

$$P(c|I) = \frac{1}{Z} \exp\left(\sum_{d,k} \lambda_{d,k} f_{d,k}(I,c)\right) = \frac{1}{Z} \exp\left(\sum_{k} \lambda_{c,k} g_k(I)\right) \,.$$

Thus, "universal" features g_k become associated with class-specific weights $\lambda_{c,k}$. All our feature functions will be defined in this way. Note, however, that the exponential framework also allows completely different features for representing each class.

4.1.2 Texton Vocabulary and Feature Functions

In this section, we describe the application of the maximum entropy framework to texture recognition. First, we describe our texton-based representation, and in Section 4.1.3 we discuss experiments on the Brodatz database (Section 3.1.2.3, Figure 3.7) and the UIUC texture database (Section 3.1.2.2, Figure 3.5). For the Brodatz database, we use scaleinvariant Laplacian features. Recall from Section 3.2.2 that scale-invariance is sufficient for the Brodatz database, which does not feature any significant geometric deformations between different samples from the same class. By contrast, the UIUC database contains arbitrary rotations, perspective distortions and non-rigid deformations. As we have seen in Section 3.2.2, this greater degree of geometric variability requires a greater degree of invariance in the low-level features. Therefore, we process the UIUC database with the affinely adapted Laplacian detector, which returns elliptical regions. In both cases, the appearance of the detected regions is represented using SIFT descriptors.

Instead of clustering descriptors in all training images individually to form signatures, as in Chapter 3, we now represent images as histograms of texton labels from a universal vocabulary. A universal vocabulary is necessary to us in this section because the maximum entropy classifier requires that each image be characterized by a fixed-length, ordered vector of feature functions. By contrast, the SVM classification framework used in Section 3.2 required only the ability to evaluate a kernel function for any pair of images, which we could do based on the variable-length signature representation.

To form the universal vocabulary, we run K-means clustering on a randomly selected subset of all training descriptors. To limit the memory requirements of the K-means algorithm, we cluster each class separately and concatenate the resulting textons. We find K = 10 and K = 40 textons per class for the Brodatz and the UIUC database, respectively, resulting in dictionaries of size 1110 and 1000. Finally, each descriptor from a new image is assigned the label of the closest cluster center.

The next step is to define the feature functions for the exponential model. For text classification, Nigam et al. [114] use scaled counts of word occurrences in a document. By analogy, we define feature functions based on texton frequencies:

$$g_k(I) = \frac{N_k(I)}{\sum_{k'} N_{k'}(I)},$$

where $N_k(I)$ is the number of times texton label k occurs in the image I. To enrich the feature set, we also define functions $g_{k,\ell}$ that encode the probability of co-occurrence of pairs of labels at nearby locations. Let $k \diamond \ell$ denote the event that a region labeled ℓ is adjacent to a region labeled k. Specifically, we say that $k \diamond \ell$ if the center of ℓ is contained in the neighborhood obtained by "growing" the shape (circle or ellipse) of the kth region by a constant factor (4 in the implementation). Let $N_{k\diamond\ell}(I)$ denote the number of times the relation occurs in the image I, and define

$$g_{k,\ell}(I) = \frac{N_{k \diamond \ell}(I)}{\sum_{k',\ell'} N_{k' \diamond \ell'}(I)}$$

An image model incorporating co-occurrence counts of pairs of adjacent labels is a counterpart of a *bigram language model* that estimates the probabilities of two-word strings in natural text. Just as in language modeling, we must deal with sparse probability estimates due to many relations receiving extremely low counts in the training set. Thus, we are led to consider smoothing techniques for probability estimates [17]. One of the most basic techniques, interpolation with marginal probabilities, leads to the following modified definition of the co-occurrence features:

$$\tilde{g}_{k,\ell}(I) = (1-\alpha)g_{k,\ell}(I) + \alpha \Big(\sum_{\ell'} g_{k,\ell'}(I)\Big)\Big(\sum_{k'} g_{k',\ell}(I)\Big)\,,$$

where α is a constant (0.1 in our implementation). Informally, a co-occurrence relation $k \diamond \ell$ should have higher probability if both k and ℓ occur frequently in samples of the class, and if they each have many neighbors.

While smoothing addresses the problem of unreliable probability estimates, we are still left with millions of possible co-occurrence relations, and it is necessary to use feature selection to reduce the model to a manageable size. Possible feature selection techniques include greedy selection based on increase of likelihood under the exponential model [6], mutual information [23, 114] and likelihood ratio [23]. However, since more frequently occurring relations yield more reliable estimates, we have chosen a simpler likelihood-based scheme: For each class, we find a fixed number of relations that have the highest probability in the training set, and then combine them into a global "relation dictionary." In the implementation, we select 10 K features per class (recall that K is the number of textons per class); because the relations selected for different classes sometimes coincide, the total number of $g_{k,\ell}$ features is slightly less than ten times the total size of the texton dictionary.

4.1.3 Experimental Results: UIUC Database and Brodatz

Table 4.1 shows a comparison of classification rates obtained using various methods on the two databases. All the rates are averaged over 10 runs with different randomly selected training subsets; standard deviations of the rates are also reported. The training set consists of 3 (resp. 10) images per class for the Brodatz (resp. UIUC) database. The first row shows results for a popular baseline method using nearest-neighbor classification of texton histograms with the χ^2 distance (for an example of such an approach, see, e.g., [157]). The second row shows results for a Naive Bayes baseline using the *multinomial event model* [102]:

$$P(I|c) = \prod_{k} P(k|c)^{N_k(I)} ,$$

where P(k|c) is given by the frequency of texton k in the training images for class c. The results for the two baseline methods on the Brodatz database are comparable, though Naive

	Brodatz	UIUC	
χ^2	83.09 ± 1.18	94.25 ± 0.59	
Naive Bayes	85.84 ± 0.90	94.08 ± 0.67	
Exp. g_k	87.37 ± 1.04	97.41 ± 0.64	
Exp. $g_{k,\ell}$	75.20 ± 1.34	92.40 ± 0.93	
Exp. $g_k + g_{k,\ell}$	83.44 ± 1.17	97.19 ± 0.57	
Exp. $\tilde{g}_{k,\ell}$	80.51 ± 1.09	95.85 ± 0.62	
Exp. $g_k + \tilde{g}_{k,\ell}$	83.36 ± 1.14	97.09 ± 0.47	

Table 4.1 Texture classification results using the maximum entropy classifier. See text for a comparison with the results presented in the previous chapter.

Bayes has a potential advantage over the χ^2 method, since it does not treat the training samples as independent prototypes, but combines them in order to compute the probabilities P(k|c). This may help to account for the slightly better performance of Naive Bayes on the Brodatz database.

The third and fourth rows show results for exponential models based on individual g_k (textons only) features and $g_{k,\ell}$ (relations only) features, respectively, and the fifth row shows results for the exponential model with both kinds of features combined. For both databases, the texton-only exponential model performs much better than the two baseline methods; the relations-only models are inferior to the baseline. Interestingly, combining textons and relations does not improve performance. To test whether this is due to overfitting, we compare performance of the $g_{k,\ell}$ features with the smoothed $\tilde{g}_{k,\ell}$ features (last two rows). While the smoothed features do perform better, combining them with textons-only features once again does not bring any improvement. Thus, texton-only features clearly supercede the co-occurrence relations.



Figure 4.1 Histograms of classification rates for the Brodatz database (left) and the UIUC database (right).

To get a more detailed look at the performance of the exponential model, refer to Figure 4.1, which shows the histograms of classification rates achieved by the parts-only exponential model for individual classes. With the parts-only exponential model, 100% recognition rate is achieved by 61 classes from the Brodatz database and by 8 classes from the UIUC database. This distribution of classification rates, in particular for the Brodatz database, suggests another reason (besides overfitting) for the lack of improvement afforded by co-occurrence features. Namely, most classes in the database can be represented quite well without taking texton co-occurrences into account, while a few are either extremely nonhomogeneous or extremely perceptually similar to another class. Consequently, adding relations to the exponential model cannot improve the recognition of either the "easy" or the "difficult" classes.

Overall, the g_k exponential model performs the best for both texture databases. For Brodatz, our present rate of 87.37% is comparable to the rate of 87.44% achieved with nearest-neighbor classification in Section 3.1.2.3 (Table 3.4). However, it is well below the 94.1% accuracy achieved in Section 3.2.1 with the kernel-based classifier (Table 3.5). One possible reason for the apparent disadvantage of the maximum entropy classifier over the SVM in this case is the extremely small training set size for the Brodatz database (only three images), which may be causing the exponential model to overfit. For the UIUC database, our present result of 97.41% exceeds the nearest-neighbor result of 92.61% (Section 3.1.2.2, Table 3.2) and is very similar to the 98% achieved with SIFT features and the kernel-based approach (Section 3.2.2, Table 3.8). Note that though the latter result was achieved with 20 training images per class instead of ten as in this section, Figure 3.11 shows that classification accuracy of the kernel-based method on the UIUC database changes very little between these two training set sizes.

4.2 A Two-Stage Approach for Recognizing Local Texture Regions

In the rest of this chapter, we address the problem of recognizing individual regions in multi-texture images. Potential practical applications of this problem include texture-based retrieval in image databases [131, 137], classification of natural scenes [69], and segmentation of images into regions having different semantic categories, e.g., natural vs. man-made [11, 68].

So far in this dissertation, we have demonstrated several times that it is possible to robustly classify whole single-texture images by comparing distributions of local features gathered over a large field. Determining the class of an individual region is a more challenging task because the local appearance of the region itself is usually an ambiguous cue (this is the *aperture problem* that shows up in many areas of computer vision). A common approach to reducing the ambiguity of local appearance is to augment the local image representation with a description of the spatial relationship between neighboring regions. The systems developed by Malik et al. [96] and Schmid [137] are examples of this two-layer architecture, with intensity-based descriptors (textons) at the first level and histograms of texton distributions at the second. This section proposes a conceptually similar two-stage approach to texture modeling.

The first stage of our approach consists in estimating the distribution of local intensity descriptors. Just as in Chapter 3, we automatically select the shape of support regions over which the descriptors are computed by using the affine adaptation process. Though affine adaptation usually does not produce neighborhoods large enough to serve as representative texture samples, it does succeed in finding the characteristic scale of primitive elements such as blobs. The descriptors computed over this scale are already somewhat distinctive and can be classified with a moderate degree of success, as demonstrated in Section 4.2.2.1. We represent the distribution of descriptors in each class by a Gaussian mixture model where each component corresponds to a "sub-class". This generative model is used to assign the most likely sub-class label to each region extracted from a training image. At the second stage of the modeling process, co-occurrence statistics of different sub-class labels are computed over neighborhoods adaptively defined using the affine shape of local regions. Test images may contain multiple textures, and they are also processed in two stages. First, the generative model is used to assign initial estimates of the probability for each sub-class to each feature vector. These estimates are then refined using a relaxation step that incorporates co-occurrence statistics.

The most basic form of the modeling process is *fully supervised*, making use of training data containing only single-texture images. However, we show in Section 4.2.1.1 that a weaker form of supervision is possible: the training data may include unsegmented multi-texture images labeled by the set of texture classes that they contain.

The following outline of the organization of our system may serve as a guide to the rest of this section:

- Feature extraction: Detect a sparse set of affinely adapted regions in an image and compute an intensity-based descriptor for each region. This stage is the same as in Chapter 3.
- 2. Density estimation (Section 4.2.1.1): Use the EM algorithm to estimate class conditional distributions of descriptors as mixtures of Gaussian "sub-classes".
- 3. Neighborhood statistics (Section 4.2.1.2): Define the neighborhood of each local region as a function of its affine shape, and compute co-occurrence statistics of sub-class labels over this neighborhood.
- 4. Testing (Section 4.2.1.3): Obtain initial probability estimates using the generative model and perform relaxation to refine these estimates.

In Section 4.2.2, we evaluate the proposed texture representation on two data sets. The first set consists of photographs of textured surfaces taken from different viewpoints and featuring significant scale changes and perspective distortions. The second set consists of images of three types of animals whose appearance can be adequately modeled by texture-based methods.

The research described in this chapter has been published in [71].

4.2.1 Modeling Textures

At the feature extraction stage, our implementation uses the Laplacian blob detector introduced in the previous chapter. The descriptors used are 10×10 spin images, treated as 100-dimensional vectors in the following.

4.2.1.1 Density Estimation

In the supervised framework, the training data consists of single-texture sample images from texture classes with labels C_{ℓ} , $\ell = 1, ..., L$. Thus, the class-conditional densities $p(y|C_{\ell})$

can be estimated using all the feature vectors extracted from the images belonging to class C_{ℓ} . We model the class-conditional density of the feature vectors y given the class labels ℓ as $p(y|C_{\ell}) = \sum_{m=1}^{M} p(y|c_{\ell m}) p(c_{\ell m})$, where the components $c_{\ell m}$, $m = 1, \ldots, M$, are thought of as sub-classes. Each $p(y|c_{\ell m})$ is assumed to be a Gaussian with mean $\mu_{\ell m}$ and covariance matrix $\Sigma_{\ell m}$. We use the EM algorithm to estimate the parameters of the mixture model, namely the means $\mu_{\ell m}$, covariances $\Sigma_{\ell m}$, and mixing weights $p(c_{\ell m})$. This process can be thought of as "soft" clustering, where each component corresponds to a cluster in the feature space. Accordingly, EM is initialized with the output of the K-means algorithm. In this work, we use the same number of mixture components for each class (M = 15 and M = 10,respectively, for the experiments reported in Sections 4.2.2.1 and 4.2.2.2). To help avoid numerical problems in dealing with 100-dimensional features, we have scaled all feature vectors by a value empirically determined to give the best convergence. In addition, we have limited the number of free parameters in the optimization by using *spherical* Gaussians with covariance matrices of the form $\Sigma_{\ell m} = \sigma_{\ell m}^2 I$. This restriction also helps to prevent the covariance matrices from becoming singular. Another, more severe, restriction would be to constrain all components to have equal covariance matrices. We have experimented with both methods and determined that estimating a separate covariance value for each component creates more discriminative models. In the future, we plan to experiment with dimensionality reduction techniques to determine whether they will improve performance.

The EM framework for learning texture models provides a natural way of incorporating unsegmented multi-texture images into the training set. Our approach is inspired by the work of Nigam et al. [115], who have proposed several techniques for using unlabeled data to improve the accuracy of text classification. Suppose we are given a multi-texture image annotated with the set \mathcal{L} of class indices that it contains—that is, each feature vector y extracted from this image has an *incomplete* label of the form $C_{\mathcal{L}} = \{C_{\ell} | \ell \in \mathcal{L}\}$. To accommodate incomplete labels, the density estimation framework needs to be modified: instead of partitioning the training data into subsets belonging to each class and separately estimating L mixture models with M components each, we now use all the data simultaneously to estimate a single mixture model with $L \times M$ components. The estimation process must start by selecting some initial values for the parameters of the model (means, covariances, mixing weights). During the *expectation* (E) step, we use the parameters to compute probabilistic sub-class membership weights given the feature vectors y and the incomplete labels $C_{\mathcal{L}}$: $p(c_{\ell m}|y, C_{\mathcal{L}}) \propto p(y|c_{\ell m}) p(c_{\ell m}|C_{\mathcal{L}})$, where $p(c_{\ell m}|C_{\mathcal{L}}) = 0$ for all $\ell \notin \mathcal{L}$ and $\sum_{\ell \in \mathcal{L}} \sum_{m=1}^{M} p(c_{\ell m}|C_{\mathcal{L}}) = 1$. During the *maximization* (M) step, we use the computed weights to re-estimate the parameters by maximizing the expected likelihood of the data in the standard fashion [8].

We have conducted experiments that exercise the EM framework in two different ways. The data set described in Section 4.2.2.1 contains both single- and multi-texture training images, which are used respectively to initialize and refine the parameters of the full generative model. By contrast, the data set of Section 4.2.2.2 does not contain any segmented images at all.

Overall, the incorporation of incompletely labeled data requires only a slight modification of the EM implementation used for estimating the class-conditional densities $p(y|C_{\ell})$. However, this modification is of great utility, since the task of segmenting training examples by hand becomes an odious chore even for moderately-sized data sets. In situations where it is difficult to obtain large amounts of fully labeled examples, adding incompletely labeled or unlabeled data to the training set also helps to improve classification performance [115].

4.2.1.2 Neighborhood Statistics

We describe in this section the second layer of our texture representation, which accumulates information about the distribution of pairs of sub-class labels belonging to neighboring regions. Following the density estimation step, each region in the training image is assigned the sub-class label that maximizes the posterior probability $p(c_{\ell m}|y, C_{\mathcal{L}})$. Next, we need to a method for computing the neighborhood of a given elliptical region. Just as in Section 4.1, we can define the neighborhood as the set of all points obtained by "growing" the ellipse by a constant factor. However, during the relaxation step this definition sometimes produces undesirable results since points with small ellipses get too few neighbors, and points with large ellipses get too many. Therefore, we modify our neighborhood definition as follows: instead of multiplying the ellipse axes by a constant, we extend them by a small absolute amount (15 pixels in the implementation), and let the neighborhood consist of all points that fall inside this enlarged ellipse. In this way, the size and shape of the neighborhood still depends on the affine shape of the region, but the neighborhood structure is more balanced.

Once we have defined a neighborhood structure for the affine regions contained in an image, we can effectively turn this image into a directed graph with arcs emanating from the center of each region to other centers that fall within its neighborhood. The existence of an arc from a region with sub-class label c to another region with label c' is a joint event (c, c') (note that the order is important since the neighborhood relation is not symmetric). For each possible pair of labels, we estimate p(c, c') from the relative frequency of its occurrence, and also find the marginal probabilities $\hat{p}(c) = \sum_{c'} p(c, c')$ and $\check{p}(c') = \sum_{c} p(c, c')$. Finally, we compute the values

$$r(c,c') = \frac{p(c,c') - \hat{p}(c)\,\check{p}(c')}{\left[\left(\hat{p}(c) - \hat{p}^2(c)\right)\left(\check{p}(c') - \check{p}^2(c')\right)\right]^{\frac{1}{2}}}$$

representing the correlations between the events that the labels c and c', respectively, belong to the source and destination nodes of the same arc. The values of r(c, c') must lie between -1 and 1; negative values indicate that c and c' rarely co-occur as labels at endpoints of the same edge, while positive values indicate that they co-occur often.

In our experiments, we have found that the values of r(c, c') are reliable only in cases when c and c' are sub-class labels of the same class C. Part of the difficulty in estimating correlations across texture classes is the lack of data in the training set. Even if the set contains multi-texture images, only a small number of edges actually fall across texture boundaries. Unless the number of texture classes is very small, it is also quite difficult to create a training set that would include samples of every possible boundary. Moreover, since we do not make use of "ground-truth" segmented images, the boundaries found in multitexture images are not reliable. For these reasons, whenever c and c' belong to different classes, we set r(c, c') to a constant negative value that serves as a "smoothness constraint" in the relaxation algorithm described in the next section.

4.2.1.3 Relaxation

We have implemented the probability-based iterative relaxation algorithm described in the classic paper by Rosenfeld et al. [128]. The initial estimate of the probability that the *i*th region has label c, denoted $p_i^{(0)}(c)$, is obtained from the learned Gaussian mixture model as the posterior probability $p(c|y_i)$. Note that since we run relaxation on unlabeled test data, these probabilities must be computed for all $L \times M$ sub-class labels corresponding to all possible classes. At each iteration, new probability estimates $p_i^{(t+1)}(c)$ are obtained by updating the current values $p_i^{(t)}(c)$ using the equation

$$p_i^{(t+1)}(c) = \frac{p_i^{(t)}(c) \left[1 + q_i^{(t)}(c)\right]}{\sum_c p_i^{(t)}(c) \left[1 + q_i^{(t)}(c)\right]},$$

$$q_i^{(t)}(c) = \sum_j w_{ij} \left[\sum_{c'} r(c, c') p_j^{(t)}(c')\right].$$
(4.3)

The scalars w_{ij} are weights that indicate how much influence region j exerts on region i. We treat w_{ij} as a binary indicator variable that is nonzero if and only if the jth region belongs to the ith neighborhood. Note that the weights are required to be normalized so that $\sum_j w_{ij} = 1$ [128].

The update equation (4.3) can be justified in qualitative terms as follows. First of all, it can easily be seen that $p_j^{(t)}(c')$ has no practical effect on $p_i^{(t)}(c)$ when the *i*th and *j*th regions are not neighbors, when *c* and *c'* are uncorrelated, or when the probability $p_j^{(t)}(c')$ is low. However, the effect is significant when the *j*th region belongs to the *i*th neighborhood and the value of $p_j^{(t)}(c')$ is high. Intuitively, the correlation r(c, c') expresses how "compatible" the labels *c* and *c'* are at nearby locations. Thus, $p_i^{(t)}(c)$ is increased (resp. decreased) by the largest amount when r(c, c') has a large positive (resp. negative) value. Overall, the probabilities of different sub-class labels at neighboring locations reinforce each other in an intuitively satisfying fashion. Unfortunately, the iteration of (4.3) has no convergence guarantees, though the constraints built into the update equation do ensure that the $p_i^{(t)}(c)$ stay nonnegative and sum to 1 [128]. Despite the lack of a formal convergence proof, we have found the relaxation algorithm to behave well on our data. To obtain the results presented in Sections 4.2.2.1 and 4.2.2.2, we run relaxation for 200 iterations.

4.2.1.4 Classification and Retrieval

Individual regions are classified in the obvious way, by assigning them to the class that maximizes $p_i(C_\ell) = \sum_{m=1}^M p_i(c_{\ell m})$. To perform classification and retrieval at the image level, we need to define a "global" score for each texture class. In the experiments reported in the next section, the score for class C_ℓ is computed by summing the probability of C_ℓ over all N regions found in the image: $\sum_{i=1}^N \sum_{m=1}^M p_i(c_{\ell m})$, where the $p_i(c_{\ell m})$ are the probability estimates following relaxation. Classification of single-texture images is carried out by assigning the image to the class with the highest score, and retrieval for a given texture model proceeds from highest scores to lowest.

4.2.2 Experimental Results

4.2.2.1 The Indoor Scene

Our first data set contains seven different textures present in the atrium of the Beckman Institute (Figure 4.2). Figure 4.3 shows two sample images of each texture. To test the range of invariance in our representation, we have gathered images over a wide range of viewpoints and scales. The data set is partitioned as follows: 10 single-texture training images of each class; 10 single-texture validation images of each class; 13 two-texture training images; and 45 multi-texture test images.



Figure 4.2 The indoor scene that is the source of the textures in Figure 4.3.

Table 4.2 shows classification results for the single-texture validation images following training on single-texture images only. The columns labeled "image" show the fraction of images classified correctly using the score described in Section 4.2.1.4. As can be seen from the numbers in the first column, successful classification at the image level does not require relaxation: good results are achieved in most cases by using the probabilities output by the generative model of Section 4.2.1.1. Interestingly, for class T6 (marble), the classification rate actually drops as an artifact of relaxation. When the right class has relatively low initial probabilities in the generative model, the self-reinforcing nature of relaxation of ten serves



Figure 4.3 Samples of the texture classes used in the experiments of Section 4.2.2.1.

to diminish these probabilities even further. The columns labeled "region", which show the fraction of all *individual regions* in the validation images that were correctly classified based on the probabilities $p_i(C_\ell)$, are much more indicative of the impact of relaxation: for all seven classes, classification rates improve dramatically.

For an even more detailed look at classification performance of individual regions, refer to the confusion matrices on Figure 4.4. The entry in the *i*th row and *j*th column of each matrix represents the proportion of all regions from class T_i that were classified as T_j . Note that the values on the diagonal of the respective matrices correspond to the values in columns 2 and 4 of Table 4.2.

Next, we evaluate the performance of the system for retrieval of images containing a given texture. Figure 4.5 shows the results in the form of ROC curves that plot the positive detection rate (the number of correct images retrieved over the total number of correct images) against the false detection rate (the number of false positives over the total number of negatives in the data set). The top row shows results obtained after fully supervised training using single-texture images only, as described in Section 4.2.1.1. The bottom row shows the results obtained after re-estimating the generative model following the incorporation of 13 two-texture images into the training set. Following relaxation, a modest improvement in performance is achieved for most of the classes. A more significant increase in performance could probably be achieved by using a larger number of incompletely labeled examples [115].

	Before relaxation		After relaxation		
Class	image	region	image	region	
T1	100	61	100	97	
Τ2	100	58	100	99	
Т3	90	70	90	85	
T4	100	61	100	99	
T5	100	45	100	95	
Т6	90	29	80	67	
T7	60	41	70	73	

 Table 4.2 Classification rates for single-texture images.

The final stage of evaluation is purely qualitative: following relaxation, does the system succeed in providing a perceptually accurate segmentation of the image into regions of different texture? Part (a) of Figure 4.6 shows a typical example of the difference made by relaxation in the assignment of class labels to individual regions. Part (b) shows more examples where the relaxation was successful. Note in particular the top example of part (b), where the perceptually similar classes T4 and T5 are unambiguously separated. Part (c) of Figure 4.6 shows two examples of segmentation failure. In the bottom example, classes T2 (carpet) and T3 (chair) are confused, which can be partly explained by the fact that the scales at which the two textures appear in this image are not well represented in the training set. Overall, we have found the relaxation process to be sensitive to initialization of probabilities using the generative model, in the sense that poor initial probability estimates lead to severe artifacts in the final assignment. In the future, we plan to address this problem by implementing the relaxation step using a more modern and principled *belief propagation* algorithm [38, 67].



Figure 4.4 Confusion matrices: before relaxation (left) and after (right). Lighter colors correspond to higher values.



Figure 4.5 ROC curves (positive detection rate vs. false detection rate) for retrieval in the test set of 45 multi-texture images. The dashed (resp. solid) line represents performance before (resp. after) relaxation. Top row: single-texture training images only, bottom row: single-texture and two-texture training images.





(a) Initial labeling of regions (top) vs. the final labeling following relaxation (bottom).

Orig. image	T1 (Brick)	T2 (Carpet)	T3 (Chair)	T4 (Floor 1)	T5 (Floor 2)	T6 (Marble)	T7 (Wood)
	85p.			- 64,	A-D-		
				ig the		-	
		(b) Suc	cesstul segn	nentation ex	amples.		
Orig. image	T1 (Brick)	T2 (Carpet)	T3 (Chair)	T4 (Floor 1)	T5 (Floor 2)	T6 (Marble)	T7 (Wood)
				À			

(c) Unsuccessful segmentation examples.

Figure 4.6 Segmentation results.

4.2.2.2 Animals

Our second data set consists of unsegmented images of three kinds of animals: cheetahs, giraffes, and zebras. The training set contains 10 images from each class, and the test set contains 20 images from each class, plus 20 "negative" images not containing instances of the target animal species. To account for the lack of segmentation, we introduce an additional "background" class, and each training image is labeled as containing the appropriate animal and the background. Recall from Section 4.2.1.1 that the EM algorithm is initialized with the output of K-means. To initialize K-means on this incompletely labeled data, we randomly assign each feature vector either to the appropriate animal class, or to the background. The ROC curves for each class are shown in Figure 4.7, and segmentation results are shown in Figure 4.8. Overall, our system appears to have learned very good models for cheetahs and zebras, but not for giraffes.



Figure 4.7 ROC curves for the animal dataset. The dashed (resp. solid) line represents performance before (resp. after) relaxation.



Figure 4.8 Segmentation on the animal dataset.

We conjecture that several factors account for the weakness of the giraffe model. Some of the blame can be placed on the early stage of feature extraction. Namely, the Laplacianbased affine region detector is not well adapted to the giraffe texture whose blobs have a relatively complex shape. At the learning stage, the system also appears to be "distracted" by background features, such as sky and trees, that occur more commonly in training samples of giraffes than of the other animals. In the bottom image of Figure 4.8, "giraffe-ness" is associated with some parts of the background, as opposed to the animals themselves. Part of the problem is that the artificial "background" class is simply too inhomogeneous to be successfully represented in the mixture framework. A principled solution to this problem would involve partitioning the background into a set of natural classes (e.g., grass, trees, water, rocks, etc.) and building larger training sets that would include these classes in different combinations.

Despite somewhat uneven performance, the results presented in this section are promising. Unlike many other methods suitable for modeling natural textures, ours does not require negative examples. The EM framework shows surprising aptitude for automatically separating positive areas of the image from negative ones, without the need for specially designed significance scores such as the ones used by Schmid [137].

4.3 Discussion

The first half of this chapter has presented an approach to texture classification that combines textons and co-occurrence relations using a discriminative maximum entropy framework. While the maximum entropy classifier has proven to be effective for a basic bag-offeatures image representation, augmenting this representation with co-occurrence relations did not produce any improvement. This negative result is echoed in the experiments of Section 4.2.2.1, where we have observed that, even though relaxation improves the labeling of individual regions, it is not required for effective classification of single-texture images. A similar observation was also made by Lu et al. [91] who found that segmenting scenes into their semantic components (or basic material categories) was not necessary for recognizing the scenes as a whole. Interstingly, these findings in the vision literature are consistent with the conventional wisdom in the document classification community, that simple orderless "bag-of-words" represententations [9, 102, 114] perform best for document classification.

It is possible that the whole-image classification task is intrinsically "simpler" than the region labeling task, and does not require detailed statistical modeling of spatial relations. It is also possible that current image databases are not challenging enough, in that all classes can be distinguished using purely local features. Maybe co-occurrence statistics do not work because existing datasets are too small to allow them to be learned reliably, or maybe stronger spatial relations are needed in order to observe an improvement in performance. These are all interesting questions that the current vision literature has not even begun to address. To investigate them further, it is necessary to conduct larger-scale tests on more difficult texture databases that include a wider variety of classes.

The two-stage texture representation framework presented in the second half of this chapter, while showing promise for weakly supervised learning, also has several important shortcomings that should be addressed in subsequent work. Perhaps the most fundamental one is the separation between the two modeling steps of learning the distribution of local appearance and finding the co-occurrence statistics. In particular, the co-occurrence statistics computed during the second step cannot back-propagate to influence the sub-class probability distributions computed in the first step. At testing time, once the posterior probabilities of different sub-class labels have been computed from the generative model of local appearance, the subsequent updates of the probability values no longer depend on the image data. This feature of the relaxation algorithm described in Section 4.2.1.3 is at least partially responsible for the segmentation artifacts present in our results. Our current formulation of co-occurrence relations also has several disadvantages: First, the correlation between two neighboring locations depends only on their labels, not on the pixel-level data; and second, correlations between sub-class labels of different textures cannot be estimated reliably in a weakly supervised framework, and must therefore be set by hand. At worst, this can reduce the co-occurrence values to the role of generic smoothness constraints, which blindly enforce continuity in the final segmentation, regardless of the pixel-level evidence. A possible solution to many of the above problems is to represent the image as a *conditional random field* (CRF) [70, 68]. CRFs provide a principled framework for connecting low-level features with higher-level labels, and for combining local features and relations within the same probabilistic model. The major barrier to adopting CRFs to our problem is the lack (to our knowledge) of efficient learning methods for the weakly supervised setting. This is an extremely challenging problem, one that we hope to address in future research.

CHAPTER 5

Semi-Local Parts for Object Recognition

Achieving true 3D object recognition is one of the most important challenges of computer vision. As a first step towards this goal, it is necessary to develop geometrically invariant object models that can support the identification of object instances in novel images in the presence of viewpoint changes, clutter, and occlusion. This chapter describes an approach to object recognition that represents objects using dictionaries of composite *semi-local parts*, or geometric configurations of regions that are stable across a range of views of an object, and also across multiple instances of the same object category. We can think of each individual semi-local part as corresponding to an approximated planar and rigid region of the object. For example, the semi-local parts for representing side views of cars shown in Figure 5.1 (a) cover different rigid components (front, back, middle, or just a wheel) that any two given vehicles are likely to share in common. Moreover, because semi-local parts do not have to cover the whole object and can move freely with respect to one another, collections of them are in principle suitable for describing even non-rigid 3D objects, such as the owls in Figure 5.1 (b).

The rest of this chapter is organized as follows. Section 5.1 motivates our work, and Section 5.2 describes our weakly supervised procedure for creating dictionaries of semi-local parts for representing multiple object classes. This procedure first performs two-image matching to create *candidate parts* (Section 5.2.1) and then *validates* these parts against additional images from positive and negative classes (Section 5.2.2). Section 5.3 presents recognition



(b) Owls

Figure 5.1 Collections of semi-local parts for representing two object classes. The pairs of images that were originally matched to obtain the part are shown one above the other with the matched local features (yellow circles) superimposed. The examples shown here use the scale-invariant local features described in Section 5.3.2 and a scale-and-translation alignment model. The aligning transformation between the corresponding groups of features is indicated by the bounding boxes: the axis-aligned box in the top image is mapped onto the parallelogram in the bottom image. (As explained in Section 5.2.1, we use an affine alignment model and then discard any transformation that induces too much distortion.) Note that the boxes are shown for visualization purposes only; they are not required as input to the matching algorithm.

experiments with affine- and scale-invariant parts (Sections 5.3.1 and 5.3.2, respectively). In our recognition experiments, we use two different discriminative models: in Section 5.3.1, where relatively little training data is available, we use a simple voting scheme to classify images according to which class they contain; in Section 5.3.2 we work with larger training sets that allow us to learn a maximum entropy classifier that was first introduced in Section 4.1. Finally, the chapter closes with a discussion of several conceptual issues relevant to our approach. The material presented in this chapter has been published in [73, 74].

5.1 Motivation

In recent literature, perhaps the most effective approach to weakly supervised learning of part-based object models is Bayesian [28, 33, 162, 163]. Despite their promise, current Bayesian approaches have substantial shortcomings that limit their applicability to general 3D object recognition. In this chapter, we take aim at two of these shortcomings: lack of geometric invariance and an inefficient strategy for dealing with the correspondence problem.

Geometric invariance. To date, the state of the art for Bayesian approaches is scale and translation invariance [33], which means that they are mostly applicable to recognizing fronto-parallel views of upright objects. Unfortunately, while achieving scale invariance already requires a very intricate generative probabilistic framework, a principled probabilistic treatment of rotation or affine invariance is orders of magnitude more complex [78]. Even if one assumes a Gaussian joint distribution on the coordinates of a set of image features, the distribution in the affine-normalized space obtained by expressing all points in the basis given by a reference triple becomes non-Gaussian. Moreover, the normalized positions become correlated even if one starts out with uncorrelated errors in individual feature positions. On the other hand, if one assumes a more direct geometric approach to invariance (see, e.g., [36]), one gains access to robust and simple methods for computing affine-invariant object representations.

The correspondence problem. When attempting unsupervised learning of part-based models, one must inevitably confront the intractable combinatorial problem of determining *correspondence*, or the assignment of object parts to image regions in the presence of occlusion and clutter. In the typical generative approach to recognition, the *Expectation* Maximization (EM) algorithm is used for estimating the parameters of probabilistic object models. EM treats correspondence as missing data to be integrated out, which in principle involves computing expectations over the exponentially large space of all possible correspondences, but in practice is usually handled by a variety of approximation techniques. The combinatorics of EM-based learning severely limit the total number of parts in an object model. For example, the system developed by Fergus et al. [33] is, according to its authors, limited to models consisting of up to 6 or 7 parts learned from images containing 20 to 30 features. In turn, this limits both the expressiveness of the corresponding object models and the amount of clutter and/or occlusion that can be tolerated during learning. Perhaps as importantly for us, the "missing data" approach to correspondence is not satisfying from a philosophical standpoint. For humans, determining correspondence is a crucial step in forming judgments of similarity, which in turn influence such complex cognitive processes as memory and problem solving [43]. For computer vision algorithms, the ability to establish perceptually plausible correspondences is a convincing demonstration of success in the recognition task, one that has yet to be consistently achieved. Therefore, the approach proposed in this chapter is built on the idea that establishing *unique* correspondence between model and image features (as opposed to averaging over all possible correspondences) is central for successful recognition.

5.2 Semi-Local Parts

For each object class, we construct a dictionary of composite *semi-local parts* [73], or groups of several nearby regions whose appearance and spatial configuration occurs repeatably in the training set. The key idea is that consistent occurrence of (approximately) rigid groups of simple features in multiple images is very unlikely to be accidental, and must thus be a strong cue for the presence of the object. Note that, unlike the parts used in [1, 28, 33, 162, 163] (points or individual regions), our semi-local parts are not atomic, but are made up of multiple regions, each of which is associated with its own shape (circle or ellipse) and intensity pattern. Making parts composite improves their expressiveness and distinctiveness, and provides a layer of abstraction in representing complex 3D objects. We also propose a direct geometric procedure for learning parts from small collections of input images. Our strategy is to reduce the intractable problem of simultaneous alignment of multiple images to pairwise matching: *candidate parts* are formed by matching pairs of images, followed by a *validation* step to discard spurious matches (Figure 5.2). Even though finding optimal correspondence between features in two images is still intractable [46], effective suboptimal solutions can be found using a non-exhaustive constrained search procedure that relies on a number of strong geometric and appearance-based consistency constraints. As a result of these constraints, our matching algorithm is able to return hundreds of candidate parts consisting of a dozen or more features each. Note that the sizes of the parts are not fixed in advance, but are automatically determined by the matching procedure.

5.2.1 Matching of Pairs of Images

As explained above, the fundamental operation in our approach to learning object parts is matching pairs of images from the same class. Algorithm 1 describes our procedure for two-image matching and Figure 5.3 illustrates its major steps. This algorithm is similar to classical alignment search [27, 41, 47, 54], but it uses strong appearance (descriptor similarity) Input: Sets of features in two images.

Output: A list of candidate parts.

for each feature in the first image do

Find a list of potential appearance-based matches in the second image.

end for

Initialize seed groups:

for each feature i in the first image and each pair j, k of its neighbors do

Enumerate all triples i', j', k' in the second image such that i' (resp. j', k')

is a potential match of i (resp. j,k) and j', k' are neighbors of i' (Figure 5.3, top).

end for

Grow seed groups:

for each seed group of matching features \mathbf{do}

repeat

Estimate affine transformation that aligns corresponding features in both images

(Figure 5.3, middle).

 ${\bf if}$ transformation is geometrically consistent ${\bf then}$

Align two images using the estimated transformation.

Search for additional consistent matches in the neighborhood of existing group.

Add these matches to the group (Figure 5.3, bottom).

end if

until transformation is no longer consistent or no more matches can be found.

if current group is larger than a minimum size then

Add the group to the list of candidate parts.

end if

end for

Algorithm 1: Two-image matching. For specificity, we give here our affine alignment algorithm, but handling of other transformation models is straightforward (see text).



Figure 5.2 Our procedure for creating semi-local parts: A candidate part is found by matching two images from the same class and validated by matching against additional training images.

constraints to reduce the space of all possible correspondence hypotheses. Initially, for each feature in one image, we create a short list of *potential matches* in the other image by setting a threshold on descriptor similarity. Then we enumerate all possible seed groups of the smallest size necessary to estimate an aligning transformation between the two images. The most general model considered in our work is affine, for which seed groups consist of triples of matches (Figure 5.3, top). Note, however, that it is straightforward to handle more restricted 2D alignment models, such as scaling or similarity, within the same computational framework. In our implementation, we use linear least squares to estimate an affine alignment between the two groups of regions, and then enforce additional geometric constraints by rejecting any transformation that deviates too much from the desired model. For example, for the scaling-only model used in Section 5.3.2, we reject transformations that include too much skew, rotation, and anisotropic scaling. After estimating an alignment for a given seed group (Figure 5.3, middle), we conduct a greedy search in the same local neighborhood for additional matches satisfying geometric and appearance consistency constraints (Figure 5.3, bottom). Geometric consistency is measured by comparing the shapes of the corresponding features aligned in the same coordinate system. For affine features, we place constraints on the residual of the linear transformations between the centers of the corresponding ellipses,



Step 1: An initial seed triple. Arrows represent neighborhood constraints and dotted lines represent appearance consistency constraints.



Step 2: Estimate an affine transformation aligning (i', j', k') with (i, j, k). Features from image 2 mapped onto their corresponding features in image 1 are shown by dashed outlines.



Step 3: Grow the group by two additional matches that satisfy geometric and appearance consistency constraints.

Figure 5.3 Illustration of three steps of Algorithm 1.

as well as on the difference in their orientations and axis lengths. The correspondence search terminates when the alignment is no longer geometrically consistent, or when no further consistent matches can be found. Note that the number of features in the group (the size of the candidate part) is determined automatically as a result. Moreover, since Algorithm 1 generates a large number of seed groups and attempts to grow them all, multiple candidate parts can be returned as a result of just a single matching operation. In the implementation, we use a post-processing step to merge candidate parts that ovelap by a significant amount.

The candidate parts returned by the above procedure are (roughly) affinely rigid by construction, i.e., the mapping between two different instances of the same part can be well approximated by a 2D affine transformation. Note that we do not make the overly restrictive assumption that the entire object is planar and/or rigid — it is sufficient for the object to possess some (approximately) planar and rigid components. Because of this non-global notion of affine invariance, our method is suitable for modeling rigid 3D objects (Figure 5.4), as well as non-rigid objects such as faces (Figure 5.5). Also refer back to Figure 5.1 for matching examples using scale-invariant features and a scale-and-translation alignment model.

In existing literature, similar procedures for growing groups of matches based on geometric and appearance consistency constraints have been successfully applied to the recognition of the same object instance in multiple views [34]; one of the key insights of our work is that such procedures are also quite effective for building models of object classes with substantial intra-class variation. Because of the strong consistency constraints that must be satisfied by semi-local parts, they are much more discriminative than atomic parts, and much less likely to give rise to false detections.

Note that our two-image matching procedure can match an image to itself, and is thus directly applicable to the problem of detecting repeated structures and symmetries within a single image [79, 86, 151]. The only change necessary in the implementation is the addition of



Figure 5.4 Candidate parts for three 3D objects: (a) a salt can, (b) a shoe, and (c) a car. The two input images to the matching procedure are shown side by side, with the matched ellipses superimposed. For visualization purposes, we also show bounding boxes around the matched ellipses: the axis-aligned box in the left image is mapped onto the parallelogram in the right image by the affine transformation that aligns the matched ellipses.

checks to prevent trivial hypotheses that match (almost) every region to itself. A particularly attractive feature of our matching procedure for this application is its ability to discover patterns in substantial amounts of clutter, which is not possible with other, more specialized approaches, e.g., [86]. Figure 5.6 shows the symmetries and repetitions detected in several images. The examples of repetitions detected in the images of fishes and badgers suggest that a single image may be sufficient to construct highly discriminative models for some object categories. This is certainly true for butterflies, which will be used as test subjects in Section 5.3.1.


(b) Grace Kelly

(c) Grace Kelly and Cary Grant

Figure 5.5 Candidate parts for face images. (c) Curiously, the face of Grace Kelly in the left image fails to match to her face in the right image, but the matched regions do reveal a structural similarity between her face and Cary Grant's.

5.2.2 Validation

After candidate parts have been formed, they are *validated* by matching against additional training images. The purpose of validation is to reject spurious parts that arise from clutter and incorrect matching hypotheses. We have experimented with two types of validation: *non-discriminative* (Section 5.3.1), where a part is only matched against *positive images* (i.e., images that contain the same object class that was used to generate the candidate part); and *discriminative* (Section 5.3.2), where a part is matched against both positive and negative images. The latter procedure has the advantage of rejecting parts that occur just as often in non-object images as in object images. For example, a wheel may be a part that occurs in both cars and trucks, and can thus be used in the description of either class, but if we are trying to discriminate between cars and trucks, then the presence of a wheel is not a useful cue.



(d) Badgers: repetition

(e) Fishes: repetition

Figure 5.6 Detecting symmetry (top) and repetition (bottom) in single images. In the top images, the axis of reflective symmetry is shown as a dashed line. Note that for the palace window, two different local symmetries are detected.

The procedure for detecting a part in a validation image is very similar to two-image matching (Algorithm 1), made simpler and more efficient by the part's relatively small size and (presumed) lack of clutter. A detected instance of a candidate part in a validation image may have multiple features missing because of occlusion, failure of the salient region detector, etc. We define the *repeatability* of an individual feature within a part as the proportion of detected part instances in positive images in which that feature was present, and refine parts by removing from them all features whose repeatability falls below a certain threshold. Note that in principle, we can use the correspondences between the parts and the positive validation images to further refine the parts by modifying their appearance and shape. At present, we have not implemented this extension because we have generally been satisfied with the quality of parts obtained following validation and removal of spurious features.

For each detected instance of a part in a validation image, we define a *response score* as the number of features in that instance normalized by the total number of features in the part. Next, for each part k and each validation image I, we define a global score $\rho_k(I)$ as the



Training images



Figure 5.7 An example of a face part (see Figure 5.12) and its response histograms for discriminative validation. The solid red line (resp. dash-dot blue line) indicates the response histogram for the positive (resp. negative) class.

highest response score over all detected instances of part k in that image. This implicitly assumes that an object image can contain at most one instance of each part. In the future, we plan to improve our feature representation to allow for multiple detected instances of the same part. This would allow us to perform more accurate localization for classes such as cars (which have two wheels) or faces (which have two eyes). Finally, note that if part k is not detected in image I at all, we have $\rho_k(I) = 0$.

For non-discriminative validation, we derive an overall "quality" measure of a part from its average response score over the validation set, and retain a fixed number of top-scoring parts for each class. For discriminative validation, we compute a quality measure for the part by taking the χ^2 distance between the histograms of part responses over the positive and negative images (see Figure 5.7 for an example). This validation score can range from 1, when the two histograms have no overlap at all, to 0, when they are identical. Just as for non-discriminative validation, a fixed number of highest-scoring parts is retained to represent each class.

5.2.3 Recognition

The part creation, detection, and validation procedures presented so far in this chapter is general in the sense that it can be used with a wide variety of learning and classification schemes. To complete the discussion of our part-based framework, let us briefly sketch the testing or recognition step of our framework while deferring most of the details to our presentation of experimental results in Section 5.3. Given a novel test image I, we perform the detection procedure for each part k from the dictionary, and, just as during the validation step, record the response score $\rho_k(I)$ of this part for this image. Next, our goal is to decide which object instance is present in the image based on the vector of response scores for all parts. In the experiments, we use several ways of accomplishing this goal. In Section 5.3.1, we simply let each part "vote" for the class it represents with its response score, and assign the test image to the class with the highest cumulative vote. While this approach is somewhat ad-hoc, it does not require a separate step of training a discriminative model over the response scores, and therefore it works well in situations when very few training images are available. On the other hand, in Section 5.3.2, we work with larger datasets, and we are able to train two classifiers. The first is a Naive Bayes classifier that works by estimating $P(\rho_k|c)$, the probability distributions of response scores for each part k given that the image contains an object of class c. The second is the maximum entropy classifier first described in Section 4.1 that uses feature functions based on individual response scores as well as on the overlap between pairs of parts. Details of these feature functions and our training protocol will be further discussed in Section 5.3.2.

5.3 **Recognition Experiments**

This section demonstrates the promise of our approach with recognition experiments on three datasets. Section 5.3.1 presents results on a dataset consisting of seven classes of butterflies. The large amount of geometric variability in this dataset requires us to use



Figure 5.8 The butterfly dataset. Three samples of each of the seven classes are shown. Monarch 1 and 2 are the same butterfly species with wings closed and open, respectively. This database is publicly available at http://www-cvr.ai.uiuc.edu/ponce_grp/data.

a full affine alignment model. Section 5.3.2 considers two datasets that have wider intraclass appearance variability, but a narrower range of geometric variation that can be handled with a scale-and-translation alignment model. The first is a four-class subset of Caltech6 [33] consisting of airplanes, cars, faces, and motorbikes; and the second is a dataset consisting of six species of birds.

5.3.1 Affine-Invariant Parts

In this section, we report results for multi-class recognition as well as binary detection on a dataset composed of 619 images of butterflies collected from the Internet. Seven butterfly classes are represented (see Figure 5.8 for sample images of each). The dataset is extremely diverse, with pictures varying widely in terms of size and quality (motion blur, lack of focus, resampling and compression artifacts are among the factors affecting quality). From one viewpoint, butterflies may be considered pretty "toy subjects" for testing the descriptive power of our local affine models, since the geometry of a butterfly is locally planar for each wing (though usually not globally planar). In addition, the species identity of a butterfly is determined by a basically stable geometric wing pattern, though appearance can be significantly affected by individual variations, lighting, and imaging conditions. But from another viewpoint, recognizing butterflies is a challenging task. For one, a typical butterfly part is sufficiently complex to require at least a dozen features to be discriminative, which is beyond the range of most of the current state-of-the-art recognition systems [1, 28, 33, 162, 163]. Moreover, the levels of invariance (translation and scale) possessed by existing algorithms are clearly insufficient for recognizing insects in natural imagery. In our dataset, butterflies appear in highly cluttered environments, with most images containing on the order of a thousand local features.

For the experiments of this section, we perform feature extraction using the affine-adapted Laplacian detector. We chose this detector over the Harris because it tends to find regions that are centered away from object boundaries and because blobs are clearly salient for butterflies, whose wing patterns feature spots and stripes. As in Section 3.1, we use SPIN and RIFT descriptors to compare regions. Note that the RIFT descriptor as originally introduced in that section is not invariant to reflection of the normalized patch. Unlike in our texture recognition work, we require reflection invariance for finding semi-local parts. Reflection of the normalized patch reverses the order of directions in the histogram of gradient orientations within each circular subregion of the RIFT descriptor. Thus, when finding the distance between two RIFT descriptors, we simply take the minimum over both orders. One other issue is how to combine SPIN and RIFT descriptors in determining the appearance-based dissimilarity or *matching score* between two patches. We set the matching score to be the *minimum* of the Euclidean distances between the two spin images or RIFT descriptors (note that both descriptors are normalized to have zero mean and unit norm, thus Euclidean distances between them lie in the same range and are comparable). Empirically, this approach performs better than other ways of combining descriptors, since it provides robustness against instabilities in region extraction and intensity normalization. The descriptors (particularly

Class	Part size		Test images	Correct (rate)	
Admiral	179	(12/28)	85	74	(0.871)
Swallowtail	252	(18/29)	16	12	(0.750)
Machaon	148	(12/21)	57	55	(0.965)
Monarch 1	289	(14/67)	48	35	(0.729)
Monarch 2	275	(19/36)	58	53	(0.914)
Peacock	102	(8/14)	108	108	(1.000)
Zebra	209	(16/31)	65	58	(0.892)
Total			437	395	(0.904)

Table 5.1 Classification results for the butterflies. The second column shows the total part size for each class (the sum of sizes of individual parts), and the size of the smallest and the largest parts are listed in parentheses. For this database, the kernel-based bag-of-features method of Section 3.2 achieved 87% accuracy, below our rate of 90.4%.

spin images) can be sensitive to transformations of the intensity values (i.e., noise, JPEG compression, sharpening) and to shifts in the position of the center of the normalized patch, so a large distance between two spin images or two RIFT descriptors is not always a reliable indication of perceptual difference. Thus, in determining the matching score between two patches, it makes sense to trust the descriptor that produces the lower distance.

For modeling, 16 images from each class were chosen *at random*. Candidate parts were formed by matching between eight pairs (an alternative approach would have been to match every possible pair of images in the training set, but this did not prove necessary, as sufficient numbers of candidate parts were obtained from the eight pairs). Ten positive verification images were used for each class to remove unstable regions and to rank parts according to their non-discriminative validation score. Top ten parts were retained and used for recognition.

Multi-class classification is performed as follows. First, the parts for *all* classes are detected in each training image. Though multiple instances of the same part may be found, we



Figure 5.9 Classification rate vs. number of parts.

retain only the single instance with highest *absolute response score*, or the number of detected features. The cumulative score for a given class is given by the *relative response* of all its parts, or the total number of features detected in all parts divided by the sum of part sizes. Each image is assigned to the class having the maximum relative response score. Table 5.1 shows classification results obtained using the above approach. Our overall classification rate is 90.4%. This is higher than the baseline rate of 87% obtained with the kernel-based bag-of-features method of Section 3.2 using rotation-invariant (H+L)(SPIN+SIFT) features.¹ Note that our relative response score tends to favor classes with smaller part size, as larger parts tend to be much "looser" than smaller ones: that is, the fraction (though not the absolute number) of regions they pick up tends to be smaller. Figure 5.9 shows how the performance of our system is affected by using different numbers of parts for classification. Though the classification rate is already respectable with just a single part, using multiple parts is clearly important for improving performance.

¹Several feature combinations and levels of invariance were tested, and the feature combination reported here is the one that obtained the highest performance.



Figure 5.10 Butterfly modeling and detection. (a) The part with the highest validation score for each class. (b) Example of detecting the part from (a) in a single test image. Detected regions are shown in yellow and occluded ones are reprojected from the model in blue. The total number of detected features (absolute response) and the corresponding relative response are shown below each image. Note that the correspondences for the swallowtail and zebra are incorrect. (c) ROC curves for detection. Three thresholds for relative response (0.33, 0.5, 0.66) are marked on the curve. The dark square marks the ROC equal error rate, which is listed in parentheses next to the threshold value at at which it is attained.

We can get an alternative assessment of performance by considering the binary detection task, where for each image and each class, we ask whether an instance of the from this class is present. This decision can be made by setting a threshold on the relative response score. By considering all possible thresholds, we get an ROC curve, a plot of the true positive rate (the number of correct positives detected over the total number of positives for the class) vs. the false positive rate (the number of false positives over the total number of negatives in the dataset). These curves, given in Figure 5.10, also show the ROC *equal error rate* for each class. This is the true positive rate that is equal to one minus the false positive rate. The equal error rates achieved by our system are high, showing that detection can indeed be performed successfully.

5.3.2 Scale-Invariant Parts

This section presents recognition results obtained on two multi-class object databases. The first is a subset of the publicly available CalTech database [33], which we have already used in Section 3.2. We have taken 300 images each from four classes: airplanes, rear views of cars, faces, and motorbikes (for examples of Caltech images, refer back to Figure 3.13). The second database, which we collected from the Web, consists of 100 images each of six different classes of birds: egrets, mandarin ducks, snowy owls, puffins, toucans, and wood ducks (Figure 5.11). Because these databases mostly feature objects in canonical poses (i.e., upright heads, frontal or side views of birds), they do not require the full affine alignment model; scale with translation are sufficient.

For our texture recognition experiments of Chapter 3 and butterfly experiments of Section 5.3.1, Laplacian region detectors have proven to be successful, since most of the classes in these experiments could be characterized quite well by salient blob-like features. However, we have found Laplacian features to be much less repeatable for classes with complex internal structures, e.g., eyes, wheels, heads, etc. Instead, in this section we extract salient regions



Birds database

Figure 5.11 Three examples from each class of the birds database. This database is publicly available at http://www-cvr.ai.uiuc.edu/ponce_grp/data.

using the scale-invariant detector of Jurie and Schmid [61], which finds salient circular configurations of edge points, and is robust to clutter and texture variations inside the regions. Just as in Section 4.1.2, the appearance of the extracted regions is represented using SIFT descriptors.

For the Caltech database, 50 randomly chosen images per class are used for creating candidate parts. Each image is paired up to two others, for a total of 100 initialization pairs. Of the several hundred candidate parts yielded by this matching process, the 50 largest ones undergo discriminative validation. Specifically, these candidate parts are matched against every image from the validation set, which also contains 50 randomly chosen images per class, and 20 highest-scoring parts per class are retained to form the part dictionary. Finally, the remaining 200 images per class make up the test set. We follow the same protocol for the bird dataset, except that 20 images per class are used for finding candidate parts, another 30 for validation, and the remaining 50 for testing.

Figures 5.12 and 5.13 illustrate training and validation (part selection). As can be seen from the plots of validation scores for all selected parts, the quality of part dictionaries found for different classes varies widely. Extremely stable, repeatable parts are formed for faces,



Figure 5.12 Learning part vocabularies for the CalTech database. (a) The highest-scoring part for each class. (b) Response histograms for the top part. The solid red line (resp. dashed blue line) indicates the histogram of response scores of the part in all positive (resp. negative) training images. Recall that the validation score of the part is given by the χ^2 distance between the two histograms. (c) Plots of top 20 part scores following validation.



Figure 5.13 Learning part vocabularies for the birds database. (a) The highest-scoring part for each class superimposed on the two original training images. (b) Validation response histograms for the top parts. (c) Validation scores for the top 20 parts from each class.

motorbikes, and ducks. The classes with the weakest parts are airplanes for the CalTech database and egrets for the bird database. Both airplanes and egrets lack characteristic texture, and often appear against busy backgrounds that generate a lot of detector responses (buildings, people, and airport machinery in case of planes, or grass and branches in case of egrets). In addition, both egrets and airplanes are "thin" objects, so the local regions that overlap the object also capture a lot of background. Thus, the SIFT descriptors computed over these regions end up describing mostly clutter. To alleviate this problem, we plan to experiment with alternative descriptors that capture the shape of the edges close to the boundary of the scale-invariant regions [61], as opposed to the internal texture, as the SIFT descriptor does. Note that our part selection framework is suitable for choosing between semi-local parts based on different descriptors, since it abstracts from the low-level details of matching (i.e., how appearance similarity is computed, what aligning transformation is used, or how the correspondence search is performed), and looks only at the end result of the matching on the training set (i.e., how repeatable the resulting parts are, and whether they can be used to distinguish between positive and negative examples for a given class).

The parts obtained for classes other than airplanes and egrets have higher scores and capture much more salient object features. Interestingly, though, for cars, even the highestscoring part includes spurious background detections along the horizontal line at the eye level of the image. This comes from the relative visual monotony of the car class: all the rear views of cars were apparently captured through the windshield by a person in the front seat. Thus, the "horizon" formed by the converging sides of the road is approximately in the same location in all the images, and the scenery at the roadside (trees, buildings) gives rise to a lot of features in stable positions that are consistently picked up by the matching procedure.

In this section, we use the discriminative maximum entropy framework introduced in Section 4.1.1 to build a principled probabilistic model for object recognition. Recall from Section 4.1.1 that the use of the maximum framework requires us to define a set of *feature* functions for each class. For each part k and each training image I, we compute a normalized feature function based on its response score $\rho_k(I)$ as defined in Section 5.2.2:

$$g_k(I) = \frac{\rho_k(I)}{\sum_{k'} \rho_{k'}(I)}.$$

Feature function values computed on images from the validation set are used as training data to estimate the parameters of the exponential model.

Just as in the texture recognition experiments of Section 4.1, we also investigate whether, and to what extent, incorporating relations into the object representation improves classification performance. To this end, we define *overlap* relations between pairs of parts that belong to the same class. Let $\omega_{k,\ell}(I)$ be the overlap between detected instances of parts kand ℓ in the image I, i.e., the ratio of the intersection of the two parts to their union. This ratio ranges from 0 (disjoint parts) to 1 (coincident parts). Then we define

$$g_{k,\ell}(I) = \frac{\omega_{k,\ell}(I)}{\sum_{k',\ell'} \omega_{k',\ell'}(I)}.$$

The overlap relations are very flexible — in effect, they enforce only spatial coherence. This flexibility potentially allows us to deal with non-rigid and/or articulated objects. In the future, we plan to experiment with more elaborate relations that take into account the distance, relative scale, or relative orientations of the two parts [1]. Finally, it is important to note that we currently do not use feature selection techniques to reduce the number of overlap relations within the exponential model. Because of the small size of the part dictionaries used in the experiments presented in the next section (namely, 20 parts per class), the resulting number of overlap relations (190 per class) is quite manageable, unlike in the texture recognition experiments of Section 4.1, where we had to contend with millions of potential co-occurrence relations.

Tables 5.2 and 5.3 show classification performance of several methods with 20 parts per class. The first column of the tables shows the performance of a baseline Naive Bayes

CalTech	Naive	Exp.	Exp.	Exp. parts &	
database	Bayes	parts	relations	relations	
Airplanes	98.0	88.0	78.0	87.5	
Cars (rear)	95.5	99.5	90.5	99.5	
Faces	96.5	98.5	96.5	98.0	
Motorbikes	97.5	99.5	83.0	99.5	
All classes	96.88	96.38	87.0	96.13	

 Table 5.2 Classification rates for the CalTech database using 20 parts per class.

 approach with likelihood given by

$$P(I|c) = \prod_{k} P(\rho_k(I)|c)$$

The distributions $P(\rho_k|c)$ are found by histogramming the response scores of part k on all training images from class c. Note that we take into account the responses of parts on images from all classes, not only the class which they describe. Roughly speaking, we expect $P(\rho_k(I)|c)$ to be high if part k describes class c and $\rho_k(I)$ is high, or if part k does not describe class c and $\rho_k(I)$ is low or zero. Thus, to conclude that an object from class c is present in the image, we not only have to observe high-response detections of parts from class c, but also low-response detections of parts from other classes. The exponential model, which encodes the same information in its feature functions, also uses this reasoning.

The second (resp. third, fourth) columns of Tables 5.2 and 5.3 show the classification performance obtained with exponential models using the g_k features only (resp. the $g_{k,\ell}$ only, g_k and $g_{k,\ell}$ combined). For the CalTech database, the Naive Bayes and the exponential parts-only models achieve very similar results, though under the exponential model, airplanes have a lower classification rate, which is intuitively more satisfying given the poor part dictionary for this class. Note that our classification accuracy of over 96% on the four CalTech classes is comparable to other recently published results [20, 23]. For the bird database, the

Birds	Naive	Exp.	Exp.	Exp. parts &	
database	Bayes	parts	relations	relations	
Egret	68	90	72	88	
Mandarin	66	90	66	90	
Snowy owl	66	98	52	96	
Puffin	88	94	94	94	
Toucan	88	82	82	82	
Wood duck	96	100	86	100	
All classes	78.67	92.33	75.33	91.67	

Table 5.3 Classification rates for the birds database using 20 parts per class. For this database, the kernel-based bag-of-features method of Section 3.2 achieved 83% accuracy, below our best result of 92.33%.

exponential model outperforms Naive Bayes. Moreover, our best classification rate of 92.33%is higher than the best rate of 83% obtained with the kernel-based bag-of-features method of Section 3.2 using scale-invariant (H+L)(SIFT) features. For both databases, relations-only features alone perform considerably worse than the parts-only features, and combining partsbased with relation-based features brings no improvement. Figure 5.14 shows a plot of the classification rate for the exponential model as a function of part dictionary size. Note that the curves are not monotonic — adding a part to the dictionary can decrease performance. This behavior may be an artifact of our scoring function for part selection, which is not directly related to classification performance. In the future, we plan to experiment with part selection based on increase of likelihood under the exponential model [6].

Though we did not conduct a quantitative evaluation of localization accuracy, the reader may get a qualitative idea by examining Figures 5.15 and 5.16, which show examples of part detection on several test images. A poorer part vocabulary for a class tends to lead to poorer localization quality, though this is not necessarily reflected in lower classification



Figure 5.14 Classification rate (exp. parts) as a function of dictionary size: CalTech database (left), birds database (right). For the CalTech database, because three of the four classes have extremely strong and redundant parts, performance increases very little as more parts are added. For the bird database, diminishing returns set in as progressively weaker parts are added.

rates. Specifically, an object class represented by a relatively poor part vocabulary may still achieve a high classification rate, provided that parts for other classes do not generate too many false positives on images from this class. The airplane example in Figure 5.15 (b) is a good illustration of this phenomenon: only three airplane parts are detected in this image, yet the airplane is recognized correctly since the image does not contain enough clutter to generate false detections of parts from other classes.

Somewhat frustratingly, we have found that inter-part relations do not improve recognition performance. This is consistent with the findings of Section 4.1, where texton cooccurrence relations also did not improve the accuracy of maximum entropy texture classification. From examining the part detection examples in Figures 5.15 and 5.16, it seems intuitively clear that the pattern of overlap of different part instances encodes useful information: the part detections that lie on the object tend to be clustered close together, while false detections are frequently scattered all over the image. We currently conjecture that the



Figure 5.15 CalTech results. (a), (b) Two examples of correctly classified images per class. Left of each column: original image. Right of each column: transformed bounding boxes of all detected part instances for the given class superimposed on the image. (c) Examples of misclassified images. Note that localization is poor for airplanes and very good for faces (notice the example a changed facial expression). For motorbikes, the front wheel is particularly salient. Out of the entire test set, only one bike image was misclassified, and in it the front wheel is not properly visible.

overlap information may be more useful for localization than for recognition. In the future, we plan to evaluate localization quantitatively.

5.4 Discussion

In this chapter, we have presented a weakly supervised framework for modeling 3D objects in terms of geometrically invariant *semi-local parts*. The two-image matching procedure that forms the core of our method is also applicable to identifying repeated structures and symmetries within a single image — an interesting application which today is rarely treated



Figure 5.16 Birds database results. (a), (b) Two examples of successfully classified images per class. The original test image is on the left, and on the right is the image with superimposed bounding boxes of all detected part instances for the given class. Notice that localization is fairly good for mandarin and wood ducks (the head is the most distinctive feature). Though owl parts are more prone to false positives, they do capture salient characteristics of the class: the head, the eye, and the pattern of the feathers on the breast and wings. (c) Misclassified examples. The wood duck class has no example because it achieved 100% classification rate.

in the same context as recognition (see [85] for a model-based recognition approach that takes advantage of symmetries). For our primary goal of 3D object recognition, the proposed approach has the advantages of robustness and flexibility. Namely, it is capable of learning multiple variable-sized parts from images containing a significant amount of noise and clutter. The promise of our part-based method is also demonstrated by its ability to outperform the bag-of-features method of Chapter 3 on our butterfly and bird databases. We conclude this presentation with a discussion of several conceptual issues relevant to our work.

Alignment. The search algorithm described in Section 5.2.1 is reminiscent of the alignment techniques used in model-based vision [27, 41, 47, 54]. While the process of detecting an existing part in a test image may indeed be thought of as alignment (albeit using a novel representation based on constellations of local features), our overall modeling approach is not. Whereas in classical alignment globally rigid models are built manually and/or from segmented images, our method is capable of handling unsegmented, heavily cluttered input. Another key departure from classical alignment is that our algorithm does not seek a *global* transformation between two images, nor does it assume that the object being modeled is either planar and/or globally rigid. Instead, we exploit the fact that smooth surfaces are planar in the small, and that local affine models are sufficient to handle large viewpoint variations for approximately coplanar, close-by patches, as well as small non-rigid transformations of the layout of these patches. Because the part model is local, it is also robust to large global deformations.

Training set size. Our procedure for learning semi-local parts requires just a pair of images for candidate part creation, and a relatively small number (up to thirty) for validation. Recently, it has been observed that very few training images are actually necessary for learning object models provided the learner is equipped with a good prior on the parameters of the model [28]. In our case, the "prior" is the strong notion of visual similarity, making it possible to learn part-based models from extremely small training sets. Modeling the background. Many previous approaches to object detection, whether classifier-based (discriminative) [1] or probabilistic (generative) [28, 33, 162, 163], require an explicit model of the background. However, we believe that the notion of modeling the background is problematic, since intuitively, it is much harder to describe the concept of the "complement" of the class of interest than the class itself. By contrast, our approach completely avoids the necessity of modeling the background by virtue of its reliance on strong geometric consistency constraints. In our work, the background is implicitly assumed to be non-repeatable, an assumption that sometimes fails for extremely small sets of images, necessitating the use of the validation procedure.

Inter-part relations. The most important negative result of this chapter is the lack of performance improvement from overlap relations in Section 5.3.2. Recall that we have observed the same behavior for textons-and-relations models in Section 4.1. For object recognition, the lack of improvement can be ascribed, at least partly, to the weakness of our overlap relations, especially compared to the strong geometric consistency constraints encoded within semi-local parts. In the future, we plan to investigate geometric relations that capture more discriminative information, and to test their behavior for classification on additional object databases.

CHAPTER 6

Spatial Pyramid Matching for Scene Recognition

In this chapter, we consider the problem of recognizing the semantic category of an image. For example, we may want to classify a photograph as depicting a scene (forest, street, office, etc.) or as containing a certain object of interest. As a solution to this problem, we propose a "global" non-invariant scene representation based on aggregating statistics of local features over fixed subregions and a kernel-based recognition method that computes a rough geometric correspondence using an efficient approximation technique adapted from the *pyramid matching* scheme of Grauman and Darrell [45]. Our method involves repeatedly subdividing the image and computing histograms of local features at increasingly fine resolutions. As shown by experiments in Section 6.4, this simple operation suffices to significantly improve performance over a basic bag-of-features representation, and even over methods based on detailed geometric correspondence. The research presented in this chapter has been published in [76].

6.1 Motivation

Previous research has shown that statistical properties of the scene considered in a holistic fashion, without any analysis of its constituent objects, yield a rich set of cues to its semantic category [117]. Our own experiments confirm that global representations can be surprisingly effective not only for identifying the overall scene, but also for categorizing images as containing specific objects, even when these objects are embedded in heavy clutter and vary significantly in pose and appearance. This said, we do not advocate the direct use of a global method for object recognition (except for very restricted sorts of imagery). Instead, we envision a subordinate role for this method. It may be used to capture the "gist" of an image [150] and to inform the subsequent search for specific objects (e.g., if the image, based on its global description, is likely to be a highway, we have a high probability of finding a car, but not a toaster). In addition, the simplicity and efficiency of our proposed method, in combination with its tendency to yield unexpectedly high recognition rates on seemingly challenging data, could make it a good baseline for "calibrating" newly acquired datasets and for evaluating more sophisticated recognition approaches.

In computer vision, histograms have a long history as a method for image description (see, e.g., [136, 146]). Koenderink and Van Doorn [66] have generalized histograms to *locally* orderless images, or histogram-valued scale spaces (i.e., for each Gaussian aperture at a given location and scale, the locally orderless image returns the histogram of image features aggregated over that aperture). Our spatial pyramid approach can be thought of as an alternative formulation of a locally orderless image, where instead of a Gaussian scale space of apertures, we define a fixed hierarchy of rectangular windows. Koenderink and Van Doorn have argued persuasively that locally orderless images play an important role in visual perception. Our retrieval experiments (Figure 6.4) confirm that spatial pyramids can capture perceptually salient features and suggest that "locally orderless matching" may be a powerful mechanism for estimating overall perceptual similarity between images.

It is important to contrast our proposed approach with *multiresolution histograms* [48], which involve repeatedly subsampling an image and computing a global histogram of pixel values at each new level. In other words, a multiresolution histogram varies the resolution at which the features (intensity values) are computed, but the histogram resolution (intensity scale) stays fixed. We take the opposite approach of fixing the resolution at which the features are computed, but varying the spatial resolution at which they are aggregated.

This results in a higher-dimensional representation that preserves more information (e.g., an image consisting of thin black and white stripes would retain two modes at every level of a spatial pyramid, whereas it would become indistinguishable from a uniformly gray image at all but the finest levels of a multiresolution histogram). Finally, unlike a multiresolution histogram, a spatial pyramid, when equipped with an appropriate kernel, can be used for approximate geometric matching.

The operation of "subdivide and disorder" — i.e., partition the image into subblocks and compute histograms (or histogram statistics, such as means) of local features in these subblocks — has been practiced numerous times in computer vision, both for global image description [44, 147, 150] and for local description of interest regions [89]. Thus, though the operation itself seems fundamental, previous methods leave open the question of what is the right subdivision scheme (although a regular 4×4 grid seems to be the most popular implementation choice), and what is the right balance between "subdividing" and "disordering." The spatial pyramid framework suggests a possible way to address this issue — namely, the best results may be achieved when multiple resolutions are combined in a principled way. It also suggests that the reason for the empirical success of "subdivide and disorder" techniques is the fact that they actually perform approximate geometric matching.

6.2 Spatial Pyramid Matching

We first describe the original formulation of pyramid matching [45], and then introduce our application of this framework to create a *spatial pyramid* image representation.

6.2.1 Pyramid Match Kernels

Let X and Y be two sets of vectors in a d-dimensional feature space. Grauman and Darrell [45] propose *pyramid matching* to find an approximate correspondence between these two sets. Informally, pyramid matching works by placing a sequence of increasingly coarser grids over the feature space and taking a weighted sum of the number of matches that occur at each level of resolution. At any fixed resolution, two points are said to match if they fall into the same cell of the grid; matches found at finer resolutions are weighted more highly than matches found at coarser resolutions. More specifically, we construct a sequence of grids at resolutions $0, \ldots, L$, such that the grid at level ℓ has 2^{ℓ} cells along each dimension, for a total of $D = 2^{d\ell}$ cells. Let H_X^{ℓ} and H_Y^{ℓ} denote the histograms of X and Y at this resolution, so that $H_X^{\ell}(i)$ and $H_Y^{\ell}(i)$ are the numbers of points from X and Y that fall into the *i*th cell of the grid. Then the number of matches at level ℓ is given by the *histogram intersection* function [146]:

$$\mathcal{I}(H_X^{\ell}, H_Y^{\ell}) = \sum_{i=1}^{D} \min\left(H_X^{\ell}(i), H_Y^{\ell}(i)\right).$$
(6.1)

In the following, we will abbreviate $\mathcal{I}(H_X^{\ell}, H_Y^{\ell})$ to \mathcal{I}^{ℓ} .

Note that the number of matches found at level ℓ also includes all the matches found at the finer level $\ell + 1$. Therefore, the number of *new* matches found at level ℓ is given by $\mathcal{I}^{\ell} - \mathcal{I}^{\ell+1}$ for $\ell = 0, \ldots, L - 1$. The weight associated with level ℓ is set to $\frac{1}{2^{L-\ell}}$, which is inversely proportional to cell width at that level. Intuitively, we want to penalize matches found in larger cells because they involve increasingly dissimilar features.

Putting all the pieces together, we get the following definition of a *pyramid match kernel*:

$$\kappa^{L}(X,Y) = \mathcal{I}^{L} + \sum_{\ell=0}^{L-1} \frac{1}{2^{L-\ell}} \left(\mathcal{I}^{\ell} - \mathcal{I}^{\ell+1} \right)$$
(6.2)

$$= \frac{1}{2^{L}}\mathcal{I}^{0} + \sum_{\ell=1}^{L} \frac{1}{2^{L-\ell+1}}\mathcal{I}^{\ell}.$$
 (6.3)

Both the histogram intersection and the pyramid match kernel are Mercer kernels [45].

6.2.2 Spatial Matching Scheme

As introduced in [45], a pyramid match kernel works with an orderless image representation. It allows for precise matching of two collections of features in a high-dimensional appearance space, but discards all spatial information. This chapter advocates an "orthogonal" approach: perform pyramid matching in the two-dimensional image space, and use traditional clustering techniques in feature space.¹ Specifically, we quantize all feature vectors into M discrete types, and make the simplifying assumption that only features of the same type can be matched to one another. Each channel m gives us two sets of twodimensional vectors, X_m and Y_m , representing the coordinates of features of type m found in the respective images. The final kernel is then the sum of the separate channel kernels:

$$K^{L}(X,Y) = \sum_{m=1}^{M} \kappa^{L}(X_{m},Y_{m}).$$
(6.4)

This approach has the advantage of maintaining continuity with the popular "visual vocabulary" paradigm — in fact, it reduces to a standard bag of features when L = 0.

Because the pyramid match kernel (6.3) is simply a weighted sum of histogram intersections, and because $c \min(a, b) = \min(ca, cb)$ for positive numbers, we can implement K^L as a single histogram intersection of "long" vectors formed by concatenating the appropriately weighted histograms of all channels at all resolutions (Figure 6.1). For L levels and M channels, the resulting vector has dimensionality $M \sum_{\ell=0}^{L} 4^{\ell} = M \frac{1}{3} (4^{L+1} - 1)$. Several experiments reported in Section 6.4 use the settings of M = 400 and L = 3, resulting in 34000-dimensional histogram intersections. However, these operations are efficient because the histogram vectors are extremely sparse (in fact, just as in [45], the computational complexity of the kernel is linear in the number of features). It must also be noted that we did not observe any significant increase in performance beyond M = 200 and L = 2, where the concatenated histograms are only 4200-dimensional.

The final implementation issue is that of normalization. For maximum computational efficiency, we normalize all histograms by the total weight of all features in the image, in effect forcing the total number of features in all images to be the same. Because we use a

¹In principle, it is possible to integrate geometric information directly into the original pyramid matching framework by treating image coordinates as two extra dimensions in the feature space.



Figure 6.1 Toy example of constructing a three-level pyramid. The image has three feature types, indicated by circles, diamonds, and crosses. At the top, we subdivide the image at three different levels of resolution. Next, for each level of resolution and each channel, we count the features that fall in each spatial bin. Finally, we weight each spatial histogram according to eq. (6.3).

dense feature representation (see Section 6.3), and thus do not need to worry about spurious feature detections resulting from clutter, this practice is sufficient to deal with the effects of variable image size.

6.3 Feature Extraction

This section briefly describes the two kinds of features used in the experiments of Section 6.4. First, we have so-called "weak features," which are oriented edge points, i.e., points whose gradient magnitude in a given direction exceeds a minimum threshold. We extract edge points at two scales and eight orientations, for a total of M = 16 channels. We

designed these features to obtain a representation similar to the "gist" [150] or to a global SIFT descriptor [89] of the image.

For better discriminative power, we also utilize higher-dimensional "strong features" based on local image patches. In all the experiments presented in the previous chapters of this dissertation, we derived our image representations from the output of salient feature detectors. However, in this chapter, instead of using Harris or Laplacian interest points we simply sample fixed-size patches on a regular grid. Intuitively, we need features on a dense grid in order to capture uniform regions such as sky, calm water, or road surface — regions that do not generate any interest point detections, but are very important for forming complete descriptions of natural scenes. Indeed, Fei-Fei and Perona [30] have shown that regular grids work better for scene classification then interest points. Also note that our target application of scene recognition does not require local features to be geometrically invariant, since imaging conditions for "landscape" and "indoor" photographs are usually fairly stable (i.e., the camera is assumed to be distant from the objects in the scene, and its orientation is usually upright), whereas the sources of within-class variability for natural scene categories are so diverse and complex that they cannot be compensated by invariance to simple local 2D image transformations. In the experiments presented in Section 6.4, our grid spacing is 8 pixels and patch size is 16×16 . We represent the appearance of the patches using SIFT descriptors, and perform k-means clustering of a random subset of patches from the training set to form a visual vocabulary. Typical vocabulary sizes for our experiments are M = 200and M = 400.

6.4 Experiments

In this section, we report results on three diverse datasets: fifteen scene categories [30], Caltech101 [29], and Graz [118]. Recall that the last two datasets were used in the evaluation of Section 3.2; in this section, we will be able to compare the results of our spatial pyramid method to the results obtained by the bag-of-features method of Chapter 3. As is the case for all the experiments presented in this dissertation, we perform all processing in grayscale, even when color images are available. All experiments are repeated ten times with different randomly selected training and test images, and the average of per-class recognition rates² is recorded for each run. The final result is reported as the mean and standard deviation of the results from the individual runs. Multi-class classification is done with a support vector machine (SVM) trained using the one-versus-all rule: a classifier is learned to separate each class from the rest, and a test image is assigned the label of the classifier with the highest response.

6.4.1 Scene Category Recognition

Our first dataset (Figure 6.2) is composed of fifteen scene categories: thirteen were provided by Fei-Fei and Perona [30] (eight of these were originally collected by Oliva and Torralba [117]), and two (industrial and store) were collected by us. Each category has 200 to 400 images, and average image size is 300×250 pixels. The major sources of the pictures in the dataset include the COREL collection, personal photographs, and Google image search. This is probably the most complete scene category dataset used in the literature thus far, though the low resolution of its images is somewhat of a disadvantage.

Table 6.1 shows detailed results of classification experiments using 100 images per class for training and the rest for testing (the same setup as [30]). First, let us examine the performance of strong features for L = 0 and M = 200, corresponding to a standard bag of features. Our classification rate is 72.2% (74.7% for the 13 classes inherited from Fei-Fei and Perona), which is much higher than their best results of 65.2%, achieved with an orderless method and a feature set comparable to ours. We conjecture that Fei-Fei and Perona's approach is disadvantaged by its reliance on latent Dirichlet allocation (LDA) [9],

²The alternative performance measure, the percentage of all test images classified correctly, can be biased if test set sizes for different classes vary significantly. This is especially true of the Caltech101 dataset, where some of the "easiest" classes are disproportionately large.



Figure 6.2 Example images from the scene category database. The starred categories originate from Oliva and Torralba [117], the industrial and store categories were collected by us, and the remaining ones are from Fei-Fei and Perona [30].

	Weak features $(M = 16)$		Strong features $(M = 200)$		Strong features $(M = 400)$	
L	Single-level	Pyramid	Single-level	Pyramid	Single-level	Pyramid
$0 (1 \times 1)$	45.3 ± 0.5		72.2 ± 0.6		74.8 ± 0.3	
$1 (2 \times 2)$	53.6 ± 0.3	56.2 ± 0.6	77.9 ± 0.6	79.0 ± 0.5	78.8 ± 0.4	80.1 ± 0.5
$2(4 \times 4)$	61.7 ± 0.6	64.7 ± 0.7	79.4 ± 0.3	81.1 ± 0.3	79.7 ± 0.5	81.4 ± 0.5
$3 (8 \times 8)$	63.3 ± 0.8	$\textbf{66.8} \pm 0.6$	77.2 ± 0.4	80.7 ± 0.3	77.2 ± 0.5	81.1 ± 0.6

Table 6.1 Classification results for the scene category database (see text). The highest results for each kind of feature are shown in bold.

which is essentially an unsupervised dimensionality reduction technique and as such, is not necessarily conducive to achieving the highest classification accuracy. To verify this, we have experimented with probabilistic latent semantic analysis (pLSA) [52], which attempts to explain the distribution of features in the image as a mixture of a few "scene topics" or "aspects" and performs very similarly to LDA in practice [143]. Following the scheme of Quelhas et al. [122], we run pLSA in an unsupervised setting to learn a 60-aspect model of half the training images. Next, we apply this model to the other half to obtain probabilities of topics given each image (thus reducing the dimensionality of the feature space from 200 to 60). Finally, we train the SVM on these reduced features and use them to classify the test set. In this setup, our average classification rate drops to 63.3% from the original 72.2%. For the 13 classes inherited from Fei-Fei and Perona, it drops to 65.9% from 74.7%, which is now very similar to their results. Thus, we can see that latent factor analysis techniques can adversely affect classification performance, which is also consistent with the results of Quelhas et al. [122].

Next, let us examine the behavior of spatial pyramid matching. For completeness, Table 6.1 lists the performance achieved using just the highest level of the pyramid (the "single-level" columns), as well as the performance of the complete matching scheme using multiple levels (the "pyramid" columns). For all three kinds of features, results improve dramatically as we go from L = 0 to a multi-level setup. Though matching at the highest pyramid level seems to account for most of the improvement, using all the levels together confers a statistically significant benefit. For strong features, single-level performance actually drops as we go from L = 2 to L = 3. This means that the highest level of the L = 3 pyramid is too finely subdivided, with individual bins yielding too few matches. Despite the diminished discriminative power of the highest level, the performance of the entire L = 3 pyramid remains essentially identical to that of the L = 2 pyramid. This, then, is the main advantage

of the spatial pyramid representation: because it combines multiple resolutions in a principled fashion, it is robust to failures at individual levels.

It is also interesting to compare performance of different feature sets. As expected, weak features do not perform as well as strong features, though in combination with the spatial pyramid, they can also achieve acceptable levels of accuracy (note that because weak features have a much higher density and much smaller spatial extent than strong features, their performance continues to improve as we go from L = 2 to L = 3). Increasing the visual vocabulary size from M = 200 to M = 400 results in a small performance increase at L = 0, but this difference is all but eliminated at higher pyramid levels. Thus, we can conclude that the coarse-grained geometric cues provided by the pyramid have more discriminative power than an enlarged visual vocabulary. Of course, the optimal way to exploit structure both in the image and in the feature space may be to combine them in a unified multiresolution framework; this is subject for future research.

Figure 6.3 shows the confusion table between the fifteen scene categories. Not surprisingly, confusion occurs between the indoor classes (kitchen, bedroom, living room), and also between some natural classes, such as coast and open country. Figure 6.4 shows examples of image retrieval using the spatial pyramid kernel and strong features with M = 200. These examples give a qualitative sense of the kind of visual information captured by our approach. In particular, spatial pyramids seem successful at capturing the organization of major pictorial elements or "blobs," the directionality of dominant edges, and the perspective (amount of foreshortening, location of vanishing points). Because the spatial pyramid is based on features computed at the original image resolution, even high-frequency details can be preserved. For example, query image (b) shows white kitchen cabinet doors with dark borders. Three of the retrieved "kitchen" images contain cabinets of a similar design, the "office" image shows a wall plastered with white documents in dark frames, and the "inside city" image shows a white building with darker window frames.



Figure 6.3 Confusion table for the scene category dataset. Average classification rates for individual classes are listed along the diagonal. The entry in the *i*th row and *j*th column is the percentage of images from class *i* that were misidentified as class *j*.



Figure 6.4 Retrieval from the scene category database. The query images are on the far left, and the eight images giving the highest values of the spatial pyramid kernel (for L = 2, M = 200) are shown in decreasing order of kernel value from left to right. The actual classes of incorrectly retrieved images are listed below them.

6.4.2 Caltech101

Our second set of experiments is on the Caltech101 database [29], which was introduced in Section 3.2.3 (Figure 3.13). As in Section 3.2.3, we follow the experimental setup of Grauman and Darrell [45], namely, we train on 30 images per class and test on the rest. For efficiency, we limit the number of test images to 50 per class. Note that, because some categories are very small, we may end up with just a single test image per class. Table 6.2 gives a breakdown of classification rates for different pyramid levels for weak features and strong features with M = 200. The results for M = 400 are not shown, because just as for the scene category database, they do not bring any significant improvement. For L = 0, strong features give 41.2%, which is slightly below the 43% reported by Grauman and Darrell. Our best result is 64.6%, achieved with strong features at L = 2. This exceeds the highest classification rate known to us so far, that of 53.9% achieved by our bag-of-features method in Section 3.2.3 (Table 3.13)³ Berg et al. [5] report 48% accuracy using 15 training images per class. Our average recognition rate with this setup is 56.4%. The behavior of weak features on this database is also noteworthy: for L = 0, they give a classification rate of 15.5%, which is consistent with a naive graylevel correlation baseline [5], but in conjunction with a four-level spatial pyramid, their performance rises to 54% — on par with the best results achieved with much more discriminative local features by an orderless method.

Figure 6.5 shows a few of the "easiest" and "hardest" object classes for our method. The successful classes are either dominated by rotation artifacts (like minaret), have very little clutter (like windsor chair), or represent coherent natural "scenes" (like joshua tree and okapi). The least successful classes are either textureless animals (like beaver and cougar), animals that camouflage well in their environment (like crocodile), or "thin" objects (like ant). Table 6.3 further illuminates the performance of our method by showing the top five of its confusions, all of which are between closely related classes.

³This table lists the percentage of all test images classified correctly, which overestimates the per-class average, as discussed earlier.
	Weak features		Strong features (200)	
L	Single-level	Pyramid	Single-level	Pyramid
0	15.5 ± 0.9		41.2 ± 1.2	
1	31.4 ± 1.2	32.8 ± 1.3	55.9 ± 0.9	57.0 ± 0.8
2	47.2 ± 1.1	49.3 ± 1.4	63.6 ± 0.9	64.6 ± 0.8
3	52.2 ± 0.8	54.0 ± 1.1	60.3 ± 0.9	64.6 ± 0.7

Table 6.2 Classification results for the Caltech101 database.



Figure 6.5 Caltech101 results. Top: some classes on which our method (L = 2, M = 200) achieved high performance. Bottom: some classes on which our method performed poorly.

alags 1 / alags 2	class 1 misclassified	class 2 misclassified	
Class 1 / Class 2	as class 2	as class 1	
ketch / schooner	21.6	14.8	
lotus / water lily	15.3	20.0	
crocodile / crocodile head	10.5	10.0	
crayfish / lobster	11.3	9.1	
flamingo / ibis	9.5	10.4	

Table 6.3 Top five confusions for our method (L = 2, M = 200) on the Caltech101 database.

To summarize, our method has outperformed both state-of-the-art orderless methods [45, 168] and methods based on precise geometric correspondence [5]. Significantly, all these methods rely on sparse features (interest points or sparsely sampled edge points). However, because of the geometric stability and lack of clutter of Caltech101, dense features combined with global spatial relations seem to capture more discriminative information about the objects.

6.4.3 The Graz Dataset

As seen from Sections 6.4.1 and 6.4.2, our proposed approach does very well on global scene classification tasks, or on object recognition tasks in the absence of clutter with most of the objects assuming "canonical" poses. However, it was not designed to cope with heavy clutter and pose changes. It is interesting to see how well our algorithm can do by exploiting the global scene cues that still remain under these conditions. Accordingly, our final set of experiments is on the Graz dataset [118], which was introduced in Section 3.2.3 (Figure 3.15). For this dataset, the range of scales and poses at which exemplars are presented is very diverse, e.g., a "person" image may show a pedestrian in the distance, a side view of a complete body, or just a closeup of a head. For this database, we perform two-class detection (object vs. background) using an experimental setup consistent with that of Opelt et al. [118]. Namely, we train detectors for persons and bikes on 100 positive and 100 negative images (of which 50 are drawn from the other object class and 50 from the background), and test on a similarly distributed set. We generate ROC curves by thresholding raw SVM output, and report the ROC equal error rate averaged over ten runs.

Table 6.4 summarizes our results for strong features with M = 200. Note that the standard deviation is quite high because the images in the database vary greatly in their level of difficulty, so the performance for any single run is dependent on the composition of the training set (in particular, for L = 2, the performance for bikes ranges from 81% to 91%).

Class	L = 0	L = 2	Opelt [118]	Zhang
Bikes	82.4 ± 2.0	86.3 ± 2.5	86.5	92.0
People	79.5 ± 2.3	82.3 ± 3.1	80.8	88.0

Table 6.4 Results of our method (M = 200) for the Graz database and comparison with two bag-of-features methods (refer back to Table 3.14).

For this database, the improvement from L = 0 to L = 2 is relatively small. This makes intuitive sense: when a class is characterized by high geometric variability, it is difficult to find useful global features. Despite this disadvantage of our method, we still achieve results very close to those of Opelt et al. [118]. This is somewhat surprising, considering that their method uses a sparse, locally invariant feature representation, which is expected (at least in principle) to make it more robust to clutter and deformations. Note, however, that for this dataset, we cannot outperform our kernel-based orderless method of Section 3.2.3.

6.5 Discussion

This chapter has presented a "holistic" approach for image categorization based on a modification of pyramid match kernels [45]. Our method, which works by repeatedly subdividing an image and computing histograms of image features over the resulting subregions, has shown promising results on three large-scale, diverse datasets. Note that in the experiments of this section, we validate the performance of the pyramid method by using several baseline tests: one baseline is obtained by omitting subdivision, resulting in a standard bag of features, and another one by using weak features instead of strong ones. Despite the simplicity of our method, and despite the fact that it works not by constructing explicit object models, but by using global cues as indirect evidence about the presence of an object, it consistently achieves an improvement over an orderless image representation. This is not a trivial accomplishment, since we have established in Chapter 3 that a well-designed bag-of-features method can outperform more sophisticated approaches based on parts and geometric relations. Our results also underscore the surprising and ubiquitous power of global scene statistics: even in highly variable datasets, such as Graz, they can still provide useful discriminative information. It is important to develop methods that take full advantage of this information — either as stand-alone scene categorizers, as "context" modules within larger object recognition systems, or as tools for evaluating biases present in newly collected datasets.

CHAPTER 7

Conclusion

In the final chapter of this dissertation, we briefly recapitulate the main contributions of our research and discuss possible directions for future work.

7.1 Summary

In this dissertation, we have considered a variety of image representations for recognizing textures, objects, and scenes. Our research has progressed from purely *local* models, which lack any geometric constraints save the coherence of the intensity pattern within each individual image patch, to *semi-local* models, which incorporate strong spatial constraints between relatively small groups of patches located close to one another, and finally, to *global* models, which record the absolute position of each feature within an image. In the following, we summarize the image representations considered in our work and highlight the most important experimental findings.

Bag-of-features models. Chapter 3 has presented an orderless *bag-of-features* method for recognizing images of textured surfaces subject to large scale changes, rotations, perspective distortions, and non-rigid deformations. The use of local invariant features in this method has enabled us to handle a wider range of geometric variability than other existing texture analysis methods. This chapter has also presented extensive experiments on several texture and object datasets that have convincingly demonstrated the suitability of orderless methods for classification of object images, despite substantial noise in local feature distributions introduced by clutter and uncorrelated backgrounds. Because this method is conceptually primitive, yet surprisingly effective on real-world data, we view it as a good baseline against which to evaluate alternative methods that incorporate more sophisticated spatial constraints.

Neighorhood co-occurrence relations. Chapter 4 has considered methods for extending the purely local bag-of-features model with statistical co-occurrence relations between pairs of nearby features. The resulting image representation is a conceptual analogue of a bigram model for natural language. In the experiments of this chapter, relations did not improve the accuracy of whole-image classification, but they did prove useful for the more challenging task of labeling individual features in multi-texture images.

Semi-local parts. Chapter 5 has introduced a structured object representation based on multiple composite geometrically stable parts. This method has more descriptive power for modeling objects than orderless bags of features, and it achieves higher classification rates on our bird and butterfly databases. Moreover, because our approach relies on explicit geometric alignment search, it is capable of higher degrees of invariance than probabilistic constellation models, and can find larger object models in substantial clutter.

Spatial pyramid matching. Chapter 6 has presented a *global* method for recognizing scene categories that works by establishing approximate geometric correspondence between sets of local features found in two images. This method is simple and efficient, and shows much higher performance than the bag-of-features baseline on one scene and two object datasets. Even though this approach is not geometrically invariant, it makes effective use of multiple training exemplars to learn intra-class appearance variations.

Overall, each of the models considered in this dissertation has demonstrated its usefulness on a particular class of recognition problems. At this stage, however, we still have a fairly incomplete understanding of models that combine local features or parts with co-occurrence relations. As shown by our experiments with the discriminative maximum entropy framework (Section 4.1.3 and Section 5.3.2), co-occurrence relations for textons and overlap relations for parts did not improve the performance of whole-image classification of textures and objects, respectively. However, co-occurrence relations did prove useful for labeling individual regions in multi-texture images (Section 4.2.2), and we conjecture that inter-part relations will likewise prove useful for object localization. Intuitively, the more demanding the recognition task (i.e., the more detailed the explanation of the image that must be supplied by the recognition algorithm), the more elaborate the image representation should be. In most experiments, we have considered exclusively whole-image classification, which is the simplest task, since it requires a single label for the entire image. In the future, we plan to devote more attention to more complex image interpretation tasks, which should yield additional insights into the role of spatial relations in image models.

7.2 Extensions

In the future, we will work towards improving the flexibility, discriminative power, and expressiveness of our image models to make them applicable to a wider range of challenging imagery, including objects that lack characteristic texture, and those that are highly nonrigid and articulated (Figure 7.1).



Figure 7.1 Images that challenge the capabilities of our current representation methods: Objects that lack texture, such as the plaster head and the glass vase, do not lend themselves to a patch-based representation. The spots and colors of the cows are not characteristic of their class, and neither is the clothing that people wear. Human bodies are highly articulated and their limb structure may not be clearly visible in a photograph.

Feature representation. The salient region detectors and appearance-based descriptors used in this dissertation are best at capturing stable, distinctive texture patterns. For this reason, we currently cannot handle objects whose texture and appearance is not characteristic of their class (Figure 7.1). To overcome this difficulty, we need to develop detectors and descriptors that are more suited to the description of shape, as opposed to texture. In the future, we also plan to develop and evaluate low-level features that are more appropriate for describing the shape of non-rigid and articulated objects. In addition, we will work on overcoming our current reliance on ad-hoc combinations of hand-designed appearance detectors and descriptors by replacing them with discriminative feature extractors that can be automatically learned for a given recognition task [77]. Finally, we would like to develop representations that support the sharing of features and/or parts between structurally similar categories [149].

Dynamical parts-and-shape models. The development of better parts-and-shape object models, as well as their quantitative evaluation for localization tasks are important future directions. We plan to develop more sophisticated methods for reasoning about the relationship between different semi-local parts, which will enable us to handle complex classes of articulated objects such as human beings [32, 55], animal species [123], and more general non-rigid shapes. An effective representation of articulated objects would have to involve a full dynamical model describing how different parts move together. We would like to develop algorithms for learning dynamical models from video sequences. By taking advantage of the large volume of training data and the strong temporal continuity cues contained in video, it is possible to conduct semi-supervised learning of relational models consisting of many parameters and describing characteristic motion patterns for human or animal classes. The dynamical models learned in this fashion can then be used for tracking, and also for localizing object instances in still images. In fact, promising recent work [123, 124] has shown that tracking can be carried out successfully by (essentially) repeated localization of the object

in every frame. We plan to exploit this insight to extend to video the techniques developed for object recognition in still images. Finally, we would like to harness the expressive power of dynamical representations to build models whose part configuration and appearance can change dynamically depending on the aspect of the object being observed, thereby provide a multiview representation capable of capturing true 3D structure.

APPENDIX A

Harris and Laplacian Region Detectors

This appendix presents the details of the scale- and affine-invariant¹ Harris [105] and Laplacian [40, 82] region detectors used in Chapters 3-5 of this dissertation. First, scaleinvariant regions are obtained by performing automatic spatial and scale selection (Section A.1). Next, the affine shape of these regions is estimated through an *affine adaptation* process (Section A.2).

A.1 Spatial and Scale Selection

This section discusses the procedure for extracting *scale-adapted* local regions, i.e., interest points equipped with characteristic scales. The Laplacian detector extracts image regions whose locations and characteristic scales are given by scale-space maxima of the Laplace operator. The Harris detector uses the same operator for scale selection, but finds the locations of interest points as the local maxima of a "cornerness" measure based on the second moment matrix.

Let us begin by summarizing Lindeberg's procedure for spatial and scale selection using the Laplace operator [82]. Let I(x) denote the image intensity as a function of position. To simplify the presentation, we will treat I(x) as a continuous differentiable function. We can form a basic linear scale space by considering all possible images that result from convolving

¹Recall from Chapter 2 that the technically correct term is not *invariant* but *covariant*, but we use the former term throughout this dissertation to maintain consistency with existing literature.

I(x) with an isotropic Gaussian of standard deviation σ :

$$I(x;\sigma) = G(x;\sigma) * I(x) = \int G(x - x';\sigma) I(x') dx',$$

$$G(x;\sigma) = \frac{1}{2\pi\sigma^2} \exp\left(-\frac{|x|^2}{2\sigma^2}\right).$$

For any fixed scale σ , we can compute any combination of spatial derivatives of $I(x; \sigma)$ (effectively, these derivatives are computed by convolving the image I(x) with the appropriate combination of derivatives of the Gaussian $G(x; \sigma)$). For instance, we may want to customdesign a differential operator tuned to certain kinds of image structures, such as edges, ridges, or blobs [82]. A desirable property for a scale-space differential operator is that it should always produce the same response to an idealized scale-invariant structure like a step edge. However, if we simply take a "blurred derivative" of a step edge, we will obtain weaker responses at larger scales. This motivates the definition of *scale-normalized* differential operators, whose output remains constant if the image is scaled (resized) by an arbitrary factor. One particularly useful normalized differential quantity is the scalenormalized Laplacian $\hat{\Delta}I(x; \sigma)$, which is one of the simplest scale selection operators [82]:

$$\hat{\Delta}I(x;\sigma) = \sigma^2 \left(\frac{\partial^2 I(x;\sigma)}{\partial x^2} + \frac{\partial^2 I(x;\sigma)}{\partial y^2}\right)$$

$$= \sigma^2 \left(\frac{\partial^2 G(x;\sigma)}{\partial x^2} + \frac{\partial^2 G(x;\sigma)}{\partial y^2}\right) * I(x) = \hat{\Delta}G(x;\sigma) * I(x).$$
(A.1)

As an illustration of why the scale-normalized Laplacian is appropriate for selecting the characteristic scale of local image structures, consider the one-dimensional toy example of Figure A.1. The left-most picture shows a one-dimensional binary image (signal):

$$I(x) = \begin{cases} 1, & -8 \le x \le 8\\ 0, & \text{otherwise.} \end{cases}$$

The next six pictures show the signal I(x) convolved with one-dimensional scale-normalized Laplacian kernels $\hat{\Delta}G(x;\sigma)$ for different values of σ . The family of convolved signals $\hat{\Delta}G(x;\sigma)$ * I(x) has a clear global minimum at $\sigma = 8$ and x = 0. Thus, the scale-normalized Laplacian shows the strongest response at the characteristic scale of the signal.



Figure A.1 Scale selection: one-dimensional example (see text). The dotted vertical line is x = 0.

Scale-selection operators like the Laplacian can also be used to find interest points in an image. We define a Laplacian *scale-space interest point* as a point where $\hat{\Delta}I(x;\sigma)$ *simultane-ously* achieves a local maximum with respect to the scale parameter and the spatial variables. In the example of Figure A.1, $(x, \sigma) = (0, 8)$ is a scale-space interest point. By computing such interest points, we treat scale and spatial selection as parts of the same problem.

The Harris detector is obtained by a simple modification of the Laplacian detector, where for spatial selection instead of the Laplacian we use the "cornerness" measure based on the *second moment matrix*, also known as the *local shape matrix* or the *auto-correlation matrix* [40, 105]:

$$M_I(x;\gamma,\sigma) = G(x;\sigma) * \left(\nabla I(x;\gamma) \ \nabla I(x;\gamma)^T\right) . \tag{A.2}$$

Note that $M_I(x; \gamma, \sigma)$ simultaneously lives in two scale spaces with parameters γ and σ . The *inner scale* γ is the scale at which smoothed image derivatives are computed, while the *outer scale* σ is the scale of integration — it defines the size of the window over which second moment information is accumulated. Mikolajczyk and Schmid [105] point out that the inner scale γ is less critical than the outer characteristic scale σ (though they do implement a procedure for selecting γ automatically). Intuitively, the inner scale describes the intrinsic resolution of the image measurement process — any information that is lost after blurring the image with γ is lost forever. In our implementation, we set γ to a constant value of 2.

Informally, the eigenvectors of the second moment matrix represent the dominant (orthogonal) edge orientations of the window, and the eigenvalues represent the amount of edge energy along these directions. The Harris detector looks for points like corners and junctions that have significant amounts of energy in two orthogonal directions. The Harris "cornerness" measure is defined as follows:

$$\det \left(M_I(x;\gamma,\sigma) \right) - \alpha \operatorname{tr}^2 \left(M_I(x;\gamma,\sigma) \right) \, ,$$

where α is a constant (0.05 in the implementation). Local maxima of this measure determine the locations of Harris interest points, and then the Laplacian scale selection procedure is applied at these locations to find their characteristic scales.

A.2 Affine Adaptation

Following spatial and scale selection, each interest point is described by a spatial location x and a characteristic scale σ . However, one could argue that a single characteristic scale parameter does not adequately describe the local geometry of a texture patch. After all, many textures are highly anisotropic, having different orientations and varying scales along different directions. To obtain a more accurate estimate of local shape for some point with location x and scale σ , we compute the second moment matrix $M_I(x; \gamma, \sigma)$, as defined by eq. (A.2). Consider the equation

$$(x - x_0)^T M(x - x_0) = 1,$$

where $M = M_I(x_0; \gamma, \sigma)$. It describes an ellipse centered at x_0 , with principal axes corresponding to eigenvectors of M_I and axis lengths equal to the square roots of the inverse of the eigenvalues. This ellipse captures the local geometry of the corresponding texture patch, provided σ matches its characteristic scale. In the implementation, we scale the axes of each ellipse so that its area is proportional to the area of the circle with radius σ . Suppose that we have computed an estimate M of the local shape matrix at some image location x_0 . Without loss of generality, let us assume that x_0 is the origin. Given M, we can *normalize* the image patch centered at x_0 by computing a coordinate transformation U that would send the ellipse corresponding to M to a unit circle:

$$\hat{x} = Ux = M^{1/2}x$$

where $M^{1/2}$ is any matrix such that $M = (M^{1/2})^T (M^{1/2})$. Since M is a symmetric positivedefinite matrix, it can be diagonalized as $M = Q^T D Q$ where Q is orthogonal and D is diagonal (with positive entries), so we can set $M^{1/2} = D^{1/2}Q$.

Second-moment matrices can also be defined in *affine Gaussian scale space* where local and integration scales are no longer described by single parameters γ and σ , but by 2 × 2 covariance matrices Γ and Σ [105]:

$$M_{I}(x;\Gamma,\Sigma) = G(x;\Sigma) * \left(\nabla I(x;\Gamma) \ \nabla I(x;\Gamma)^{T}\right), \quad \text{where}$$
$$G(x;\Sigma) = \frac{1}{2\pi\sqrt{\det\Sigma}} \exp\left(-\frac{x^{T}\Sigma^{-1}x}{2}\right),$$
$$\nabla I(x;\Gamma) = \nabla G(x;\Gamma) * I(x).$$

Consider two texture patches I and I' centered at locations x_0 and x'_0 (we can take x_0 and x'_0 as the origins of the respective local coordinate systems). Suppose that these patches are related by an affine transformation x' = Ax:

$$I(x) = I'(x') \; .$$

Then we can show [40, 105] that their second moment matrices are related as

$$M_{I}(x_{0}; \Gamma, \Sigma) = A^{T} M_{I'}(x_{0}'; A \Gamma A^{T}, A \Sigma A^{T}) A \quad \text{or}$$

$$M = A^{T} M' A .$$
(A.3)

We can visualize the above transformation by thinking of the two ellipses $x^T M x = 1$ and $x'^T M' x' = 1$ related by the coordinate change x' = Ax. If we have computed M and M', we can normalize the local neighborhoods of I and I' as in (A.2):

$$\hat{x} = M^{1/2}x, \qquad \hat{x}' = M'^{1/2}x'.$$

In a realistic situation, A, the coordinate transformation between the image regions I and I', will be unknown. In fact, the two regions may not be related by an affine transformation at all. For this reason, (A.3) is really an equation where $M^{1/2}$ and $M'^{1/2}$ are known, and A must be determined. As it turns out, the solution is not unique [40]:

$$A = M'^{-1/2} R M^{1/2},$$

where R is an arbitrary orthogonal matrix. We can rewrite the above equation as

$$M'^{1/2}Ax = RM^{1/2}x$$
$$\hat{x}' = R\hat{x} .$$

We have arrived at the following conclusion: if the local coordinate systems of I and I' are related by an affine transformation, then the respective normalized systems are related by an arbitrary orthogonal transformation (a combination of rotation and reflection). Thus, we cannot obtain a complete registration between two image patches by using the local shape matrices alone, even if these can be computed exactly. As explained elsewhere in this dissertation, we can resolve the remaining ambiguity either by computing rotation-invariant descriptors of the patches, or by aligning their dominant gradient orientations.

Finally, note that the geometry estimate provided by the local shape matrix can be reasonably accurate only when the neighborhood over which the local shape matrix is computed matches the true shape of the texture patch. This is a bootstrapping problem that can be solved by an iterative approach [83, 105]. In our own implementation, the inclusion of an iterative scheme has not resulted in any measurable change in performance, and therefore we use just a single step of affine adaptation for all the experiments presented in this dissertation.

REFERENCES

- S. Agarwal and D. Roth. Learning a sparse representation for object detection. In Proceedings of the European Conference on Computer Vision, volume 4, pages 113– 130, 2002.
- [2] D.H. Ballard and C.M. Brown. Computer Vision. Prentice-Hall, Englewood Cliffs, NJ, 1982.
- [3] A. Baumberg. Reliable feature matching across widely separated views. In Proceedings of the IEEE Conference on Computer Vision and Pattern Recognition, volume 1, pages 774–781, 2000.
- [4] S. Belongie, J. Malik, and J. Puzicha. Shape matching and object recognition using shape contexts. *IEEE Transactions on Pattern Analysis and Machine Intelligence*, 24(4):509–522, 2002.
- [5] A. Berg, T. Berg, and J. Malik. Shape matching and object recognition using low distortion correspondences. In *Proceedings of the IEEE Conference on Computer Vision* and Pattern Recognition, volume 1, pages 26–33, 2005.
- [6] A. Berger, S. Della Pietra, and V. Della Pietra. A maximum entropy approach to natural language processing. *Computational Linguistics*, 22(1):39–71, 1996.
- [7] I. Biederman. Aspects and extension of a theory of human image understanding. In Z. Pylyshyn, editor, *Computational Processes in Human Vision: An Interdisciplinary Perspective*. Ablex Publishing Corporation, Norwood, New Jersey, 1988.
- [8] C. Bishop. Neural Networks for Pattern Recognition. Oxford University Press, 1995.
- [9] D. Blei, A. Ng, and M. Jordan. Latent Dirichlet allocation. *Journal of Machine Learning Research*, 3:993–1022, 2003.
- [10] D. Blostein and N. Ahuja. A multiscale region detector. Computer Vision, Graphics and Image Processing, 45:22–41, 1989.
- [11] B. Bradshaw, B. Schölkopf, and J. Platt. Kernel methods for extracting local image semantics. Technical Report MSR-TR-2001-99, Microsoft Research, 2001.
- [12] P. Brodatz. Textures: A Photographic Album for Artists and Designers. Dover, New York, 1966.
- [13] R. Brooks, R. Greiner, and T. Binford. The acronym model-based vision system. In Proc. International Joint Conference on Artificial Intelligence, pages 105–113, 1979.

- [14] M. Burl, M. Weber, and P. Perona. A probabilistic approach to object recognition using local photometry and global geometry. In *Proceedings of the European Conference on Computer Vision*, pages 628–641, 1998. Lecture Notes in Computer Science 1407.
- [15] B. Caputo, C. Wallraven, and M.-E. Nilsback. Object categorization via local kernels. In Proceedings of the International Conference on Pattern Recognition, volume 2, pages 132–135, 2004.
- [16] O. Chapelle, P. Haffner, and V. Vapnik. Support vector machines for histogram-based image classification. *IEEE Transactions on Neural Networks*, 10(5):1055–1064, 1999.
- [17] S. Chen and J. Goodman. An empirical study of smoothing techniques for language modeling. In Proc. Conf. of the Association for Computational Linguistics, pages 310– 318, 1996.
- [18] F.S. Cohen, Z. Fan, and M.A.S. Patel. Classification of rotated and scaled textured images using Gaussian Markov field models. *IEEE Transactions on Pattern Analysis* and Machine Intelligence, 13(2):192–202, 1991.
- [19] J.L Crowley and A.C. Parker. A representation of shape based on peaks and ridges in the difference of low-pass transform. *IEEE Transactions on Pattern Analysis and Machine Intelligence*, 6:156–170, 1984.
- [20] G. Csurka, C. Dance, L. Fan, J. Willamowski, and C. Bray. Visual categorization with bags of keypoints. In ECCV Workshop on Statistical Learning in Computer Vision, 2004.
- [21] O.G. Cula and K.J. Dana. Compact representation of bidirectional texture functions. In Proceedings of the IEEE Conference on Computer Vision and Pattern Recognition, volume 1, pages 1041–1047, 2001.
- [22] K.J. Dana, B. van Ginneken, S.K. Nayar, and J.J. Koenderink. Reflectance and texture of real world surfaces. ACM Transactions on Graphics, 18(1):1–34, 1999.
- [23] G. Dorko and C. Schmid. Selection of scale-invariant parts for object class recognition. In Proceedings of the International Conference on Computer Vision, volume 1, pages 634–640, 2003.
- [24] G. Dorko and C. Schmid. Object class recognition using discriminative local features. Technical Report RR-5497, INRIA, 2005.
- [25] J. Eichhorn and O. Chapelle. Object categorization with SVM: kernels for local features. Technical report, Max Planck Institute for Biological Cybernetics, Tuebingen, Germany, 2004.
- [26] M. Everingham, A. Zisserman, C. Williams, L. Van Gool, M. Allan, C. Bishop, O. Chapelle, N. Dalal, T. Deselaers, G. Dorko, S. Duffner, J. Eichhorn, J. Farquhar, M. Fritz, C. Garcia, T. Griffiths, F. Jurie, D. Keysers, M. Koskela, J. Laaksonen, D. Larlus, B. Leibe, H. Meng, H. Ney, B. Schiele, C. Schmid, E. Seemann, J. Shawe-Taylor, A. Storkey, S. Szedmak, B. Triggs, I. Ulusoy, V. Viitaniemi, and J. Zhang. The 2005 PASCAL visual object classes challenge. In F. d'Alche Buc, I. Dagan, and

J. Quinonero, editors, *Selected Proceedings of the first PASCAL Challenges Workshop*. LNAI, Springer, 2006. http://www.pascal-network.org/challenges/VOC/.

- [27] O. Faugeras and M. Hebert. A 3-d recognition and positioning algorithm using geometric matching between primitive surfaces. In Proc. International Joint Conference on Artificial Intelligence, pages 996–1002, 1983.
- [28] L. Fei-Fei, R. Fergus, and P. Perona. A Bayesian approach to unsupervised oneshot learning of object categories. In *Proceedings of the International Conference on Computer Vision*, 2003.
- [29] L. Fei-Fei, R. Fergus, and P. Perona. Learning generative visual models from few training examples: an incremental Bayesian approach tested on 101 object categories. In *IEEE CVPR Workshop on Generative-Model Based Vision*, 2004. http://www.vision.caltech.edu/Image_Datasets/Caltech101.
- [30] L. Fei-Fei and P. Perona. A Bayesian hierarchical model for learning natural scene categories. In *Proceedings of the IEEE Conference on Computer Vision and Pattern Recognition*, 2005.
- [31] L. Fei-Fei, R. VanRullen, C. Koch, and P. Perona. Natural scene categorization in the near absense of attention. *Proceedings of the National Academy of Sciences USA*, 99(14):9596–9601, 2002.
- [32] P. Felzenszwalb and D. Huttenlocher. Pictorial structures for object recognition. International Journal of Computer Vision, 61(1):55–79, 2005.
- [33] R. Fergus, P. Perona, and A. Zisserman. Object class recognition by unsupervised scale-invariant learning. In *Proceedings of the IEEE Conference on Computer Vision* and Pattern Recognition, volume 2, pages 264–271, 2003.
- [34] V. Ferrari, T. Tuytelaars, and L. Van Gool. Simultaneous object recognition and segmentation by image exploration. In *Proceedings of the European Conference on Computer Vision*, 2004.
- [35] M. Fischler and R. Elschlager. The representation and matching of pictorial structures. *IEEE Transactions on Computers*, 22(1):67–92, 1973.
- [36] A. Fitzgibbon and A. Zisserman. On affine invariant clustering and automatic cast listing in movies. In *Proceedings of the European Conference on Computer Vision*, volume 3, pages 304–320, 2002.
- [37] W. Freeman and E. Adelson. The design and use of steerable filters. *IEEE Transactions* on Pattern Analysis and Machine Intelligence, 13(9):891–906, 1991.
- [38] B. Frey. *Graphical Models for Machine Learning and Digital Communication*. MIT Press, Cambridge, MA, 1998.
- [39] D. Gabor. Theory of communication. J. Inst. Electr. Eng., 93:429–457, 1946.
- [40] J. Gårding and T. Lindeberg. Direct computation of shape cues using scale-adapted spatial derivative operators. *International Journal of Computer Vision*, 17(2):163–191, 1996.

- [41] P.C. Gaston and T. Lozano-Pérez. Tactile recognition and localization using object models: The case of polyhedra in the plane. *IEEE Transactions on Pattern Analysis* and Machine Intelligence, 6(3), 1984.
- [42] B. Georgescu, I. Shimshoni, and P. Meer. Mean shift based clustering in high dimensions: A texture classification example. In *Proceedings of the International Conference* on Computer Vision, pages 456–463, 2003.
- [43] R. Goldstone. Similarity, interactive activation, and mapping. *Journal of Experimental Psychology: Learning, Memory, and Cognition*, 20(1):3–28, 1994.
- [44] M. Gorkani and R. Picard. Texture orientation for sorting photos "at a glance". In IAPR International Conference on Pattern Recognition, volume 1, pages 459–464, 1994.
- [45] K. Grauman and T. Darrell. Pyramid match kernels: Discriminative classification with sets of image features. In Proceedings of the International Conference on Computer Vision, 2005.
- [46] W. E. L. Grimson. The combinatorics of object recognition in cluttered environments using constrained search. Artificial Intelligence Journal, 44(1-2):121–166, 1990.
- [47] W.E.L. Grimson and T. Lozano-Pérez. Model-based recognition and localization from sparse range or tactile data. *International Journal of Robotics Research*, 3(3), 1984.
- [48] E. Hadjidemetriou, M. Grossberg, and S. Nayar. Multiresolution histograms and their use in recognition. *IEEE Transactions on Pattern Analysis and Machine Intelligence*, 26(7):831–847, 2004.
- [49] R. Haralick. Statistical and structural approaches to texture. Proceedings of the IEEE, 67:786–804, 1979.
- [50] C. Harris and M. Stephens. A combined corner and edge detector. In M. M. Matthews, editor, *Proceedings of the 4th Alvey Vision Conference*, pages 147–151, 1988.
- [51] E. Hayman, B. Caputo, M. Fritz, and J.-O. Eklundh. On the significance of real-world conditions for material classification. In *Proceedings of the European Conference on Computer Vision*, pages 253–266, 2004.
- [52] T. Hofmann. Unsupervised learning by probabilistic latent semantic analysis. Machine Learning, 42(1):177–196, 2001.
- [53] D. Hoiem, A. Efros, and M. Hebert. Geometric context from a single image. In *Proceedings of the International Conference on Computer Vision*, 2005.
- [54] D. P. Huttenlocher and S. Ullman. Object recognition using alignment. In *Proceedings* of the International Conference on Computer Vision, pages 102–111, 1987.
- [55] S. Ioffe and D. Forsyth. Probabilistic methods for finding people. *International Journal* of Computer Vision, 43(1):45–68, 2001.

- [56] J. Jeon and R. Manmatha. Using maximum entropy for automatic image annotation. In Proc. Conf. on Image and Video Retrieval, pages 24–32, 2004.
- [57] F. Jing, M. Li, H.-J. Zhang, and B. Zhang. Support vector machines for region-based image retrieval. In *IEEE International Conference on Multimedia and Expo*, 2003.
- [58] A. Johnson and M. Hebert. Using spin images for efficient object recognition in cluttered 3D scenes. *IEEE Transactions on Pattern Analysis and Machine Intelligence*, 21(5):433-449, 1999.
- [59] B. Julesz. Visual pattern discrimination. IRE Transactions on Information Theory, IT-8:84–92, 1962.
- [60] B. Julesz. Textons, the elements of texture perception and their interactions. *Nature*, 290:91–97, 1981.
- [61] F. Jurie and C. Schmid. Scale-invariant shape features for recognition of object categories. In *Proceedings of the IEEE Conference on Computer Vision and Pattern Recognition*, 2004.
- [62] T. Kadir and M. Brady. Scale, saliency and image description. International Journal of Computer Vision, 45(2):83–105, 2001.
- [63] L. Kaufman and P. Rousseeuw. Finding Groups in Data: An Introduction to Cluster Analysis. John Wiley & Sons, New York, 1990.
- [64] D. Keysers, F. Och, and H. Ney. Maximum entropy and gaussian models for image object recognition. In *DAGM Symposium for Pattern Recognition*, 2002.
- [65] J. Koenderink and A. Van Doorn. Representation of local geometry in the visual system. *Biological Cybernetics*, 55:367–375, 1987.
- [66] J. Koenderink and A. Van Doorn. The structure of locally orderless images. International Journal of Computer Vision, 31(2/3):159–168, 1999.
- [67] F. Kschischang, B. Frey, and H.-A. Loeliger. Factor graphs and the sum-product algorithm. *IEEE Trans. on Information Theory*, 47(2):498–519, 2001.
- [68] S. Kumar and M. Hebert. Discriminative random fields: A discriminative framework for contextual interaction in classification. In *Proceedings of the International Confer*ence on Computer Vision, volume 2, pages 1150–1157, 2003.
- [69] S. Kumar, A.C. Loui, and M. Hebert. An observation-constrained generative approach for probabilistic classification of image regions. *Image and Vision Computing*, 21:87–97, 2003.
- [70] J. Lafferty, A. McCallum, and F. Pereira. Conditional random fields: Probabilistic models for segmenting and labeling sequence data. In *Proc. Int. Conf. on Machine Learning*, pages 282–289, 2001.
- [71] S. Lazebnik, C. Schmid, and J. Ponce. Affine-invariant local descriptors and neighborhood statistics for texture recognition. In *Proceedings of the International Conference* on Computer Vision, pages 649–655, 2003.

- [72] S. Lazebnik, C. Schmid, and J. Ponce. A sparse texture representation using affineinvariant regions. In *Proceedings of the IEEE Conference on Computer Vision and Pattern Recognition*, volume 2, pages 319–324, 2003.
- [73] S. Lazebnik, C. Schmid, and J. Ponce. Semi-local affine parts for object recognition. In *Proceedings of the British Machine Vision Conference*, 2004.
- [74] S. Lazebnik, C. Schmid, and J. Ponce. A maximum entropy framework for part-based texture and object recognition. In *Proceedings of the International Conference on Computer Vision*, 2005.
- [75] S. Lazebnik, C. Schmid, and J. Ponce. A sparse texture representation using local affine regions. *IEEE Transactions on Pattern Analysis and Machine Intelligence*, 27(8):1265– 1278, 2005.
- [76] S. Lazebnik, C. Schmid, and J. Ponce. Beyond bags of features: Spatial pyramid matching for recognizing natural scene categories. In *Proceedings of the IEEE Confer*ence on Computer Vision and Pattern Recognition, 2006.
- [77] Y. LeCun, L. Bottou, Y. Bengio, and P. Haffner. Gradient-based learning applied to document recognition. *Proceedings of the IEEE*, 86(11):2278–2324, November 1998.
- [78] T. Leung, M. Burl, and P. Perona. Probabilistic affine invariants for recognition. In Proceedings of the IEEE Conference on Computer Vision and Pattern Recognition, pages 678–684, 1998.
- [79] T. Leung and J. Malik. Detecting, localizing and grouping repeated scene elements from an image. In *Proceedings of the European Conference on Computer Vision*, volume 1, pages 546–555, 1996.
- [80] T. Leung and J. Malik. Recognizing surfaces using three-dimensional textons. International Journal of Computer Vision, 43(1):29–44, 2001.
- [81] E. Levina and P. Bickel. The Earth Mover's distance is the Mallows distance: Some insights from statistics. In *Proceedings of the International Conference on Computer Vision*, volume 2, pages 251–256, 2001.
- [82] T. Lindeberg. Feature detection with automatic scale selection. International Journal of Computer Vision, 30(2):77–116, 1998.
- [83] T. Lindeberg and J. Gårding. Shape-adapted smoothing in estimation of 3-d depth cues from affine distortions of local 2-d brightness structure. *Image and Vision Computing*, 15:415–434, 1997.
- [84] F. Liu and R. W. Picard. Periodicity, directionality, and randomness: Wold features for image modeling and retrieval. *IEEE Transactions on Pattern Analysis and Machine Intelligence*, 18(7):722–733, 1996.
- [85] J. Liu, J.L. Mundy, D. Forsyth, A. Zisserman, and C. Rothwell. Efficient recognition of rotationally symmetric surfaces and straight homogeneous generalized cylinders. In *Proceedings of the IEEE Conference on Computer Vision and Pattern Recognition*, pages 123–128, New York City, NY, 1993.

- [86] Y. Liu, R. Collins, and Y. Tsin. A computational model for periodic pattern perception based on frieze and wallpaper groups. *IEEE Transactions on Pattern Analysis and Machine Intelligence*, 26(3):354 – 371, March 2004.
- [87] X. Llado, J. Marti, and M. Petrou. Classification of textures seen from different distances and under varying illumination direction. In *IEEE Int. Conf. Image Processing*, volume 1, pages 833–836, 2003.
- [88] D. Lowe. Perceptual Organization and Visual Recognition. Kluwer Academic Publishers, Boston, 1985.
- [89] D. Lowe. Towards a computational model for object recognition in IT cortex. In *Biologically Motivated Computer Vision*, pages 20–31, 2000.
- [90] D. Lowe. Distinctive image features from scale-invariant keypoints. International Journal of Computer Vision, 60(2):91–110, 2004.
- [91] L. Lu, K. Toyama, and G. Hager. A two-level approach for scene recognition. In Proceedings of the IEEE Conference on Computer Vision and Pattern Recognition, volume 1, pages 688–695, 2005.
- [92] S. Lyu. Mercer kernels for object recognition with local features. In Proceedings of the IEEE Conference on Computer Vision and Pattern Recognition, volume 2, pages 223–229, 2005.
- [93] S. Mahamud and M. Hebert. The optimal distance measure for object detection. In *Proceedings of the IEEE Conference on Computer Vision and Pattern Recognition*, 2003.
- [94] S. Mahamud, M. Hebert, and J. Lafferty. Combining simple discriminators for object discrimination. In *Proceedings of the European Conference on Computer Vision*, 2002.
- [95] J. Maleson, C. Brown, and J. Feldman. Understanding natural texture. In Proc. DARPA IU Workshop, pages 19–27, 1977.
- [96] J. Malik, S. Belongie, T. Leung, and J. Shi. Contour and texture analysis for image segmentation. International Journal of Computer Vision, 43(1):7–27, 2001.
- [97] J. Malik and P. Perona. Preattentive texture discrimination with early vision mechanisms. J. Opt. Soc. Am. A, 7(5):923–932, 1990.
- [98] B.S. Manjunath and W.Y. Ma. Texture features for browsing and retrieval of image data. *IEEE Transactions on Pattern Analysis and Machine Intelligence*, 18(5):837– 842, 1996.
- [99] J. Mao and A. Jain. Texture classification and segmentation using multiresolution simultaneous autoregressive models. *Pattern Recognition*, 25:173–188, 1992.
- [100] R. Marée, P. Geurts, J. Piater, and L. Wehenkel. Random subwindows for robust image classification. In *Proceedings of the IEEE Conference on Computer Vision and Pattern Recognition*, volume 1, pages 34–40, 2005.

- [101] J. Matas, O. Chum, U. Martin, and T Pajdla. Robust wide baseline stereo from maximally stable extremal regions. In *Proceedings of the British Machine Vision Conference*, volume 1, pages 384–393, 2002.
- [102] A. McCallum and K. Nigam. A comparison of event models for naive bayes text classification. In AAAI-98 Workshop on Learning for Text Categorization, pages 41– 48, 1998.
- [103] K. Mikolajczyk and C. Schmid. Indexing based on scale invariant interest points. In Proceedings of the International Conference on Computer Vision, pages 525–531, 2001.
- [104] K. Mikolajczyk and C. Schmid. An affine invariant interest point detector. In Proceedings of the European Conference on Computer Vision, volume 1, pages 128–142, 2002.
- [105] K. Mikolajczyk and C. Schmid. Scale and affine invariant interest point detectors. International Journal of Computer Vision, 60(1):63–86, 2004.
- [106] K. Mikolajczyk and C. Schmid. A performance evaluation of local descriptors. *IEEE Transactions on Pattern Analysis and Machine Intelligence*, 2005. Accepted.
- [107] K. Mikolajczyk, T. Tuytelaars, C. Schmid, A. Zisserman, J. Matas, F. Schaffalitzky, T. Kadir, and L. Van Gool. A comparison of affine region detectors. *International Journal of Computer Vision*, 2005. Accepted.
- [108] H. Moravec. Towards automatic visual obstacle avoidance. In Proc. International Joint Conference on Artificial Intelligence, page 584, 1977.
- [109] P. Moreno, P. Ho, and N. Vasconcelos. A Kullback-Leibler divergence based kernel for SVM classification in multimedia applications. In Advances in Neural Information Processing Systems, 2003.
- [110] H. Murase and S. Nayar. Visual learning and recognition of 3D objects from appearance. International Journal of Computer Vision, 14(1):5–24, 1995.
- [111] K. Murphy, A. Torralba, and W. Freeman. Using the forest to see the trees: a graphical model relating features, objects and scenes. In Advances in Neural Information Processing Systems, 2003.
- [112] R. Nevatia and T.O. Binford. Description and recognition of complex curved objects. Artificial Intelligence Journal, 8:77–98, 1977.
- [113] W. Niblack, R. Barber, W. Equitz, M. Fickner, E. Glasman, D. Petkovic, and P. Yanker. The QBIC project: Querying images by content using color, texture and shape. In SPIE Conference on Geometric Methods in Computer Vision II, 1993.
- [114] K. Nigam, J. Lafferty, and A. McCallum. Using maximum entropy for text classification. In *IJCAI Workshop on Machine Learning for Information Filtering*, pages 61–67, 1999.
- [115] K. Nigam, A. McCallum, S. Thrun, and T. Mitchell. Text classification from labeled and unlabeled documents using em. *Machine Learning*, 39(2/3):103–134, 2000.

- [116] M.-E. Nilsback and B. Caputo. Cue integration through discriminative accumulation. In Proceedings of the IEEE Conference on Computer Vision and Pattern Recognition, pages 578–585, 2004.
- [117] A. Oliva and A. Torralba. Modeling the shape of the scene: a holistic representation of the spatial envelope. *International Journal of Computer Vision*, 42(3):145–175, 2001.
- [118] A. Opelt, M. Fussenegger, A. Pinz, and P. Auer. Weak hypotheses and boosting for generic object detection and recognition. In *Proceedings of* the European Conference on Computer Vision, volume 2, pages 71–84, 2004. http://www.emt.tugraz.at/~pinz/data.
- [119] R. Picard, T. Kabir, and F. Liu. Real-time recognition with the entire Brodatz texture database. In Proceedings of the IEEE Conference on Computer Vision and Pattern Recognition, pages 638–639, 1993.
- [120] M. Pontil and A. Verri. Support vector machines for 3D object recognition. IEEE Transactions on Pattern Analysis and Machine Intelligence, 20(6):637–646, 1998.
- [121] J. Puzicha, Y. Rubner, C. Tomasi, and J. Buhmann. Empirical evaluation of dissimilarity measures for color and texture. In *Proceedings of the International Conference* on Computer Vision, volume 2, pages 1165–1172, 1999.
- [122] P. Quelhas, F. Monay, J.-M. Odobez, D. Gatica, T. Tuytelaars, and L. Van Gool. Modeling scenes with local descriptors and latent aspects. In *Proceedings of the International Conference on Computer Vision*, 2005.
- [123] D. Ramanan, D. Forsyth, and K. Barnard. Detecting, localizing, and recovering kinematics of textured animals. In *Proceedings of the IEEE Conference on Computer Vision and Pattern Recognition*, 2005.
- [124] D. Ramanan, D. Forsyth, and A. Zisserman. Strike a pose: Tracking people by finding stylized poses. In Proceedings of the IEEE Conference on Computer Vision and Pattern Recognition, 2005.
- [125] T. Randen and J. Husøy. Filtering for texture classification: A comparative study. *IEEE Transactions on Pattern Analysis and Machine Intelligence*, 21(4):291–310, 1999.
- [126] L. Renninger and J. Malik. When is scene identification just texture recognition? *Vision Research*, 44:2301–2311, 2004.
- [127] L. Roberts. Machine perception of three-dimensional solids. In J. Tippett, editor, Optical and Electro-Optical Information Processing, chapter 9, pages 159–197. MIT Press, Cambridge, MA, 1965.
- [128] A. Rosenfeld, R. Hummel, and S. Zucker. Scene labeling by relaxation operations. *IEEE Trans. on Systems, Man, and Cybernetics*, 6(6):420–433, 1976.
- [129] F. Rothganger, S. Lazebnik, C. Schmid, and J. Ponce. Segmenting, modeling and matching video clips containing multiple moving objects. In *Proceedings of the IEEE* Conference on Computer Vision and Pattern Recognition, 2004.

- [130] F. Rothganger, S. Lazebnik, C. Schmid, and J. Ponce. 3d object modeling and recognition using affine-invariant patches and multi-view spatial constraints. *International Journal of Computer Vision*, 2006.
- [131] Y. Rubner and C. Tomasi. Texture-based image retrieval without segmentation. In Proceedings of the International Conference on Computer Vision, pages 1018–1024, 1999.
- [132] Y. Rubner, C. Tomasi, and L. Guibas. The Earth Mover's distance as a metric for image retrieval. International Journal of Computer Vision, 40(2):99–121, 2000.
- [133] G. Salton and M. McGill. An Introduction to Modern Information Retrieval. McGraw-Hill, New York, 1983.
- [134] F. Schaffalitzky and A. Zisserman. Viewpoint invariant texture matching and wide baseline stereo. In *Proceedings of the International Conference on Computer Vision*, volume 2, pages 636–643, 2001.
- [135] F. Schaffalitzky and A. Zisserman. Multi-view matching for unordered image sets, or "How do I organize my holiday snaps?". In *Proceedings of the European Conference* on Computer Vision, volume 1, pages 414–431, 2002.
- [136] B. Schiele and J. Crowley. Recognition without correspondence using multidimensional receptive field histograms. *International Journal of Computer Vision*, 36(1):31–50, 2000.
- [137] C. Schmid. Constructing models for content-based image retrieval. In Proceedings of the IEEE Conference on Computer Vision and Pattern Recognition, volume 2, pages 39–45, 2001.
- [138] C. Schmid and R. Mohr. Local greyvalue invariants for image retrieval. IEEE Transactions on Pattern Analysis and Machine Intelligence, 19(5):530–535, 1997.
- [139] C. Schmid, J. Ponce, M. Hebert, and A. Zisserman, editors. Towards Category-Level Object Recognition. Springer Lecture Notes in Computer Science, 2006.
- [140] H. Schneiderman and T. Kanade. Object detection using the statistics of parts. International Journal of Computer Vision, 56(3):151–177, 2004.
- [141] B. Schölkopf and A. Smola. Learning with Kernels: Support Vector Machines, Regularization, Optimization and Beyond. MIT Press, Cambridge, MA, 2002.
- [142] T. Serre, L. Wolf, and T. Poggio. Object recognition with features inspired by visual cortex with sets of image features. In *Proceedings of the IEEE Conference on Computer Vision and Pattern Recognition*, 2005.
- [143] J. Sivic, B. Russell, A. Efros, A. Zisserman, and W. Freeman. Discovering objects and their location in images. In *Proceedings of the International Conference on Computer Vision*, 2005.
- [144] J. Sivic and A. Zisserman. Video Google: A text retrieval approach to object matching in videos. In Proceedings of the International Conference on Computer Vision, pages 1470–1477, 2003.

- [145] D. Squire, W. Muller, H. Muller, and J. Raki. Content-based query of of image databases, inspirations from text retrieval: inverted files, frequency-based weights and relevance feedback. In *Proceedings of the 11th Scandinavian conference on image anal*ysis, pages 143–149, 1999.
- [146] M. Swain and D. Ballard. Color indexing. International Journal of Computer Vision, 7(1):11–32, 1991.
- [147] M. Szummer and R. Picard. Indoor-outdoor image classification. In IEEE International Workshop on Content-Based Access of Image and Video Databases, pages 42–51, 1998.
- [148] S. Thorpe, D. Fize, and C. Marlot. Speed of processing in the human visual system. *Nature*, 381:520–522, 1996.
- [149] A. Torralba, K. Murphy, and W. Freeman. Sharing features: Efficient boosting procedures for multiclass object detection. In Proceedings of the IEEE Conference on Computer Vision and Pattern Recognition, 2004.
- [150] A. Torralba, K. P. Murphy, W. T. Freeman, and M. A. Rubin. Context-based vision system for place and object recognition. In *Proceedings of the International Conference* on Computer Vision, 2003.
- [151] A. Turina, T. Tuytelaars, T. Moons, and L. Van Gool. Grouping via the matching of repeated patterns. In *Proceedings of the International Conference on Advances in Pattern Recognition*, pages 250–259, 2001.
- [152] M. Turk and A. Pentland. Face recognition using eigenfaces. In *Proceedings of the IEEE Conference on Computer Vision and Pattern Recognition*, pages 586–591, 1991.
- [153] T. Tuytelaars and L. Van Gool. Matching widely separated views based on affinely invariant neighbourhoods. International Journal of Computer Vision, 59(1):61–85, 2004.
- [154] S. Ullman, E. Sali, and M. Vidal-Naquet. A fragment-based approach to object representation and classification. In C. Arcelli et al., editor, *Proceedings of the International Workshop on Visual Form*, pages 85–100, 2001. Springer-Verlag Lecture Notes in Computer Science 2059.
- [155] A. Vailaya, A. Jain, and H.-J. Zhang. On image classification: city vs. landscape. In Proceedings of the IEEE Workshop on Content-Based Access of Image and Video Libraries, pages 3–8, 1998.
- [156] M. Varma and A. Zisserman. Classifying images of materials: Achieving viewpoint and illumination independence. In *Proceedings of the European Conference on Computer Vision*, volume 3, pages 255–271, 2002.
- [157] M. Varma and A. Zisserman. Texture classification: Are filter banks necessary? In Proceedings of the IEEE Conference on Computer Vision and Pattern Recognition, volume 2, pages 691–698, 2003.
- [158] J. K. M. Vetterli. Wavelets and subband coding. Prentice Hall, 1995.

- [159] J. Vogel and B. Schiele. A semantic typicality measure for natural scene categorization. In *DAGM Annual Pattern Recognition Symposium*, 2004.
- [160] H. Voorhees and T. Poggio. Detecting textons and texture boundaries in natural images. In Proceedings of the International Conference on Computer Vision, pages 250–258, 1987.
- [161] C. Wallraven, B. Caputo, and A. Graf. Recognition with local features: the kernel recipe. In *Proceedings of the International Conference on Computer Vision*, volume 1, pages 257–264, 2003.
- [162] M. Weber, M. Welling, and P. Perona. Towards automatic discovery of object categories. In *Proceedings of the IEEE Conference on Computer Vision and Pattern Recognition*, volume 2, pages 2101–2109, 2000.
- [163] M. Weber, M. Welling, and P. Perona. Unsupervised learning of models for recognition. In Proceedings of the European Conference on Computer Vision, volume 1, pages 18– 32, 2000.
- [164] J. Willamowski, D. Arregui, G. Csurka, C. R. Dance, and L. Fan. Categorizing nine visual classes using local appearance descriptors. In *ICPR Workshop on Learning for Adaptable Visual Systems*, 2004.
- [165] J. Wu and M. J. Chantler. Combining gradient and albedo data for rotation invariant classification of 3d surface texture. In *Proceedings of the International Conference on Computer Vision*, volume 2, pages 848–855, 2003.
- [166] K. Xu, B. Georgescu, D. Comaniciu, and P. Meer. Performance analysis in contentbased retrieval with textures. In *Proceedings of the International Conference on Pattern Recognition*, volume 4, pages 275–278, 2000.
- [167] J. Zhang, P. Fieguth, and D. Wang. Random field models. In A. Bovik, editor, Handbook of Image and Video Processing, pages 301–312. Academic Press, San Diego, CA, 2000.
- [168] J. Zhang, M. Marszalek, S. Lazebnik, and C. Schmid. Local features and kernels for classification of texture and object categories: An in-depth study. Technical Report RR-5737, INRIA Rhône-Alpes, 2005.
- [169] S.C. Zhu, Y.N. Wu, and D. Mumford. Filters, random fields, and maximum entropy (FRAME): Towards a unified theory for texture modeling. *International Journal of Computer Vision*, 27(2):1–20, 1998.
- [170] S.C. Zhu and A.L. Yuille. FORMS: a flexible object recognition and modeling system. In *Proceedings of the International Conference on Computer Vision*, pages 450–472, Boston, MA, 1995.

AUTHOR'S BIOGRAPHY

Svetlana Lazebnik was born in Kiev, Ukraine, in 1979 and emigrated to the United States in 1992. She has received the B.S. degree in computer science from DePaul University in Chicago, IL, in 2000 and the M.S. degree in computer science from the University of Illinois at Urbana-Champaign in 2002. She is completing the Ph.D. degree under the supervision of Prof. Jean Ponce. Her research interests include computer vision, object and pattern recognition, and machine learning.